CAPACITOR VOLTAGE BALANCING, FAULT DETECTION, AND FAULT TOLERANT CONTROL TECHNIQUES OF MODULAR MULTI-LEVEL CONVERTERS

MAHMOUD OSAMA MOHAMMED ELWADIE ABDELSALAM

A thesis submitted in partial fulfilment of the requirement of Staffordshire University for the degree of Doctor of Philosophy

August 2016

Abstract

Modular Multilevel Converters (MMCs) are distinguished by their modular nature that makes them suitable for wide range of high power and high voltage applications. However, they are vulnerable to internal faults because of the large number of series connected Sub-Modules. Additionally, it is highly recommended not to block the converter even if it is subjected to internal faults to secure the supply, to increase the reliability of the system and prevent unscheduled maintenance. This thesis introduces a fault tolerant control system for controlling the MMC in normal as well as abnormal operating conditions. This is done through developing a new adaptive voltage balancing strategy based on capacitor voltage estimation utilizing ADAptive LInear NEuron (ADALINE) and Recursive Least Squares (RLS) algorithms. The capacitor voltage balancing techniques that have been proposed in literature are based on measuring the capacitor voltage of each sub-module. On contrary, the proposed strategy eliminates the need of these measurements and associated communication links with the central controller.

Furthermore, the thesis presents a novel fault diagnosis algorithm using the estimated capacitor voltages which are utilized to detect and localize different types of sub-module faults. The proposed fault diagnosis algorithm surpasses the methods presented in literature by its fast fault detection capability without the need of any extra sensing elements or special power circuit.

Finally, a new Fault Tolerant Control Unit (FTCU) is proposed to tolerate the faults located inside the MMC submodules. The proposed FTCU is based on a sorting algorithm which modifies the parameters of the voltage balancing technique in an adaptive manner to overcome the reduction of the active submodules and secure the MMC operation without the need of full shut-down. Most of fault tolerant strategies that have been proposed by other researchers are based on using redundant components, while the proposed FTCU does not need any extra components.

The dynamic performance of the proposed strategy is investigated, using PSCAD/EMTDC simulations and hardware in the loop (HIL) real-time simulations, under different normal and faulty operating conditions. The accuracy and the time response of the proposed fault detection and tolerant control units result in stabilizing

the operation of the MMC under different types of faults. Consequently, the proposed integrated control strategy improves the reliability of the MMC.

Acknowledgments

First and above all, I praise God, the almighty for providing me this opportunity and granting me the capability to proceed successfully. This thesis appears in its current form due to the assistance and guidance of several people. I would therefore like to offer my sincere thanks to all of them.

Prof. Sarath Tennakon, my esteemed principal supervisor, my cordial thanks for accepting me as a PhD student, your warm encouragement, thoughtful guidance, critical comments, and correction of the thesis.

I want to express my deep thanks to my esteemed external supervisor Dr Mostafa Marei for the trust, the insightful discussion, offering valuable advice, for your support during the whole period of the study, and especially for your patience and guidance during the writing process.

I would like to thank my esteemed second supervisor, Dr Alison Griffiths for her excellent advises throughout the PhD period.

Hatem Diab, I greatly appreciate your excellent assistance and your spiritual supports for me during my PhD study. I will never forget the hard times that would never have been passed without you standing to me. Your situations have always showed a real man and best friend who always exerts the maximum effort to support others, really thank you very much.

Many thanks to the management of Staffordshire University for allowing me to use the university' facilities and helping me in getting anything I need during my study.

Thanks also to all of my research colleagues, Hesham, Hussain, Samir, Mahmoud, Raouf, Monday, Masoom, Ernest and Masoud for providing a good atmosphere in our department and for useful discussions.

I would like to express my greatest gratitude to Prof. Yasser Galal for his endless support and sharp actions that helped me to take the opportunity of studying abroad.

I also appreciate the financial support of the Arab Academy for Science and Technology and Maritime Transport, during my PhD study. I cannot finish without thanking my Father who passed away during my study, although you are not with me but I live with your advices and encouragements through our life.

I warmly thank and appreciate my mother for her spiritual support in all aspects of my life. Although I have lived far from you, but communication with you, provided emotional atmosphere for me.

I also would like to thank my sisters, Mai and Yomna, they have provided assistance in numerous ways. Also, I want to thank my father and mother in law for their support and help during this journey.

I want to express my gratitude and deepest appreciation to my lovely sweet daughter, Lilia. Your smile was one of the greatest sources of motivation to work, thanks for being in my life.

And finally, my lovely wife, Dear Amira, without your supports and encouragements, I could not have finished this work, it was you who kept the fundamental of our family, and I understand it was difficult for you, therefore, I can just say thanks for everything and may Allah give you all the best in return.

Table of Contents

| Abstract | | . I |
|----------------------|---|-----------------|
| Acknowle | dgments | II |
| Table of C | ontents | V |
| List of Fig | ures | Х |
| List of Tal | blesXV | ΊI |
| List of Ab | breviationsXV | III |
| | | |
| Chapter1 | - Introduction | .1 |
| 1.1 | Background | .1 |
| 1.2 | Motivation and Research Objectives | .3 |
| 1.3 | Thesis Outline | .6 |
| 1.4 | List of Publications | .8 |
| 1.4.1 | Journal publications | .8 |
| 1.4.2 | Conference publications | .8 |
| 1.4.3 | Co-authored conference publications | .9 |
| Chapter Modulatio | 2 – Modular Multilevel Converters: Structure, Operation, an | 1d 10 |
| 2.1 | Introduction | 10 |
| 2.2 | Multilevel Converters | 10 |
| 2.2.1 | Cascaded H-Bridge converter | 11 |
| 2.2.2 | Neutral-point clamped multilevel converters | 12 |
| 2.2.3 | Flying capacitor multilevel converters | 13 |
| 2.2.4 | Hybrid multilevel converters | 14 |
| 2.3 | Modular Multilevel Converters | 15 |
| 2.3.1 | MMC modes of operation | 16 |
| 2.3.2 | MMC mathematical model | 17 |
| 2.4 | Modulation Techniques of the MMC | 19 |
| 2.4.1 | Space vector modulation | 19 |
| 2.4.2 | Nearest level modulation | 20 |
| 2.4.3 | Carrier-based pulse-width modulation | 22 |
| 2.5 | Design Considerations of MMCs | 24 |
| 2.5.1 | Submodule capacitance | 24 |
| 2.5.2 | Arm inductance | 26 |
| 2.6 | Chapter Summary | 26 |
| Chanter 3 | – Review of Canacitor Voltage Balancing, Fault Detection, and Fau | ılt |

| 3.1 | Introduction | .27 |
|---------------------|---|------------------|
| 3.2 | Capacitor Voltage Balancing of MMC Submodules | .27 |
| 3.2.1 | Capacitor voltage balancing based on energy control | .28 |
| 3.2.2 | Capacitor voltage balancing based on distributed control | .29 |
| 3.2.3 modulation | Capacitor voltage balancing based on reduced switching frequent | ncy .30 |
| 3.2.4 | Capacitor voltage balancing based on sorting algorithms | .31 |
| 3.3 | The MMC Submodule Faults | .33 |
| 3.3.1 | Faults in power electronic devices | .34 |
| 3.3.2 | Faults in submodule capacitors | .36 |
| 3.4 | The Concept of Fault Detection | .36 |
| 3.4.1 | Hardware-based fault detection | .37 |
| 3.4.2 | Software-based fault detection | .37 |
| 3.5 | Fault Detection Techniques in MMCs | .38 |
| 3.5.1 | IGBT fault detection using the rate of change of the collector current | : 39 |
| 3.5.2 | IGBT fault detection using the behaviour of the gate voltage | .39 |
| 3.5.3 voltage | Submodule fault detection using the measurements of submodule out | put .40 |
| 3.5.4 | IGBT fault detection using artificial neural networks | .42 |
| 3.5.5 | Submodule fault detection using sliding mode observation | .43 |
| 3.6 | The Concept of Fault Tolerant Control | .45 |
| 3.6.1 | Active fault tolerant control | .45 |
| 3.6.2 | Passive fault tolerant control | .46 |
| 3.7 | Fault Tolerant Control of MMCs | .46 |
| 3.7.1 | Fault tolerant control using redundant submodules | .47 |
| 3.7.2 | Fault tolerant control using redundant IGBTs | .48 |
| 3.8 | Chapter Summary | .49 |
| Chapter Capacitor | 4 – Proposed Capacitor Voltage Balancing Technique Based Voltage Estimation Algorithms | on .51 |
| 4.1 | Introduction | .51 |
| 4.2 | Proposed Technique of Capacitor Voltage Balancing | .52 |
| 4.2.1 | The capacitor voltage estimation technique | .52 |
| 4.2.1.1 | Capacitor voltage estimation using ADALINE technique | .53 |
| 4.2.1.2 | The capacitor voltage estimation using RLS technique | .55 |
| 4.2.2 | The averaging control | .58 |
| 4.2.3 | Balancing control | .61 |
| 4.3 | MMC Simulation Models | .62 |
| 4.3.1 | MMC Model 1 | .63 |

| 4.3.2 | MMC Model 2 |
|--|---|
| 4.4 | Simulation Results |
| 4.4.1 estimation- | Case 1: Dynamic performance of the proposed capacitor voltage based control strategy |
| 4.4.2 MMC | Case 2: Dynamic performance during the boosting operation of the |
| 4.4.3 connected | Case 3: Performance of capacitor voltage estimation algorithm in grid- MMC |
| 4.5 | Chapter Summary75 |
| Chapter 5 Estimation | – Proposed Fault Detection Technique Based on Capacitor Voltage n Algorithms77 |
| 5.1 | Introduction77 |
| 5.2 | The Proposed Fault Detection Strategy |
| 5.3 | Simulation Results |
| 5.3.1 | Case 1: Performance under IGBT open circuit fault |
| 5.3.2 | Case 2: Performance under IGBT short circuit fault |
| 5.3.3 | Case 3: Performance under multiple faults in two different arms89 |
| 5.4 | Chapter Summary |
| Chapter 6 | 5 – Proposed Fault Tolerant Control Strategy Based on a Sorting |
| Algorithm | |
| Algorithm 6.1 | Introduction |
| Algorithm 6.1 6.2 | Introduction |
| Algorithm6.16.26.3Technique | Introduction |
| Algorithm 6.1 6.2 6.3 Technique 6.3.1 | Introduction |
| Algorithm 6.1 6.2 6.3 Technique 6.3.1 6.3.2 | Introduction |
| Algorithm 6.1 6.2 6.3 Technique 6.3.1 6.3.2 6.3.3 tolerant con | Introduction |
| Algorithm 6.1 6.2 6.3 Technique 6.3.1 6.3.2 6.3.3 tolerant con 6.3.4 | Introduction |
| Algorithm 6.1 6.2 6.3 Technique 6.3.1 6.3.2 6.3.3 tolerant con 6.3.4 6.4 | Introduction |
| Algorithm 6.1 6.2 6.3 Technique 6.3.1 6.3.2 6.3.3 tolerant con 6.3.4 6.4 6.4.1 | Introduction |
| Algorithm 6.1 6.2 6.3 Technique 6.3.1 6.3.2 6.3.3 tolerant con 6.3.4 6.4 6.4.1 6.4.2 | Introduction94Introduction94The Proposed Fault Tolerant Control Algorithm95Reliability Assessment of the Proposed Fault Tolerant Control98MMC reliability definitions98Reliability calculations for MMC without fault tolerant control99Reliability calculations for MMC with the redundancy based faultntrol99Reliability calculations for MMC with the proposed fault tolerant control |
| Algorithm 6.1 6.2 6.3 Technique 6.3.1 6.3.2 6.3.3 tolerant con 6.3.4 6.4 6.4.1 6.4.2 6.4.3 | Introduction 94 Introduction 94 The Proposed Fault Tolerant Control Algorithm 95 Reliability Assessment of the Proposed Fault Tolerant Control 98 MMC reliability definitions 98 Reliability calculations for MMC without fault tolerant control 99 Reliability calculations for MMC with the redundancy based fault ntrol 99 Reliability calculations for MMC with the proposed fault tolerant control 100 Simulation Results 103 Case 1: The FTCU Performance under IGBT open circuit fault 103 Case 2: The performance of the FTCU under multiple faults 109 |
| Algorithm 6.1 6.2 6.3 Technique 6.3.1 6.3.2 6.3.3 tolerant con 6.3.4 6.4 6.4.1 6.4.2 6.4.3 6.4.4 | Introduction94Introduction94The Proposed Fault Tolerant Control Algorithm95Reliability Assessment of the Proposed Fault Tolerant Control98MMC reliability definitions98Reliability calculations for MMC without fault tolerant control99Reliability calculations for MMC with the redundancy based faultntrol99Reliability calculations for MMC with the proposed fault tolerant control |
| Algorithm 6.1 6.2 6.3 Technique 6.3.1 6.3.2 6.3.3 tolerant con 6.3.4 6.4 6.4.1 6.4.2 6.4.3 6.4.4 6.5 | Introduction |
| Algorithm 6.1 6.2 6.3 Technique 6.3.1 6.3.2 6.3.3 tolerant con 6.3.4 6.4 6.4.1 6.4.2 6.4.3 6.4.3 6.4.4 6.5 Chapter 7 Real-Time | Introduction |

| 7.2 | HIL System Description |
|-----------------------|--|
| 7.3 | HIL System Implementation119 |
| 7.3.1 | The real-time physical controller120 |
| 7.3.2 | The real-time digital simulator121 |
| 7.4 | Practical Assessment of The Proposed Techniques122 |
| 7.5 | HIL Simulation Results |
| 7.5.1 of output v | Case 1: The performance of the proposed control under dynamic change roltage reference |
| 7.5.2 circuit faul | Case 2: The performance of the propsoed control under IGBT open t |
| 7.5.3 circuit faul | Case 3: The performance of the proposed control under IGBT short t |
| 7.5.4 submodule | Case 4: The performance of the proposed control under multiple faults |
| 7.5.5 failures | Case 5: The Performance of the propsoed control under cascading |
| 7.6 | Chapter Summary |
| Chapter 8 | - Conclusion |
| 8.1 | Conclusion137 |
| 8.2 | Summary of Contributions138 |
| 8.3 | Further Research and expected outcomes139 |
| Reference | s 140 |
| Appendix | A – Modelling of Wind Energy Conversion Systems152 |
| A.1 Model | ling of the Wind Turbine152 |
| A.2 Model | ling of PMSG153 |
| A.3 Model | ling of the DFIG153 |
| A.4 Contro | ol of Active and Reactive Power Injected into the Grid155 |
| A.4.1 Phas | e locked loop155 |
| A.4.2 AC a | and DC voltage control loops155 |
| A.4.3 Inne | r current loop156 |
| Appendix | B – Details of PSCAD Simulation Models 157 |
| Appendix | C – Details of HIL Real-Time Simulations170 |
| C.1 The R | Γ-LAB/Simulink Real-time Simulation Model171 |
| C.2 The cF | RIO/LABVIEW Programming Blocks179 |
| Appendix | D – Selected Publications |

List of Figures

| Figure 1.1: A 401-level MMC made by Siemens: a) a picture of the MMC, b) a picture for the submodule |
|---|
| Figure 1.3: Faults probabilities in power converters |
| Figure 2.1: Types of multilevel converters |
| Figure 2.2: Cascaded H-Bridge converter: a) the structure, b) the waveform of output voltage |
| Figure 2.3: Structure of neutral-point clamped multilevel converter13 |
| Figure 2.4: The structure of flying capacitor multilevel converters14 |
| Figure 2.5: Three-phase MMC topology and internal submodule16 |
| Figure.2.6: Different operating modes of MMC submodules17 |
| Figure 2.7: MMC average circuit diagram17 |
| Figure 2.8: Switching plan of space vector for an <i>N</i> level MMC20 |
| Figure 2.9: The triangle of stationary vectors |
| Figure 2.10: The NLM technique |
| Figure 2.11: The LS-PWM technique |
| Figure 2.12: The PS-PWM technique |
| Figure 3.1: Block diagram of capacitor voltage balancing based on energy control |
| Figure 3.2: Block diagram of capacitor voltage balancing based on distributed control |
| Figure 3.3: Block diagram of capacitor voltage balancing based on reduced switching frequency modulation |
| Figure 3.4: Block diagram of the conventional sorting algorithm based capacitor voltage balancing algorithm |
| Figure 3.5: Block diagram of the modified sorting algorithm based capacitor voltage balancing algorithm |
| Figure 3.6: Types of fault detection techniques |
| Figure 3.7: The $\frac{di_c}{dt}$ measuring circuit |
| Figure 3.8: Gate voltage-based fault detection circuit40 |
| Figure 3.9: Block diagram of the submodule output voltage and derivation of capacitor voltage |
| Figure 3.10: Flowchart of submodule output voltage-based fault detection41 |
| Figure 3.11: The block diagram of the ANN IGBT fault detection method43 |
| Figure 3.12: Flowchart of the SMO submodule fault detection technique43 |
| Figure 3.13: Structure of active FTC |
| Figure 3.14: Structure of MMC with redundant submodules47 |

| Figure 3.15: Block diagram of fault tolerant control with redundant submodules |
|--|
| Figure 3.16: The diagram of MMC submodule with IGBT redundanct for: a) half bridge, b) full bridge |
| Figure 3.17: Tolerating the faults using redundant IGBTs |
| Figure 4.1: Flow chart of capacitor voltage estimation using ADALINE technique |
| Figure 4.2: Flow chart of capacitor voltage estimation using RLS technique56 |
| Figure 4.3: Block diagram of capacitor voltage estimation unit |
| Figure 4.4: Comparison between the RLS and ADALINE performances: a) actual capacitor voltages versus RLS and ADALINE estimated signals, b) percentage error between actual voltage and RLS estimation, c) percentage error between actual voltage and ADALINE estimation |
| Figure 4.5: Block diagram of the averaging controller |
| Figure 4.6: The bode plot of the inner loop of the averaging controller60 |
| Figure 4.7: The bode plot of the outer loop of the averaging controller61 |
| Figure 4.8: Block diagram of the balancing controller62 |
| Figure 4.9: The circuit diagram of the first MMC model63 |
| Figure 4.10: The arm energy variation of the MMC simulated in the first model64 |
| Figure 4.11: The circuit diagram of the second MMC model65 |
| Figure 4.12: The arm energy variation of the MMC simulated in the second model |
| Figure 4.13: Load waveforms of Case 1, while disabling the proposed estimation unit: a) three-phase voltages, b) three-phase load currents, c) voltages of Leg <i>a</i> submodules |
| Figure 4.14: Load waveforms of Case 1, while enabling the proposed estimation unit: a) three-phase voltages, b) three-phase load currents, c) voltages of Leg <i>a</i> submodules |
| Figure 4.15: Case 1 actual and estimated voltages of the upper arm submodules of Phase <i>a</i> 70 |
| Figure 4.16: Case 1 actual and estimated voltages of the lower arm submodules of Phase <i>a</i> 70 |
| Figure 4.17: Action of the different controllers for Leg <i>a</i> during Case 1: a) actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm |
| Figure 4.18: Load waveforms of Case 2: a) three-phase voltages, b) three-phase load currents, c) voltages of Leg <i>a</i> submodules72 |
| Figure 4.19: Case 2 actual and estimated voltages of the upper arm submodules of Phase <i>a</i> during the boosting operation |
| Figure 4.20: Case 2 actual and estimated voltages of the lower arm submodules of Phase <i>a</i> during the boosting operation |

| Figure 4.21: Action of the different controllers for Leg <i>a</i> during the boosting operation (Case 2): a) actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm |
|---|
| Figure 4.22: Case 3 voltages and currents of grid: a) three-phase voltages, b) three-phase currents |
| Figure 4.23: Case 3 actual and estimated signals of submodule capacitor voltages in Leg a: a) summation of capacitor voltages in the upper arm, b) summation of capacitor voltages in the lower arm, c) capacitor voltage of one submodule in the upper arm, d) capacitor voltage of one submodule in the lower arm |
| Figure 5.1. Block diagram of the proposed FDU78 |
| Figure 5.2. Flowchart of the proposed fault detection technique79 |
| Figure 5.3. FDU performance during an open-circuit fault on the third submodule of the upper arm: a) Three-phase voltages, b) three-phase load currents, c) voltages of leg <i>a</i> submodules |
| Figure 5.4. Actual and estimated voltages of the upper arm submodules of phase <i>a</i> , under an open-circuit fault |
| Figure 5.5. Actual and estimated voltages of the lower arm submodules of phase <i>a</i> , under an open-circuit fault |
| Figure 5.6. Action of the different controllers for leg <i>a</i> during the open-circuit fault: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm |
| Figure 5.7: The performance of the FDU under IGBT open circuit fault: a) The RLS estimated voltages of the upper arm of phase a , b) The detection time consumed by the FDU to detect the open circuit fault |
| Figure 5.8: Operation during a short-circuit fault on the third submodule of the upper arm: a) Three-phase voltages, b) three- phase load currents, c) voltages of leg <i>a</i> submodules |
| Figure 5.9: Actual and estimated voltages of the upper arm submodules of phase <i>a</i> , under a short-circuit fault |
| Figure 5.10: Actual and estimated voltages of the lower arm submodules of phase <i>a</i> , under a short-circuit fault |
| Figure 5.11: Action of the different controllers for leg <i>a</i> during the short-circuit fault: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm |
| Figure 5.12: The performance of the FDU under IGBT short circuit fault: a) The RLS estimated voltages of the upper arm of phase a , b) The detection time consumed by the FDU to detect the open circuit fault |
| Figure 5.13: Operation during multiple faults: a) Three-phase voltages, b) three-phase load currents, c) voltages of leg <i>a</i> submodules90 |
| Figure 5.14: Actual and estimated voltages of the upper arm submodules of phase <i>a</i> , under two faults |

| Figure 5.15: Actual and estimated voltages of the lower arm submodules of phase <i>a</i> , under two faults92 |
|--|
| Figure 5.16: Action of the different controllers for leg a during multiple faults: a) actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm |
| Figure 5.17: The performance of the FDU under two faults: a) The RLS estimated voltages of the upper arm of phase a , b) The detection time consumed by the FDU to detect the open circuit fault |
| Figure 6.1: The proposed fault tolerant control algorithm |
| Figure 6.2: Block diagram of the proposed MMC control strategy97 |
| Figure 6.3: Markov chain for a modular multilevel converter without fault tolerant control |
| Figure 6.4: Markov chain of the MMC with the classical redundancy based fault tolerant control100 |
| Figure 6.5: Markov chain for a five-level modular multilevel converter with the proposed fault tolerant control |
| Figure 6.6: Plot of reliability density functions before and after the application of the proposed fault tolerant control |
| Figure 6.7: Dynamic performance of the proposed FTCU under an open-circuit fault in a submodule |
| Figure 6.8: Enabling the proposed FTCU for an open-circuit fault: a) Three-phase voltages, b) three-phase load currents, c) voltages of leg <i>a</i> submodules104 |
| Figure 6.9: Actual and estimated voltages of the upper arm submodules of phase <i>a</i> , under an open-circuit fault while enabling the proposed FTCU105 |
| Figure 6.10: Actual and estimated voltages of the lower arm submodules of phase <i>a</i> , under an open-circuit fault while enabling the proposed FTCU105 |
| Figure 6.11: Action of the different controllers for leg a with enabling the FTCU for an open-circuit fault: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm |
| Figure 6.12: Dynamic performance of the proposed FTCU under a short-circuit fault in a submodule |
| Figure 6.13: Enabling the proposed FTCU for a short-circuit fault: a) Three-phase voltages, b) three-phase load currents, c) voltages of leg <i>a</i> submodules107 |
| Figure 6.14: Actual and estimated voltages of the upper arm submodules of phase <i>a</i> , under a short-circuit fault |
| Figure 6.15: Actual and estimated voltages of the lower arm submodules of phase <i>a</i> , under a short-circuit fault |
| Figure 6.16: Action of the different controllers for leg a with enabling the FTCU for a short-circuit fault: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm |

| Figure 6.17: Enabling the proposed FTCU for multiple faults: a) Three-phase voltages, b) three-phase load currents, c) voltages of leg <i>a</i> submodules110 |
|---|
| Figure 6.18: Actual and estimated voltages of the upper arm submodules of phase <i>a</i> , under multiple faults |
| Figure 6.19: Actual and estimated voltages of the lower arm submodules of phase <i>a</i> , under multiple faults |
| Figure 6.20: Action of the different controllers for leg a with enabling the FTCU for multiple faults: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm |
| Figure 6.21: Dynamic performance of the proposed FTCU under cascading failure |
| Figure 6.22: Enabling the proposed FTCU for cascading failures: a) Three-phase voltages, b) three-phase load currents, c) voltages of leg <i>a</i> submodules114 |
| Figure 6.23: Actual and estimated voltages of the upper arm submodules of phase <i>a</i> , under cascading failures114 |
| Figure 6.24: Actual and estimated voltages of the lower arm submodules of phase <i>a</i> , under cascading failures115 |
| Figure 6.25: Action of the different controllers for leg <i>a</i> with enabling the FTCU for cascading failures: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm |
| Figure 7.1: Structure of the HIL platform118 |
| Figure 7.2: The HIL platform system components: a) the block diagram of the HIL platform, b) a snapshot of the HIL system |
| Figure 7.3: The cRIO-9024 real-time controller120 |
| Figure 7.4. The application logic inside the FPGA-based controller121 |
| Figure 7.5: The OP4510 real-time digital simulator121 |
| Figure 7.6: The OP4510 internal structure122 |
| Figure 7.7: 4-way LUT |
| Figure 7.8: Performance of the proposed inner controller under dynamic change of v_o^* : a) Three-phase voltages, b) three-phase load currents |
| Figure 7.9: Performance of the proposed inner controller under dynamic change of v_o^* : a) submodules voltages of upper arm in leg a, b) submodules voltages of lower arm in leg <i>a</i> |
| Figure 7.10: Action of the different controllers for leg a during dynamic change of v_o^* : a) Actual voltage of submodule 1 and its estimated signal, b) Actual and reference waveform of the circulating current |
| Figure 7.11: Action of the different controllers for leg a during dynamic change of v_o^* : a) averaging voltage reference, b) modulating signal for the first submodule of the upper arm |
| Figure 7.12: Dynamic performance of the proposed control under an open-circuit fault in <i>a</i> submodule |

| Figure 7.13: Performance of the proposed control for an open-circuit fault: a) Three- phase voltages, b) three-phase load currents |
|--|
| Figure 7.14: Performance of the proposed control for an open-circuit fault: a) submodule voltages of upper arm in leg a , b) submodule voltages of lower arm in leg a |
| Figure 7.15: RLS-estimated signals in the upper arm of leg <i>a</i> under open-circuit fault: a) first submodule (healthy), b) third submodule (faulty) |
| Figure 7.16: Action of the different controllers for leg a with enabling the proposed control for an open-circuit fault: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) modulating voltage reference for the first submodule of the upper arm |
| Figure 7.17: Dynamic performance of the proposed control under a short-circuit fault in a submodule |
| Figure 7.18: Enabling the proposed control for a short-circuit fault: a) Three-phase voltages, b) three-phase load currents |
| Figure 7.19: Enabling the proposed control for a short-circuit fault: a) submodule voltages of upper arm in leg a , b) submodule voltages of lower arm in leg a 130 |
| Figure 7.20. RLS-estimated signals in the upper arm of leg <i>a</i> under short circuit: a) first submodule (healthy), b) third submodule (faulty)130 |
| Figure 7.21: Action of the different controllers for leg a with enabling the proposed control for a short-circuit fault: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) modulating voltage reference for the first submodule of the upper arm |
| Figure 7.22: Dynamic performance of the proposed control under multiple faults: a) fault 1, b) fault 2 |
| Figure 7.24: The performance of the proposed control under multiple faults: a) submodule voltages of upper arm in leg a , b) submodule voltages of lower arm in leg a |
| Figure 7.25: Action of the different controllers for leg <i>a</i> with enabling the proposed control for multiple failure: a) The capacitor voltage estimation of submodule <i>Vcu1</i> , b) actual and reference waveform of the circulating current |
| Figure 7.26: Action of the different controllers for leg a with enabling the proposed control for multiple failure: a) averaging voltage reference, b) modulating voltage reference for the first submodule of the upper arm |
| Figure 7.27: The performance of the proposed strategy under cascading failure: a) three-phase voltages, b) three-phase load currents |
| Figure 7.28: The performance of the proposed strategy under cascading failure: a) capacitor voltages of the upper arm in phase a , b) capacitor voltages of the lower arm in phase a |
| Figure 7.29: Action of the different controllers for leg a: a) The capacitor voltage estimation of SM <i>Vcu1</i> , b) Actual and reference waveforms of the circulating current |

| Figure 7.30: Action of the different controllers for leg a : a) averaging voltage reference, b) Modulation signal for the first SM of the upper arm136 |
|--|
| Figure A.1: Active and reactive power control156 |
| Figure B.1: The simulated 5 level MMC |
| Figure B.2: The submodule model157 |
| Figure B.3: The RLS programming block |
| Figure B.4: The ADALINE programming block160 |
| Figure B.5: The block of the proposed capacitor voltage algorithm: a) The averaging controller, b) The balancing controller, c) The PS-PWM modulation technique. 162 |
| Figure B.6: The block of the proposed FDU163 |
| Figure B.7: The block of the proposed FDU164 |
| Figure C.1: The block of the proposed FDU for one leg |
| Figure C.2: The block of submodule including the fault simulation unit172 |
| Figure C.3: The blocks of the proposed FTCU172 |
| Figure C.4: The blocks of the PS-PWM switching algorithm |
| Figure C.5: The Monitoring console179 |
| Figure C.6: The LABVIEW block of the ADALINE capacitor voltage estimation algorithm for one arm |
| Figure C.7: The LABVIEW block of the RLS capacitor voltage estimation algorithm for one arm |
| Figure C.8: The LABVIEW block of the averaging control for one leg181 |
| Figure C.9: The LABVIEW block of the balancing control for one submodule181 |
| Figure C.10: The LABVIEW block of the proposed FDU (for one submodule) |
| Figure C.11: The LABVIEW graphical user interface of the capacitor voltage estimation |
| Figure C.12: The LABVIEW graphical user interface of the capacitor voltage balancing control |
| Figure C.13: The LABVIEW graphical user interface of the proposed FDU183 |

List of Tables

| Table 2.1: Comparison between different switching algorithms used for M | IMCs24 |
|--|------------------|
| Table 3.1: Faults in submodule IGBTs | 34 |
| Table 3.2: Relationship between submodule output voltage and arm current open circuit faults. | nts during 35 |
| Table 3.3: Relationship between submodule output voltage and arm current short circuit faults. | nts during |
| Table 3.4: Faults in submodule capacitors | |
| Table 3.5: Comparison between common MMC fault detection techniques in literature | proposed 45 |
| Table 4.1: Summary of the MMC parameters of the first model | 64 |
| Table 4.2: Summary of the MMC parameters of the second model | 67 |

List of Abbreviations

| ADALINE | Adaptive linear neuron. |
|---------|--|
| ANN | Artificial neural network. |
| CSC | Current source converter. |
| DFIG | Doubly fed induction generator. |
| ESR | Equivalent series resistance. |
| FACTS | Flexible alternating current transmission systems. |
| FDU | Fault detection unit. |
| FPGA | Field programmable gate array. |
| FTC | Fault tolerant control. |
| FCU | Fault tolerant control unit. |
| HIL | Hardware in the loop. |
| HVAC | High voltage AC. |
| HVDC | High voltage direct current. |
| IGBT | Insulated gate bipolar transistor. |
| LCC | Line commutated converter. |
| LS-PWM | Level shifted pulse-width modulation. |
| LUT | Look-up table. |
| MMC | Modular multilevel converter. |
| MTTF | Mean time to failure. |
| NLM | Nearest level modulation. |
| PID | Proportional integral derivative. |
| PMSG | permanent magnet synchronous generator. |
| PS-PWM | Phase shifted pulse-width modulation. |
| PWM | Pulse-width modulation. |
| RLS | Recursive least squares. |
| SMO | Sliding mode observer. |
| SVM | Space vector modulation. |
| THD | Total harmonic distortion. |
| VSC | Voltage source converter. |
| | |

Chapter 1

Introduction

1.1 Background

Voltage source converters (VSC) are increasingly being used in the transmission and distribution of bulk power, due to their advantages over the line commutated converters (LCC). This includes the ability to control active and reactive power independently, supply weak or passive networks, and lower the footprint [1]. Although VSCs have many topologies, the modular multilevel converter (MMC) is considered to be the most suitable converter to be used in high voltage circuits. This is due to their features, including the ability to transmit electric power over high voltage levels directly, the improved quality of output power, and the high efficiency when compared to other multilevel converters [2]. Figure 1.1, which shows a 401-level MMC made by Siemens, illustrates another special feature of MMCs, which is the compact size. This gives a unique advantage for MMCs over other converter topologies when the space allowed for construction is limited.

The MMC is basically formed from a series connection of submodules, which can be either half bridge or full bridge. This is the main reason for calling it modular. Increasing the number of submodules means that the number of voltage levels are increased, which is directly translated into lower harmonic content and enhances the ability to generate higher voltages.



Figure 1.1: A 401-level MMC made by Siemens [3]: a) a picture of the MMC, b) a picture for the submodule.

As illustrated in Figure 1.2, the applications of MMCs have covered most high and medium voltage applications that need the utilisation of power converters. The applications of the MMC include high voltage direct current (HVDC) transmission, MV drives, flexible AC transmission systems (FACTS), and dynamic braking choppers [4]. Furthermore, some research has been recently conducted to utilise the MMC in special applications, such as railway electric traction systems, in which the MMC works as a medium voltage transformer-less converter, which is used to supply the traction motors [5], [6]. In addition, the MMC can applied be in the propulsion systems of electric ships [7], [8].



Figure 1.2: Applications of MMCs.

1.2 Motivation and Research Objectives

Today, DC to AC conversions are found in many applications in modern power systems, particularly in the transmission and distribution of electrical energy. Recently, DC/AC converters play an essential role in the interface of offshore wind energy with the utility grid. This is simply because the transmission along HVDC lines is much more efficient compared with the high voltage alternating current (HVAC) transmission over long distances [9].

The current source converter (CSC) was the first converter invented for this particular purpose. The CSC is formed from six thyristors connected in the form of a three-phase bridge connected to the DC supply. Although the CSC is considered very old technology, it is still used in the connection of the renewable energies to the grid in various modern projects [10]. This is simply because of the many advantages, such as the ability to convert high amounts of power, the low manufacturing price, and the high reliability [11]. However, it has some disadvantages and limitations that make it unsuitable for some applications [12], [13]. The main disadvantages are the necessity to have filtering components because of the low order harmonics, the impossibility of changing the current direction, and the inability to independently control reactive power. These limitations have motivated the researchers to invent VSC [13].

The two-level VSC has a simple configuration, which is similar to the CSC but with IGBT instead of the thyristors. This construction has changed the philosophy of switching since the IGBT is fully controllable during turning on and off. On some occasions, the switching device is formed from series-connected IGBTs to withstand high voltage stress [14]. One of the important advantages of the VSC is its ability to control active and reactive powers independently. Moreover, it can supply weak and passive networks. It also has a relatively low footprint compared to the CSC [15]. Despite having all of these advantages, the two-level VSC suffers from some drawbacks that can make it unsuitable for the application of grid interconnection. For example, having series-connected IGBTs increases the switching losses and makes it difficult to predict the transient response of the

switching device. It also is necessary to incorporate DC-link capacitors, which are very big and have a short life expectancy [16]. Moreover, the two-level VSC has no immunity against DC faults, as it fails to support the AC grid with active power [17]. Adding to these drawbacks, the two-level VSCs cannot exchange electricity over high voltage levels directly without using step-up transformers [18].

Due to the problems associated with LCC and VSC, researchers have been directed to multilevel converters (particularly MMCs). The modularity of the MMC offers a generous advantage, which is the ability to generate high voltage levels using a sufficient number of submodules. Although having a series connection of submodules with floating capacitor voltages brings the advantages of a staircase output waveform with low switching frequencies, it increases the level of complexity of the system from the control point of view, as all submodules must be charged to the same voltage level to assure the accurate and stable operation of the converter [19]. Moreover, in order to maintain the output voltage level of the converter, the capacitor voltages must be kept at a reference value [20], [21]. The process of keeping the capacitor voltages within a pre-defined reference is called capacitor voltage balancing control, which is achieved by a dedicated controller called the MMC inner controller [21]. The MMC inner controller should perform some tasks in parallel with the balancing of the voltages across the submodule capacitors. One of the important tasks is protecting the MMC from internal submodule faults, such as power electronic device faults and submodule capacitor faults. As demonstrated in Figure 1.3, which shows the general probability of occurrence for these faults in power converters, power semiconductor devices and capacitors have the highest probability of fault in the MMC with a percentage of 51% [22], [23]. Since the MMC normally works in critical applications, which always require high levels of reliability and supply security, its inner control system should be able to operate with a sustained fault (e.g., insulated gate bipolar transistors (IGBT) failure) until the next maintenance event [24], which means that the inner control system should have fault tolerant control (FTC) capability. This concept increases the reliability of the MMC, as it increases the availability of the converter, even in the case of having an internal fault in order to operate the converter under fail-safe operation conditions. It is important to mention that the process of FTC must be accompanied with a fault detection facility for the purpose

of maintaining safe management of MMC faults. This process not only detects the fault but also localises it to enable the FTC algorithm to isolate the faulty submodule.





Additionally, the fault diagnosis process has a real challenge to detect the internal DC fault very quickly because, in DC systems, the DC fault current rises to its peak value in a few milliseconds [25].

The volume of the research on fault detection and FTC systems for MMCs is insignificant when compared with the huge challenges surrounding this research area. Moreover, most of the fault detection and tolerant control techniques presented in literature are considered to be expensive, as they require either redundant components or extra sensors [26].

The aim of this thesis is to develop an improved MMC inner controller, which is responsible for balancing the voltages of the submodule capacitors with integrating an ultra-fast fault diagnosis and FTC capability against submodule faults, enabling the MMC to be safely operated in normal conditions as well as abnormal conditions. Furthermore, the developed technique should not need any extra sensors, special power circuits, or even redundant components to make it competitive in cost as well as in performance. The objectives of the thesis are the following:

- To carry out an intensive literature review in the area of capacitor voltage balancing, fault diagnosis, and tolerant control in MMC systems. This review should index the different faults that can affect the MMC submodules, showing the root cause for each fault and linking it with different techniques proposed in the past.
- To implement a new MMC inner controller, which contains the following subsystems:
 - Voltage balancing of submodule capacitors without the need to measure the actual voltages so that there is no more need for capacitor voltage sensors.
 - Fault diagnostic scheme, which is able to detect and localise submodule faults with effective performance in terms of accuracy and time without adding any extra sensing elements or special power circuits.
 - A FTC scheme, which controls the MMC in abnormal operating conditions when a submodule fault occurs. This scheme should achieve a fail-safe operation until the next shut down for maintenance with increased power quality without incorporating any redundant components.
- To test the developed fault diagnosis and tolerant control techniques on the MMC model to study the effectiveness of these methods by means of PSCAD simulation tool and hardware in the loop (HIL) testing techniques, incorporating a physical controller and a real-time digital simulator to prove the effectiveness of the proposed techniques physically as well as theoretically.

1.3 Thesis Outline

Chapter 2 investigates the structure, theory of operation, mathematical representation, and modulation of MMCs. Furthermore, the design considerations that should be met while selecting the ratings of MMC components are also investigated.

Chapter 3 presents a literature review over the main topics covered in the thesis. First, common capacitor voltage balancing techniques are investigated to characterise the advantages and disadvantages of each technique, aiming to determine the real problems associated with the process of capacitor voltage balancing. Then, the MMC submodule faults are mapped with the root cause and consequences associated with each fault. Finally, popular fault diagnosis and tolerant control strategies that are proposed in literature are also discussed.

In Chapter 4, a new capacitor voltage balancing strategy based on capacitor voltage estimation techniques is presented. Two capacitor voltage estimation techniques based on the adaptive linear neuron (ADALINE) and recursive least square (RLS) algorithms are introduced, and a comparison between the two techniques is also conducted to select the most suitable one to be incorporated into the proposed capacitor voltage balancing technique. Moreover, two simulation models are introduced for the purpose of testing the proposed capacitor voltage balancing techniques under several operation conditions.

Chapter 5 introduces a new fault detection technique based on capacitor voltage estimation algorithms, which is able to detect faults associated with MMC submodules. In addition, the chapter shows some simulation results that show the behaviour and performance of the proposed technique under different faulty conditions.

Chapter 6 presents the proposed FTC technique, which is based on a special sorting algorithm. Moreover, the reliability of the MMC is studied before and after the application of the proposed technique to assess the amount of reliability enhancement. Finally, simulation results of the entire control system are discussed.

Chapter 7 shows the validation of the proposed methods mentioned in Chapters 4, 5, and 6 by means of HIL of simulation through connecting a realtime digital simulator to a real-time controller. It also presents the practical assessment of the proposed MMC inner control when applied in a real application. The assessment mainly focuses on the amount of computational burden of the proposed techniques. Chapter 8 shows the conclusion of the thesis, focusing on the main contributions to the knowledge. Furthermore, it also presents some recommendations for future research.

1.4 List of Publications

1.4.1 Journal publications

- M. Abdelsalam, M. Marei, S. Tennakoon, and A. Griffiths, 'Capacitor voltage balancing strategy based on submodule capacitor voltage estimation for modular multilevel converters'. *CSEE Journal of Power and Energy Systems*, 2(1), pp. 65-73, 2016.
- M. Abdelsalam, M. Marei, and S. Tennakoon, 'An Integrated Control Strategy with Fault Detection and Tolerant Control Capability Based on Capacitor Voltage Estimation for Modular Multi-level'. *IEEE Transactions* on Industry Applications, Early access.
- M. Abdelsalam, M. Marei, S. Tennakoon and H. Diab 'A Fault Tolerant Based Control Strategy for Modular Multi-Level Converters'. *PLOS ONE Journal.* – Under review.
- M. Abdelsalam, M. Marei, and S. Tennakoon, 'A Centralized Control Strategy for Hybrid Modular Multi-Level Converters with Fault Detection and Tolerant Control Capabilities', *Journal of Power Electronics* – Under review.

1.4.2 Conference publications

- M. Abdelsalam, S. Tennakoon, A. L. Griffiths, and M. Marei, 'Investigation of submodule fault types of modular multi-level converters in HVDC networks'. *50th International Universities Power Engineering Conference* (UPEC), (pp. 1-6), IEEE, 2015.
- M. Abdelsalam, S. Tennakoon, A. L. Griffiths, and M. Marei, 'Fault detection, diagnosis and tolerant control of MMCs for HVDC Networks'. *The 5th IET International Conference on Renewable Power Generation.*
- M. Abdelsalam, M. Marei, H. Diab, S. Tennakoon, and A. L. Griffiths, 'Detection and Diagnosis of Submodule Faults for Modular Multilevel

Converters'. 51st International Universities Power Engineering Conference (UPEC).

- M. Abdelsalam, S. Tennakoon, H. Diab, and M. Marei, 'An ADALINE Based Capacitor Voltage Estimation Algorithm for MMCs', *The XIXth International Symposium on Electrical Apparatus and Technologies (SIELA* 2016).
- M. Moussa, M. Abdelsalam, and H. Diab, 'Enhanced Approach for Modelling and Simulation of Modular Multilevel Converter Based Multiterminal DC Grids', UKSim-AMSS 18th International Conference on Modelling & Simulation.

1.4.3 Co-authored conference publications

- H. Diab, M. Marei, M. Abdelsalam, S. Tennakoon, and C. Gould, 'Real Time Simulation of a Current Flow Controller for High Voltage DC Grids Applications', 51st International Universities Power Engineering Conference (UPEC).
- H. Diab, S. Tennakoon, M. Abdelsalam, and M. Marei, 'A Current Flow Control Apparatus for Meshed Multi-Terminal DC Grids', *The XIXth International Symposium on Electrical Apparatus and Technologies (SIELA* 2016).

Chapter 2

Modular Multilevel Converters: Structure, Operation, and Modulation

2.1 Introduction

This chapter presents a literature review on the structure, operation, and modulation techniques of MMCs. First, a brief introduction about multilevel converters is presented. Then, an intensive investigation on the MMC is conducted focusing on its structure, principles of operation, and mathematical model. In addition, a review of the different modulation techniques used for MMCs is also conducted. Finally, the design considerations of MMCs are presented, showing the selection procedure of arm inductors and submodule capacitors.

2.2 Multilevel Converters

The limitations of LCC and VSC (mentioned in Chapter 1) have directed researchers to invent multilevel converters, which are considered very promising for the application of transmission and distribution systems [27]. Unlike two-level VSCs, multilevel converters have advantages, such as the following [28]:

• The ability to transfer high power levels directly without the incorporation of interface power transformers.

- Improved efficiency due to the low switching frequency, which decreases the switching losses.
- Improved quality of output voltage due to the high number of voltage levels requiring no harmonic filters.

Features such as these have motivated researchers to implement a wide range of configurations for multilevel converters, as demonstrated in Figure 2.1. These are explained in the following subsections.



Figure 2.1: Types of multilevel converters.

2.2.1 Cascaded H-Bridge converter

A cascaded multilevel inverter is formed from a series connection of Hbridge submodules. As shown in Figure 2.2(a), the cascaded H-bridge converter is formed from the series connection of full bridges. Each full bridge is connected to a dedicated DC source [29]. The output voltage of each submodule can have three levels: $+V_{dc}$, $-V_{dc}$, or 0. The desired output voltage is achieved by the insertion and bypass of these bridges; thus, the output voltage is built up, as shown in Figure 2.2(b).



Figure 2.2: Cascaded H-Bridge converter: a) the structure, b) the waveform of output voltage [27].

This topology has a number of advantages, such as its ability to control the zero-sequence component of the current [27], [29]. Moreover, there is no need for balancing algorithms, as it is supplied from multi-phase secondary winding transformers [30]. However, it suffers from many drawbacks, such as the following [31]:

- They usually need a relatively high number of full bridges and isolated DC sources.
- It is not easy to have bi-directional operation of the cascaded inverter because of the usage of isolated DC sources.

2.2.2 Neutral-point clamped multilevel converters

The neutral-point clamped converter is formed from four switches per arm instead of two switches in the two-level converters. As shown in Figure 2.3, the midpoint of the switches is connected to two clamping diodes [32]–[35]. The operation concept is similar to that of the two-level VSC, however, instead of generating two voltages, the neutral-point clamped multilevel converter is able to generate several voltage levels according to the number of capacitors or zero [36].



Figure 2.3: Structure of neutral-point clamped multilevel converter [27].

This converter offers some advantages, such as the improved quality of output power, decrease in ratings of the power switches due to the use of the clamping diodes, and elimination of the need for isolated DC sources (which was a main drawback in cascaded H-bridge converters) [37]–[40].

However, this topology has some operating limitations, especially when the DC-link capacitor voltages are not balanced, as this will cause an elevated potential between the neutral and the ground, causing distortion in the output voltage. It is also not scalable. Moreover, clamping diodes are stressed with high voltages. Adding to all of this, the number of components in this converter is relatively high, which affects its reliability [41].

2.2.3 Flying capacitor multilevel converters

The flying capacitor converter has the same structure as the neutral-point clamped converter, where the clamping diodes are replaced with capacitors. Further,

another important difference is that the flying capacitor converter has a modular structure; this has solved the scalability problem of neutral-point clamped multilevel converters. As shown in Figure 2.4, every set of two switches connected to a capacitor forms a module [42].



Figure 2.4: The structure of flying capacitor multilevel converters [27].

This gives an advantage of having redundant switching states, which helps in the regulation of capacitor voltages. Moreover, the flying capacitors provide the converter with energy storage, which helps in the ride-through of faults and voltage sags [43], [44]. However, this topology suffers from two main drawbacks [45]:

- It is not possible to have a black start capability, as the capacitors need to be pre-charged.
- There is no equal distribution between the IGBT loading.

2.2.4 Hybrid multilevel converters

The idea of hybrid converters has been recently formed to overcome the limitations and disadvantages of the standard multilevel converters. Although hybrid converters have many topologies, the basic concept of them is employing a two level conversion stage as a phase controller to change the voltage polarity. On the other hand, the arm links are used to enhance the quality of the output power. This combination offers a reduced number of components which is translated into lower footprint. However, they have some limitations and challenges that are due to the non-standard topologies. One important disadvantage is the complexity of balancing the voltage of dc link capacitors or sub-module capacitors[46]–[51].

2.3 Modular Multilevel Converters

Although multilevel converters have many topologies that have been particularly made for DC/AC transmission, the MMC is considered the most suitable topology for high voltage applications, especially for power transmission and distribution, as it was particularly made to solve all problems associated with other topologies. Their advantages include the potential to eliminate harmonic filters with a sufficient number of voltage levels, DC-link capacitors, and low switching losses [4], [52].

The MMC consists of submodules connected in series forming a leg in each phase. The submodule can be a half bridge or a full bridge, and each submodule has a capacitor that buffers the energy from the DC to the AC side and vice versa; thus, the DC-link capacitor is not required. A high-speed bypass switch is added to the output port of the submodule to isolate the submodule in case of a fault.

As shown in Figure 2.5, each phase leg is divided into two arms, the upper and lower [4], [52]. Each arm has identical numbers of submodules to generate balanced voltages in the two arms of each phase. Inductors are installed in the arms to smooth and filter the currents.



Figure 2.5: Three-phase MMC topology and internal submodule.

2.3.1 MMC modes of operation

To understand the operation of the MMC, it is important to understand the operation of the submodule. The MMC half-bridge submodule has the four following operating modes, and the energy is transferred through them [53], [54]:

- Mode 1: S1 is closed; S2 is opened, and the arm has positive polarity. The current flows into the capacitor charging it, and the submodule capacitor is in circuit (Figure 2.6(a)).
- Mode 2: S1 is opened; S2 is closed, and the arm has positive polarity. The submodule capacitor is bypassed, and the current flows towards the next submodule, keeping the capacitor charge constant and then the submodule is bypassed (Figure 2.6(b)).
- Mode 3: S1 is closed; S2 is opened, and the arm has negative polarity. The capacitor starts to discharge and then the submodule is in circuit (Figure 2.6(c)).
- Mode 4: S1 is opened; S2 is closed, and the arm has negative polarity. The current flows towards the next module, keeping the capacitor charge constant and then the submodule is bypassed (Figure 2.6(d)).



Figure.2.6: Different operating modes of MMC submodules.

2.3.2 MMC mathematical model

The converter arms are represented by variable capacitors as shown in Figure 2.7, which are connected in series with the arm resistance and inductance. The following has been assumed:

- 1. The number of submodules per arm is infinite.
- 2. The switching frequency is infinite.

These assumptions will help in simplifying the analysis and in obtaining insight into the fundamental behaviour of the circuit, as the output voltage can be considered to be sinusoidal and to be the ideal voltage balance between the arms.



Figure 2.7: MMC average circuit diagram.

The arm capacitance insertion is represented with a modulation value m_u for the upper arm and m_l for the lower arm, while m_x is varied from 0 to 1. When $m_x = 0$, this indicates that all submodules are bypassed, and when $m_x = 1$, this indicates that all submodules are inserted. The value $\sum_{i=1}^{N} v_{cxi}$ is the sum of arm capacitor voltages [55]–[58]. The arm voltages are given by:

$$v_u(t) = m_u \sum_{i=1}^N v_{cui},$$
 (2.1)

$$v_l(t) = m_l \sum_{i=1}^N v_{cli}.$$
 (2.2)

The effective arm capacitance is given by:

$$C_{effu} = \frac{C_{SM}}{N.m_u(t)},\tag{2.3}$$

$$C_{effl} = \frac{c_{SM}}{N.m_l(t)},\tag{2.4}$$

where C_{SM} is the capacitance of one submodule and N is the number of submodules per arm.

If the arm currents are $i_u(t)$ and $i_l(t)$ for the upper and lower arms, respectively, the governing differential equation is:

$$\frac{d\sum_{i=1}^{N} v_{cui}}{dt} = \frac{i_u(t)}{c_{effu}},\tag{2.5}$$

$$\frac{d\sum_{l=1}^{N} v_{cli}}{dt} = \frac{i_l(t)}{c_{effl}}.$$
(2.6)

The output ac current i_{ac} is given by:

$$i_{ac} = i_u - i_l. \tag{2.7}$$

The current i_{diff} is the differential current that flows in the phase leg; this current receives an equal contribution from the upper and lower arms because it is assumed that the arm voltages are balanced. In terms of this current, the currents of the upper and lower arms are expressed by:

$$i_u = i_{diff} + \frac{i_{ac}}{2},\tag{2.8}$$

$$i_l = i_{diff} - \frac{i_{ac}}{2}.$$
(2.9)

Adding (2.5) and (2.6) gives:

$$i_{diff} = \frac{i_u + i_l}{2}.\tag{2.10}$$

It is important to mention that the differential current does not contribute to the AC output current. However, it circulates inside the converter, increasing its losses and affecting its performance [4], [59].

The capacitor voltage dynamics can be expressed for the upper and lower arms by:

$$\frac{d\sum_{i=1}^{N} v_{cui}}{dt} = \frac{N.m_u.i_u}{C_{SM}},$$
(2.11)

$$\frac{d\sum_{i=1}^{N} v_{cli}}{dt} = \frac{N.m_l.i_l}{C_{SM}}.$$
 (2.12)
Applying the Kirchhoff voltage law to the circuit shown in Figure 2.7, the output AC voltage can be expressed as:

$$v_o = \frac{V_{dc}}{2} - R_{arm} i_u - L_{arm} \frac{di_u}{dt} - m_u \sum_{i=1}^N v_{cui}, \qquad (2.13)$$

$$v_o = -\frac{v_{dc}}{2} + R_{arm}i_l + L_{arm}\frac{di_l}{dt} + m_l \sum_{i=1}^N v_{cli.}$$
(2.14)

Subtracting (2.13) from (2.14), the differential of i_{diff} can be expressed as:

$$\frac{di_{diff}}{dt} = \frac{V_{dc} - R_{arm} i_{diff} - m_l \sum_{i=1}^N v_{cli} - m_u \sum_{i=1}^N v_{cui}}{2L_{arm}}.$$
 (2.15)

From (2.13) and (2.14) and substituting for the currents from (2.7) in (2.15), the differential equations for the phase leg are given by:

$$\frac{d}{dt} \begin{bmatrix} i_{diff} \\ \sum_{i=1}^{N} v_{cui} \\ \sum_{i=1}^{N} v_{cli} \end{bmatrix} = \begin{bmatrix} -\frac{R_{arm}}{2L_{arm}} & -\frac{m_u}{2L_{arm}} & -\frac{m_l}{2L_{arm}} \\ \frac{Nm_u}{C_{SM}} & 0 & 0 \\ +\frac{Nm_l}{C_{SM}} & 0 & 0 \end{bmatrix} \begin{bmatrix} i_{diff} \\ \sum_{i=1}^{N} v_{cui} \\ \sum_{i=1}^{N} v_{cli} \end{bmatrix} + \begin{bmatrix} \frac{V_{dc}}{2L_{arm}} \\ \frac{Nm_u i_{ac}}{2C_{SM}} \\ -\frac{Nm_l i_{ac}}{2C_{SM}} \end{bmatrix}$$
(2.16)

As shown in (2.13) and (2.14), the output voltage, e_v does not depend on the differential current, i_{diff} . Moreover, the value of the differential current depends only on the voltage of the inserted submodules, as given in (2.16). This means that this current can be decreased without affecting the output power.

2.4 Modulation Techniques of the MMC

The modulation techniques used for MMCs are considered extremely important since they are responsible for inserting and bypassing submodules, which means that they control the whole performance of the MMC. Many researchers have developed various types of switching algorithms for MMCs; however, MMC modulation techniques can be divided into three main categories: reference signalbased, carrier-based, and nearest level modulation. The following subsections illustrate the main idea behind each technique. Then, they show the selected technique to be used in the modelled MMC, explaining the reasons for this selection.

2.4.1 Space vector modulation

Space vector modulation (SVM) is commonly used in two-level VSCs. However, some researchers have proposed it for the switching of MMCs due to its attractive advantages, such as [60], [61] the following:

- i) ability to be implemented digitally,
- ii) reduction of the total harmonic distortion (THD) of output voltage and current waveforms,
- iii) optimisation of switching sequences.

Moreover, SVM for two-level VSC is extended to suit the *N* level MMC [61]. As shown in Figure 2.8, the possible switching states for an *N* level MMC is 2^{N-1} . These states are represented in a 3 x N matrix called H_{abc} :

$$H_{abc} = \begin{bmatrix} h_{a1} & h_{a2} & \dots & h_{aN} \\ h_{b1} & h_{b2} & \dots & h_{bN} \\ h_{c1} & h_{c2} & \dots & h_{cN} \end{bmatrix}$$
(2.17)



Figure 2.8: Switching plan of space vector for an N level MMC [61].

The reference voltage is generated by multiplying the switching-state matrix, H_{abc} by the submodule capacitor voltages:

$$V_{ref,abc} = H_{abc} x [v_{c1} \quad v_{c2} \quad \dots \quad v_{cN}]$$
(2.18)

Similar to the two-level converters, the reference vector is obtained from the nearest stationary vectors, which form a triangle, as shown in Figure 2.9. The dwell times of the stationary vectors are calculated from equations (2.19) and (2.20):

$$\overrightarrow{V_{ref}} T_s = \overrightarrow{V_{1,N}} t_{1,N} + \overrightarrow{V_{2,N}} t_{2,N} + \overrightarrow{V_{3,N}} t_{3,N}, \qquad (2.19)$$

$$T_s = t_{1,N} + t_{2,N} + t_{3,N}.$$
 (2.20)



Figure 2.9: The triangle of stationary vectors [61].

It is obvious that the implementation process of the SVM technique is very complex due to the high number of switching states. For example, for a 401-level MMC, the algorithm needs to select from 5 x 10¹²⁰. This makes it not recommended for many MMC applications, especially those that have a high number of voltage levels.

2.4.2 Nearest level modulation

The nearest level modulation (NLM) is a simple technique for the direct switching of submodules based on their capacitor voltages and their closeness to the reference signal. First, the required number of voltage levels to be included is determined according to equation (2.21):

$$n_{ON\,x} = round \left[N\left(\frac{1}{2} \pm \frac{V_{ref}(t)}{V_{dc}}\right) \right].$$
(2.21)

The number of bypassed modules is calculated from equation (2.22):

$$n_{OFF x} = N - n_{ON x} + 1. (2.22)$$

The average value of the capacitor voltages of the inserted submodules will equal the desired reference of the output voltage, as demonstrated in Figure 2.10 [57], [58].



Figure 2.10: The NLM technique [45].

The main advantage of this method is its simplicity, as it can be implemented easily on industrial computers due to the small number of calculations. However, this switching algorithm suffers from a main drawback, which is its relatively high harmonic content that makes it unsuitable for MMCs with a low or medium number of voltage levels [57].

2.4.3 Carrier-based pulse-width modulation

Many carrier-based pulse-width modulation (PWM) based switching algorithms have been proposed in literature [62], [63]. However, there are two major techniques that are commonly used for switching MMCs. The first is the level shifted PWM (LS-PWM). In this technique, each level has a dedicated triangular waveform, as shown in Figure 2.11. The triangular waveforms are displaced from each other by a pre-defined time interval. A sinusoidal reference voltage is compared with the carrier waves to generate the firing signals for the submodules.



Figure 2.11: The LS-PWM technique.

The second technique is the phase-shifted pulse-width modulation (PS-PWM), which is one of the well-known carrier-based techniques. As illustrated in Figure 2.12, each submodule has a dedicated triangular carrier waveform with the same magnitude but with a different phase shift.



Figure 2.12: The PS-PWM technique.

The switching signals for a submodule result from comparing its corresponding carrier wave with the sinusoidal voltage reference signal v_o^* . The phase shift between each consecutive carrier signals is given by:

$$\theta = \frac{360}{N-1}.$$
 (2.23)

Table 2.1 presents a comparison between the above mentioned techniques from the point of view of complexity, output power quality and the natural balancing of capacitor voltages.

| Switching algorithm | Ease of implementation | Quality of output power | Natural balancing |
|-----------------------------|------------------------|---|----------------------|
| Space vector modulation | Very complex | High | Good |
| Nearest level modulation | Simple | Low (particularly for low number of levels) | Bad |
| Level shifted PWM | Moderate | Moderate | Worst |
| Phase shifted PWM | Moderate | Moderate | Acceptable |

Table 2.1: Comparison between different switching algorithms used for MMCs.

According to the table, the PS-PWM method is selected to be used in the MMC model in this thesis due to the numerous advantages, such as the ease of implementation, the reduced losses due to the reduced switching frequency, and the stable performance during dynamic changes.

2.5 Design Considerations of MMCs

The following subsections show the design considerations of the MMC and the selection process of its parameters, particularly the procedure of selecting the submodule capacitor and arm inductor. It is important to mention that the sizing of these elements is very important for proper operation of the MMC.

2.5.1 Submodule capacitance

The capacitance is calculated based on the energy variation inside the capacitor. Assume that the current passes through the capacitor is i_c and that the voltage across it is v_c . Then, the energy variation inside one capacitor during a period between t_1 and t_2 is:

$$\int_{t_1}^{t_2} v_c \, i_c \, dt = \frac{c \, (v_2^2 - v_1^2)}{2}. \tag{2.24}$$

The capacitor voltage change is from V_1 and V_2 . Since the MMC is formed from a series connection of submodules, (2.24) can be applied for the whole arm instead of one submodule. The energy variation inside the upper arm (since both arms are identical) is:

$$\Delta e_u = \int v_u \ i_u \, dt. \tag{2.25}$$

Ignoring the power losses and the arm inductor voltages and assuming that output voltage v_o has a pure sinusoidal waveform, the voltage of the upper arm given in (2.26) can be rewritten as:

$$v_u = \frac{v_{dc}}{2} - v_o. (2.26)$$

The steady state representation of the output voltage and output current is given by:

$$v_o = \hat{V}_o \cos(\omega t), \qquad (2.27)$$

$$i_o = \hat{I}_o \cos(\omega t - \varphi), \qquad (2.28)$$

where \hat{V}_o and \hat{I}_o are the maximum values of the output voltage and output current, respectively. Since the power losses are ignored, the active output power can be expressed as:

$$P = \frac{\widehat{V_0}\widehat{l_0}\cos(\varphi)}{2}.$$
 (2.29)

With the aid of (2.25) and (2.29), the energy variation of the arm as a function of the output active power can be expressed as [64]:

$$\Delta e_u = \frac{P\left(\sin(\omega t - \varphi)\right)}{\omega\cos(\varphi)} - \frac{\sin(2\omega t - \varphi)}{4} - \frac{\cos(\varphi)\sin(\omega t)}{2}.$$
 (2.30)

Assuming that the allowable percentage of ripple is $k_{ripple max}$, according to (2.30), the voltage will vary between V_c and $(1+k_{ripple max}) V_c$. The cell capacitance must be chosen such that the maximum value of the energy variation, $\Delta e_{p max}$, is smaller than this variation as per (2.31) [64]:

$$\Delta e_{p \max} < \frac{N C_{SM} V_c^2 \left(\left(1 + k_{ripple \max} \right)^2 - 1 \right)}{2}.$$
(2.31)

After rearranging (2.31), the value of the capacitor, C_{SM} , must satisfy the following:

$$C_{SM} > \frac{2 \Delta e_{p max}}{N V_c^2 \left(1 + k_{ripple max}\right)^2}.$$
(2.32)

2.5.2 Arm inductance

As mentioned before, arm inductors play a valuable role in the MMC, as they are responsible for limiting the differential and transient currents that may develop during DC faults [65]. They also help in the correction of the total reactance of the system. Thus, the size of arm inductors should be carefully selected to limit DC fault currents to a value that is by far lower than the rating of the MMC IGBT switches [66]. To facilitate the estimation of differential current ripples, the maximum voltage across the arm is assumed to be equal to the submodule capacitor voltage. Additionally, the average value for changing the switching states is assumed to be $\frac{T_s}{2N}$ where T_s is the switching cycle of the IGBT. In this case, the peak-to-peak ripples imposed in the differential current can be calculated according to:

$$\Delta i_{diff} = \frac{v_c T_s}{4 N L_{arm}}.$$
(2.33)

Each arm inductor is subjected to half of the peak-to-peak ripples. In this case, after rearranging (2.33), the arm inductor should be selected according to [66]:

$$L_{arm} > \frac{V_c T_s}{8 N \Delta i_{diff}}.$$
 (2.34)

2.6 Chapter Summary

This chapter has presented a detailed review of different types of multilevel converters used for DC/AC conversion with particular focus on the MMC, including its structure, operation, and control concepts. Furthermore, it presented the popular switching algorithms used for the firing of MMC submodules, mentioning the selected switching algorithm used in the simulation. Then, the design considerations of the MMC were presented, focusing on the procedure of selecting the size of submodule capacitors and arm inductors.

Chapter 3

Review of Capacitor Voltage Balancing, Fault Detection, and Fault Tolerant Control Techniques of Modular Multilevel Converters

3.1 Introduction

This chapter presents a literature review on the main topics of the thesis. First, a review of the different techniques of capacitor voltage balancing in MMCs is presented. Then, the MMC submodule faults are investigated, explaining the root cause of each fault consequence. In addition, the concepts of fault detection and tolerant control are discussed with a comprehensive investigation of common MMC fault diagnosis and tolerant control strategies.

3.2 Capacitor Voltage Balancing of MMC Submodules

Having series-connected submodules with unequal floating capacitor voltages cannot be accepted, as this condition will increase the differential current and thus, increasing the ripple content in the submodule capacitor voltages which means that the harmonics inside the DC-link voltage will be increased [19]. Thus,

capacitor voltages should follow a pre-defined reference signal. Moreover, these voltages should be balanced to lower the differential current and converter losses [20]. Many strategies have been proposed in order to balance the voltages of the submodule capacitors by forcing the capacitor voltages to follow a desired value for the purpose of reducing ripples in these voltages, eliminating the AC components that are found in the differential current and enhancing the quality of the output power [67]. The following subsections show the most popular control strategies that are used in the industry.

3.2.1 Capacitor voltage balancing based on energy control

In this method, the capacitor voltages are balanced by controlling the total energy stored in the leg and the differential energy between the upper and lower arms [68]. As shown in Figure 3.1, the first step in this method is the measurement of submodule capacitor voltages, and then the energy stored in each arm is calculated with the aid of these voltages.



Figure 3.1: Block diagram of capacitor voltage balancing based on energy control.

A reference signal for the desired energy level is received as input to the controller. This reference is calculated based on the reference value of the output voltage, v_o^* . By knowing the energy stored in the arms and the reference value of the energy, the modulating signals (m_u and m_l) for the upper and lower arms are

generated. These modulating signals are then passed through a suitable switching algorithm to insert or bypass different submodules [68].

The performance of this method is good, as it always succeeds in balancing the voltages, even in fast dynamic actions; however, it suffers from some drawbacks, such as a distortion in the output current because of the continuous inserting and bypassing of submodules [68]. Moreover, this method is complex due to the large number of calculations and high number of control loops. It also needs continuous measurements of a large number of signals, such as submodule capacitor voltages and arm currents.

3.2.2 Capacitor voltage balancing based on distributed control

This method was particularly developed to solve some of the problems associated with the energy balancing control, particularly the distorted output power and the complex calculations and control loops [20]. As illustrated in Figure 3.2, the distributed control technique consists of two main parts.



Figure 3.2: Block diagram of capacitor voltage balancing on based distributed control.

The first is the average control of the submodule capacitors, while the second is the balancing control. The average control uses a proportional-integral-derivative (PID) controller to make the average voltage of the whole leg reach the desired reference voltage, v_c^* . The balancing control has N control loops; each loop utilises a P controller for each submodule to enable its capacitor voltage to reach the desired voltage. Both averaging and balancing control loops need the measurement of the submodule capacitors. The performance of this method is more attractive compared to the energy control, as it is very fast in tracking the reference voltages and thus balancing the voltages, and simultaneously, the distortion in the output power has significantly enhanced [20]. The main drawback of this method is the high content of differential current, as it remains at 70% of its original value. Moreover, it is considered complex when applied to MMC with a high number of voltage levels since each leg must be equipped by (2N + 1) P/PI control loops [69]. This method will be explained in detail later, as it is a part of the proposed control.

3.2.3 Capacitor voltage balancing based on reduced switching frequency modulation

This technique was developed to provide a simple method for balancing the capacitor voltages instead of depending on complex techniques, as illustrated in the two previous methods.

In this method, first, the arm current is observed to determine whether it is charging or discharging the submodules. In the case of charging the capacitors, the algorithm calculates the number of submodules that need to be inserted on N_{new} based on the measured capacitor voltages. Then, it calculates the difference between this number and the old number of submodules that is already inserted ($\Delta N_{on} =$ $N_{on} - N_{on_{old}}$). If the ΔN_{on} is zero, then the switching is kept as it is and nothing changes. However, if the ΔN_{on} is greater than zero, the arm current is checked to determine whether it is charging or recharging the arm capacitors. If the arm current i_{arm} is higher than zero, the algorithm inserts ΔN_{on} submodules that have the lowest voltages, while the algorithm bypasses ΔN_{on} submodules that have the highest voltage if the i_{arm} is less than zero. Finally, if the ΔN_{on} submodules that have the highest have the highest voltages, while the algorithm bypasses ΔN_{on} submodules that have the highest voltage if the i_{arm} is less than zero. Finally, if the ΔN_{on} submodules that have have the highest voltages, while the algorithm bypasses ΔN_{on} submodules that have lowest voltages if the i_{arm} is less than zero [70]. Figure 3.3 shows the flow chart of the reduced switching frequency technique.



Figure 3.3: Block diagram of capacitor voltage balancing based on reduced switching frequency [70].

It is clear that this method enhances the performance of the MMC since the converter losses are lowered due to the relatively low switching frequency. Moreover, there are no complex calculations or sophisticated PID control loops, which makes it simpler than other techniques. However, the ripples of capacitor voltages are increased due to the increased voltage bandwidth, which means that the differential current will be increased. Another drawback is that it still needs the measurement of each capacitor voltage.

3.2.4 Capacitor voltage balancing based on sorting algorithms

Like the previous method, this method aims to reduce the switching frequency of the submodule IGBTs through applying a simple sorting algorithm while considering reducing the ripple content. The algorithm depends on two sorting structures and a comparator that selects the submodules that need to be inserted. The first stage of applying the algorithm is the sorting stage. From its name, as shown in Figure 3.4, the role of this stage is to sort the submodules in a pre-defined order to be ready for either insertion or disconnection [71].



Figure 3.4: Block diagram of the conventional sorting algorithm based capacitor voltage balancing algorithm [71].

First, the capacitor voltages are multiplied by the sign of the arm current. The modified submodule capacitor voltage is given by:

$$v'_{cxj} = (-sign\,i_x)\,v_{cxj},\tag{3.1}$$

where x denotes the location of the arm, whether it is upper or lower, and j is the number of submodules. The modified capacitor voltages, v'_{cxj} , are then sorted in descending order. The algorithm scans the sorted list of the submodules to prepare them for activation. Submodules that will be activated are those with an activation order below or equal to the number of required submodules, n_{req} .

The direct application of this algorithm fails to reduce the switching losses, as it creates many additional switching states, which increase the switching frequency. Thus, a modification to this algorithm has to be made to restrict the number of connected submodules. This restriction is achieved by multiplying the switching states by a factor, ΔK , which is varied between 0 and 1, to enable the sorting algorithm to distinguish between the voltages of the inserted and bypassed submodules [71].

The block diagram of the modified algorithm is shown in Figure 3.5. In this case, the sorting stage will see a virtual voltage with a value of:

$$v_{cxj}^{"} = v_{cxj}^{\prime} + \Delta K. \ S_{xj}.$$
 (3.2)

This means that, when submodules are connected, the virtual voltage will be:

$$v_{cxj}^{"} = v_{cxj}^{\prime} + \Delta K, \qquad (3.3)$$

while, for bypassed submodules, the virtual voltage will be:

$$v_{cxj}^{"} = v_{cxj}^{\prime}. \tag{3.4}$$



Figure 3.5: Block diagram of the modified sorting algorithm based capacitor voltage balancing algorithm [71].

The selection of the ΔK should be carefully chosen for the necessary separation gap between the inserted and bypassed capacitor voltages. This method has a robust performance in terms of balancing the capacitor voltages, reducing the switching losses, and lowering the ripple content. However, it requires the measurements of all submodule capacitor voltages and the arm current, which increases the cost of the strategy. Moreover, applying the submodule restriction concept adds some complexity to the algorithm.

All capacitor voltage balancing techniques discussed in this chapter have a common drawback, which is their need for capacitor voltage measurements. These measurements not only add a large number of sensors but also require a high number of communication channels to transmit these signals. This has prompted the author to develop a new capacitor voltage balancing technique (presented in Chapter 4) that does not depend on the measurement of submodule capacitor voltages.

3.3 The MMC Submodule Faults

According to the literature, there are different types of submodule faults that can strike MMCs and affect their operation [26]. Since MMCs are used in important applications and are a big investment, it is extremely important to detect these faults very quickly to protect the whole system from sudden failures. For this particular purpose, the submodule faults are investigated in the following subsections. Moreover, popular fault detection techniques are discussed to discover the advantages and disadvantages of each method and to highlight what is needed in the process of fault detection of MMCs.

3.3.1 Faults in power electronic devices

The IGBT operation is always linked to the presence of freewheeling diodes. This is simply because the IGBT cannot function correctly without having a freewheeling diode dissipating the excess power due to the IGBT turn-off action. All of faults inside the power electronic device faults always depend on the semiconductor physics. Power electronic device faults are either open circuit faults or short circuit faults.

Table 3.1 shows the main causes of each fault type and their corresponding consequences [72]–[74].

| Fault Type | Failure Mechanism | Consequences | |
|---------------------|-----------------------------|---------------------------------|--|
| Open circuit | Bond wire rupture | 1) Pulsating current | |
| | Gate driver failure | 2) Distortion in output current | |
| | | and voltage | |
| Short circuit | High voltage breakdown | Overheating in a few cells at | |
| | Static/ dynamic latch up (a | first, spreading to other cells | |
| | low impedance path between | and leading to complete failure | |
| | the two regions) | | |
| | Energy shocks | | |

Table 3.1: Faults in submodule IGBTs.

In case of having an open circuit fault in the upper submodule switch, the negative capacitor current is blocked. As a result, the only path for this current is the lower freewheeling diode.

The same case occurs for the lower switch of the submodule, but the capacitor current is positive. Table 3.2 illustrates the behaviour of arm currents and submodule voltages in case an open circuit fault.

| - · · | The sign of the arm current | The submodule output voltage | | | |
|----------------------------|-----------------------------------|------------------------------|---------------------------------------|---------------------------------------|--|
| The state of the submodule | | Normal operation | Upper switch open circuit fault | Lower switch open circuit fault | |
| Bynassed | +ve | V_c | V_c | V_c | |
| Dypassed | -ve | V _c | 0 | V _c | |
| Inserted | +ve | 0 | 0 | V_c | |
| morred | -ve | 0 | 0 | 0 | |

Table 3.2: Relationship between submodule output voltage and arm currents during open circuit faults.

Regarding switch short circuit faults, the behaviour is different, as the output voltage will equal the capacitor voltage when the upper switch is short circuited. When the short circuit is applied to the lower switch, the submodule output voltage is equal to zero. Table 3.3 describes the relationship between the arm current and submodule output voltage during switch short circuit faults.

Table 3.3: Relationship between submodule output voltage and arm currents during short circuit faults.

| The state of the | The sign of | The submodule output voltage | | | |
|------------------|-------------|------------------------------|----------------|----------------|--|
| submodule | the arm | Normal | Upper switch | h Lower switch | |
| | current | operation | open circuit | open circuit | |
| | | | fault | fault | |
| Bypassed | +ve | V_c | V _c | 0 | |
| | -ve | V _c | V_c | 0 | |
| Inserted | +ve | 0 | V _c | 0 | |
| | -ve | 0 | V_c | 0 | |

It is important to mention that in case of open circuit faults, there is no threat on the safety of the converter or the whole system since the consequences are related to the power quality so there is no real need to shut off the converter. On the other hand, if the submodule is subjected to a short circuit fault in the power electronic device, it is a must to disconnect the submodule immediately [26].

3.3.2 Faults in submodule capacitors

The submodule capacitor plays an essential role in the operation of the MMC; however, it may suffer from a lot of problems that can severely affect their operation, such as sensitivity to temperature and frequency. One of the most critical problems is the wear-out degradation failure. The main consequence of this failure is the increased equivalent series resistance (ESR). This ESR increase affects the rate of charge and discharge and then the whole operation of the capacitor [75]–[77]. In addition to the wear-out failure, the submodule capacitor may be exposed to open circuit and short circuit failures, as shown in Table 3.4.

| Fault Type | Failure Mechanism | Consequences |
|---------------------|----------------------------|---------------------------------------|
| Open circuit | Leakage of electrolyte | 1) Increase in the capacitor voltage, |
| | dielectric material | current, and temperature. |
| | Loose connections of | |
| | terminals | 2) Mechanical vibration |
| Short circuit | Dielectric breakdown | Increase in the capacitor voltage, |
| | | current, and temperature |
| Fault in | Change in the electrolytic | Increase in the capacitor voltage, |
| capacitor | capacitor structure and | current, and temperature |
| structure | capacitance value | |

Table 3.4: Faults in submodule capacitors.

The capacitor short circuit fault is the only fault type that needs the converter to be shut off, while, in all other fault types, it is not required to shut off the converter [75]. The detection of capacitor faults is not considered in this thesis, as it is a different topic that depends on the monitoring of the capacitor life time.

3.4 The Concept of Fault Detection

The electric fault is defined by any failure associated with a device or with components inside the electric system [78], [79]. The fault detection scheme is responsible for continuous monitoring of the system for the purpose of detecting different types of faults, locating the fault, and diagnosing the cause of the fault. As per Figure 3.6, fault detection strategies can be divided into two main types.



Figure 3.6: Types of fault detection techniques.

The first is the hardware-based fault detection and the second is the softwarebased fault detection.

3.4.1 Hardware-based fault detection

Hardware-based fault detection strategies depend on adding extra sensors or special power circuits that are dedicated to monitoring the whole system. They can easily detect the fault by comparing the measured signals with pre-defined thresholds. When the measured signals exceed these thresholds, the fault detection unit (FDU) considers this a fault incident; thus, this fault can be easily located [23], [26]. Hardware-based FDUs are commonly used in many applications, and they are famous for their accuracy. However, they have some disadvantages that can be summarised in the following points:

- Adding extra system components for the system will increase the cost.
- The system will be more complicated, and more space will be consumed.
- The increased number of components will reduce the reliability of the entire system.

3.4.2 Software-based fault detection

The software-based FDUs depend on figuring out the different states of the system through the estimate of these states, comparing it with their corresponding actual states [80], [81]. The system states can be identified from the mathematical model of the system. In this case, the FDU will be a model-based unit or can be obtained from the analysis of historical fault data, in which case, the FDU will be knowledge based [80].

Regarding the model-based fault detection strategies, the first step in the design of such techniques is building the mathematical model of the system that needs to be protected. Then, there are two main directions. The first is to utilise an observer and compare actual measured signals with their corresponding observed signals. A fault is detected when a large error between the actual signals and their observation is created. The second route is estimating some signals and comparing them with the actual signals. If there is an inconsistency between the estimation and the actual signals, this is considered to be a fault event [81]. These methods are very attractive for researchers since they are accurate and do not need any extra sensors or additional circuits. However, there is a real challenge in these methods, which is the guarantee of the accuracy of the estimation or observation methods to achieve the selectivity of the FDU and at the same time prevent false fault detection.

Regarding the knowledge-based methods, these methods are attractive in very limited cases where there is a real difficulty in emphasising the mathematical model of the system. Thus, the FDU is designed based on historical fault data, and then a suitable artificial intelligent technique is trained using this data to be able to detect similar faults in the future [26], [82], [83]. The main problems associated with the knowledge-based methods are:

- They require a large number of training points, which consume a lot of effort.
- Their accuracy is not guaranteed if some of the system inputs are changed.

3.5 Fault Detection Techniques in MMCs

There has not been much research conducted in the area of fault detection in MMCs. Most of the proposed methods in literature are classified as hardware techniques. The following subsections present the popular fault detection methods proposed in the past.

3.5.1 IGBT fault detection using the rate of change of the collector current

This method is a hardware-based technique that is dedicated to detecting faults inside IGBTs. It depends on measuring the rate of change of the collector current $\frac{di_c}{dt}$ using the circuit shown in Figure 3.7. This value is compared with a predefined threshold, and if it exceeds this threshold, the FDU sends a signal indicating a fault incident [84].



Figure 3.7: The $\frac{di_c}{dt}$ measuring circuit [84].

This method is very accurate, as each IGBT has its own monitoring circuit. However, the cost of this method is extremely high since each IGBT has a sophisticated monitoring circuit [84]. Moreover, it is not capable of differentiating between open circuit and short circuit faults.

3.5.2 IGBT fault detection using the behaviour of the gate voltage

This method is also a hardware-based technique that depends on the charge rate of the gate voltage. As shown in Figure 3.8, the energy of the gate voltage is determined based on the value of the gate voltage and on monitoring the turn-on period. The calculated value of the energy is compared with different thresholds to classify whether the IGBT is healthy or in a fault condition [85].



Figure 3.8: Gate voltage-based fault detection circuit [85].

The important advantage of this method over the previously mentioned method is its ability to differentiate between open circuit and short circuit faults, as each fault has its dedicated threshold. However, this method still increases the cost of the FDU because of the need to have a dedicated measured signal for each IGBT [85]. Moreover, the fault detection circuit is very complex, and it is difficult to be integrated inside the IGBT.

3.5.3 Submodule fault detection using the measurements of submodule output voltage

All abovementioned fault detection techniques suffer from a common disadvantage, which is the complexity of the hardware circuit. This method, which is classified as a hardware method, was designed to eliminate this drawback through the implementation of a simple hardware circuit. The fault detection technique depends on a new measurement location. Instead of measuring the capacitor voltage, this fault detection strategy depends on the measurement of submodule output voltage [86]. Additionally, to avoid adding extra measurements, the submodule capacitor voltage is derived from the submodule output voltage, as demonstrated in Figure 3.9.

The capacitor voltage can be estimated by checking whether the state of the submodule is inserted or bypassed. If the submodule is inserted, then the capacitor voltage is equal to the submodule output voltage. However, if the cell is bypassed, the capacitor voltage will remain as it is.



Figure 3.9: Block diagram of the submodule output voltage and derivation of capacitor voltage [86].

As shown in Figure 3.10, the fault detection strategy depends on the continuous comparison between the submodule output voltage, V_{om} and V_{min} , which is a value lower than all other capacitor voltages. The output of this comparison and the submodule switching signal are passed through an exclusive disjunction or 'exclusive or' (XOR). In normal operation, the submodule output voltage, V_{om} , should have 0 V, providing that the submodule is bypassed, and in this case, the output of the XOR should be low.



Figure 3.10: Flowchart of submodule output voltage-based fault detection [86].

While in the case of a fault, the output of the XOR will be high, as the voltage across the lower IGBT will no longer remain zero. Thus, the fault condition can be easily detected. This method has many advantages; it can detect both open circuit and short circuit faults and is very sensitive and fast. Nonetheless, it suffers from many drawbacks, which can be summarised in the following points [86]:

- This method needs the continuous measurement of submodule output voltage instead of capacitor voltage, which is not standard and requires extra measurement circuits.
- The capacitor voltages, which are mandatory for the control of the MMC, are derived from the submodule output voltage; this is not accurate and can form a threat for the MMC operation.
- The high sensitivity of this method can lead to malfunction since the decision is based on one factor, which is the submodule output voltage.

3.5.4 IGBT fault detection using artificial neural networks

This method is a knowledge-based software technique, which depends on the historical data of the MMC output voltage and current because they can be affected if any IGBT has either an open circuit or short circuit. If one of the IGBTs has an open circuit fault, the output voltage of the MMC is then distorted and decreased by a considerable value. On the other hand, if any IGBT has a short circuit fault, then the output current is increased by about 1.5 its normal value. As shown in Figure 3.11, an artificial neural network (ANN) is trained with the aid of an output voltage and output current histogram through which its elements have been captured in many normal and abnormal operating conditions [26], [82].

This technique also can provide the control system with the fault location as well. Although this method is a software-based method and does not need any extra physical components, it suffers from some critical disadvantages as mentioned before, such as the accuracy is not guaranteed if the voltage and current levels are changed due to a change in the application. Moreover, the training of the ANN consumes a lot of effort and does not cover all cases [82].



Figure 3.11: The block diagram of the ANN IGBT fault detection method [26].

3.5.5 Submodule fault detection using sliding mode observation

This method is classified as model-based software technique; it fully depends on observing the arm currents and capacitor voltages of the MMC. As shown in Figure 3.12, it has two operation modes. The first mode is the monitoring mode; in this mode the arm currents are observed using the sliding mode observer (SMO).



Figure 3.12: Flowchart of the SMO submodule fault detection technique [78].

If there is an error between any arm current and its observed state, then this arm is considered to have a fault, and the algorithm starts the second mode, which is the detection mode. In the detection mode, all of the arm submodules are exchanged with redundant submodules one by one. Then, the voltage of each submodule and the arm current are observed. If the observed capacitor voltage and arm current start to follow their actual value after replacing the submodule with a redundant one, then this submodule is considered to be faulty [78].

This method is considered to be very accurate and has a very robust performance compared to other techniques. However, it has some disadvantages that are summarised in the following points [78]:

- This method requires the measurement of each capacitor voltage, which requires a large number of voltage transducers and a sophisticated communication network to transmit their signals, which not only increases the cost but also complicates the whole control system.
- It requires the incorporation of redundant submodules, which will be used if and only if there is a fault; this also adds extra cost to the system.
- The process time of the algorithm is not small since it scans the observed signals of all arm currents and capacitor voltages. Moreover, exchanging the fault arm submodules with redundant submodules one by one consumes a lot of time because some MMCs can have about 400 submodules per arm. This makes it impossible to detect short circuit faults since the required detection time should be less than a few milliseconds.

Table 3.5 shows a comparison of all fault detection techniques presented in this chapter. It is clear that hardware techniques have higher accuracy than software-based techniques; however, their costs are much higher.

Table 3.5: Comparison between common MMC fault detection techniques proposed in literature.

| Fault Detection Method | Classification | Cost | Complexity | Accuracy |
|------------------------|----------------|----------|------------|----------|
| Rate of change of the | Hardware | High | High | High |
| collector current | | | | |
| Behaviour of the gate | Hardware | High | High | High |
| voltage | | | | |
| Measurements of | Software | Moderate | Moderate | Moderate |
| submodule output | | | | |
| voltage | | | | |
| Artificial neural | Software | Low | High | Low |
| networks | | | | |
| Sliding mode | Software | Moderate | High | High |
| observation | | | | |

This has motivated the author to implement a new method for fault detection that does not need hardware circuits and simultaneously improves accuracy when compared with other software strategies.

3.6 The Concept of Fault Tolerant Control

The FTC system is the system that is able to perform its desired control functions at a high level of stability and security during the occurrence of a certain fault. The FTC systems are gaining the interest of many researchers in different research fields because of their ability to enhance the overall reliability of systems. The FTC systems are classified into two main types: passive and active [87]–[89].

3.6.1 Active fault tolerant control

As shown in Figure 3.13, active FTC systems receive continuous fault status from the FDU. Then, at the fault condition, it reconfigures the control system to a new system, which isolates the faulty part of the system and maintains the stability of the plant. It is important to mention that the redesign process of the system is performed online, which normally may allow some changes to the desired control functions or degradation in the performance [90], [91].

The main disadvantage of active FTC techniques is their slow response due to the time consumed on the fault detection and controller redesign. However, the



Figure 3.13: Structure of active FTC.

main advantage of these techniques is their ability to detect a wide range of faults [91].

3.6.2 Passive fault tolerant control

In passive FTC, the controller is designed in the offline mode and then it operates with fixed behaviour during normal and abnormal conditions. Therefore, it should be designed such that it can tolerate the faults [92], [93]. The main advantages of passive FTC techniques can be summarised in the following points:

- There is no need for the fault detection process, as the system is working with the same parameters during normal as well as fault conditions.
- Their response is very fast due to the elimination of FDU and the control redesign processes.

However, it is very difficult for one passive FTC system to detect different types of faults since they are designed in the offline mode [94].

3.7 Fault Tolerant Control of MMCs

As mentioned in the previous subsections, the MMC is very weak to certain types of faults. At the same time, MMCs should work at a high level of reliability, as they usually work in critical applications that do not have the luxury of losing the power transfer. Based on this, it is highly recommended to provide the control of the MMC with fault tolerance capability to enable the MMC able to deliver the desired power during faulty conditions without the need to block the converter [26], [95]. The following subsections investigate popular methods used for tolerating the fault in MMCs.

3.7.1 Fault tolerant control using redundant submodules

Regarding tolerating the fault using redundant submodules, the idea depends on disconnecting the faulty submodule and inserting a new redundant submodule without the disconnection of the MMC, as shown in Figure 3.14 [26], [95], [96].



Figure 3.14: Structure of MMC with redundant submodules [26].

The process of insertion of the redundant submodules is done using a strict control. At each trigger time, the control system identifies the state of each submodule with the aid of a suitable fault detection algorithm. If any submodule is subjected to high switching commutations or a fault, the control system bypasses the faulty submodule and replaces it with a redundant submodule, as shown in Figure 3.15 [96].



Figure 3.15: Block diagram of fault tolerant control with redundant submodules [96].

The concept of redundancy will improve the fault tolerance, providing maximum safety to MMCs. However, from an economic point of view, the redundancy increases the cost of the MMC, as it adds extra components that are not used during normal operation [96]. Adding to this, the circuit structure and the control system will be more complex.

3.7.2 Fault tolerant control using redundant IGBTs

In this method, the fault tolerance concept is achieved by adding two redundant IGBTs to each submodule and a high-speed AC switch, as shown in Figure 3.16. This means that, for half-bridge submodules, the number of switching devices will be increase from 2 to 4, while the number of switching devices will be increased from 4 to 6 in full-bridge submodules [95]. Figure 3.17 illustrates the way faults are tolerated using the manipulation of firing signals between faulty and redundant submodules.



Figure 3.16: The circuit diagram of MMC submodule with IGBT redundancy for: a) half bridge, b) full bridge.



Figure 3.17: Tolerating the faults using redundant IGBTs [95].

If the IGBT S1 suffers from an open circuit fault, the firing of S1 and S2 is transferred to S3 and S4, and this guarantees the continuity of the power flow. However, if the IGBT S1 is suffering from a short circuit fault, a different switching sequence should be followed since the anti-parallel diode of S1 will provide a permanent current path. In this case, the high-speed switch that connects the S2 with S4 is closed to prevent the leakage of current to the DC source through S1. Then, the switching is applied to S3 and S4.

It is obvious that the research conducted concerning FTC of MMCs is not enough, especially as most of the methods proposed in the literature depend on using redundant submodules. This has motivated the author to propose a new FTC technique that depends on sorting algorithms instead of using redundant components, as presented in Chapter 6.

3.8 Chapter Summary

This chapter presented several capacitor voltage balancing techniques that were proposed in literature, showing the main differences between them and the proposed capacitor voltage balancing technique presented in Chapter 4. In addition, different MMC faults have been mapped, showing the consequences of each fault type. Finally, popular fault detection and tolerant control techniques have been also investigated to show what has motivated the researched to implement the proposed fault detection and tolerant control techniques presented in Chapters 5 and 6.

Chapter 4

Proposed Capacitor Voltage Balancing Technique Based on Capacitor Voltage Estimation Algorithms

4.1 Introduction

As mentioned in Chapter 3, the process of capacitor voltage balancing is mandatory for a stable operation of the MMC. If these voltages are left without control, the difference between them will lead to increasing the converter differential current, which means that the converter losses will be higher. In addition, the differential current affects the quality of the output current because of the superimposed harmonics. This chapter presents a new capacitor voltage balancing technique with a reduced number of sensors. The developed technique is based on capacitor voltage estimation. For this purpose, two adaptive capacitor voltage estimation techniques are developed. The first is based on the ADALINE algorithm, while the other is based on the RLS algorithm. The performance of the two techniques is compared to determine which gives better results in terms of speed of convergence and accuracy. Then, the description of control loops that are incorporated into the proposed balancing strategy are explained. Moreover, the design considerations of these control loops, including the selection of PID gains for each loop is conducted. Finally, simulation results for different case studies are presented to show the performance of the proposed capacitor voltage balancing technique and compare it with the classical balancing strategy.

4.2 Proposed Technique of Capacitor Voltage Balancing

It can be concluded from subsection 3.2 that most of the capacitor voltage techniques proposed in the literature require the continuous measurement of capacitor voltages. This simply means that if an industrial MMC has 401 levels, the manufacturer must equip its submodules with 800 voltage sensors for each phase, keeping in mind that the acquisition of these sensors will need corresponding communication channels. This motivated the researcher to develop a new technique for balancing voltages of submodule capacitors without the need for measuring actual capacitor voltages, while getting a better performance by increasing the power quality of the MMC output power. The proposed algorithm depends on estimating submodule capacitor voltages instead of direct measurements. It consists of three main stages:

- capacitor voltage estimation,
- averaging control, and
- balancing control for capacitor voltages.

4.2.1 The capacitor voltage estimation technique

The capacitor voltage estimation is considered an extremely important step of the proposed control strategy. Although the capacitor voltage estimation eliminates the use of direct measurements, it is a must to maintain the estimation accuracy to avoid wrong control actions. Different algorithms have been developed in the past for the purpose of tracking dynamically changing signals. Sliding mode observer is one of the oldest methods that had been used to achieve this goal which is considered to be a basic technique supported by the fundamental control theory. In this technique, the procedure of the sliding mode observer gain is determined to drive the estimation error trajectories and maintaining them on the sliding surfaces. The sliding mode observers have a lot of advantages such as: being well established and known technique, their mathematics are built according to the mathematical model of the system and they are very robust as they do not get affected from system uncertainties and external noises. However, they suffer from critical disadvantages such as: the huge number of calculations (especially for systems with high number of states) and the estimated signals contains a variable steady-state errors[78], [97], [98].

Kalman filter is another technique which is very common in tracking the dynamically changing signals which particularly have multi-input and multi-output systems. Since it is a recursive method, it uses the state space model to get the best estimation of the tracked signal. The error of the estimation is minimized based on different criterions so the convergence stage is reached. Although the Kalman filter have high level of accuracy and the ability estimate invariant dynamic modes, it suffers from many drawbacks [98], [99] such as:

- It is severely affected by external noise and uncertainty of the parameters.
- It is very complex and requires huge number of calculations.
- It is very difficult to achieve real time operation when the system has higher orders such as 4 or 5.

For this particular purpose, the author proposed two capacitor voltage estimation techniques which have better performance when compared to other methods presented in literature. The first is achieved using the ADALINE algorithm, while the second algorithm uses the RLS algorithm.

4.2.1.1 Capacitor voltage estimation using ADALINE technique

The ADALINE technique is known for being a very efficient and quick online tracking technique for tracking dynamically changing voltage signals. It is utilised in many applications because of its robust performance, low calculation burden, and accurate results [100], [101]. The ADALINE algorithm is formed by simple calculations that do not consume large computing time, which is very important in the application of capacitor voltage estimation. To estimate capacitors voltages, the MMC governing equations given by (2.13) and (2.14) are rearranged and rewritten in the vector form as follows:

$$\frac{v_{dc}}{2} - v_o - v_{Lu} = \begin{bmatrix} S_{u1} & S_{u2} & \dots & S_{uN} \end{bmatrix} \begin{bmatrix} v_{cu} \ est \\ v_{cu2} \ est \\ \vdots \\ v_{cuN} \ est \end{bmatrix},$$
(4.1)

$$\frac{v_{dc}}{2} + v_o - v_{Ll} = \begin{bmatrix} S_{l1} & S_{l2} & \dots & S_{lN} \end{bmatrix} \begin{bmatrix} v_{cl1_est} \\ v_{cl2_est} \\ \vdots \\ v_{clN_est} \end{bmatrix},$$
(4.2)

where S_{xi} is the switching state for the i^{th} submodule, $v_{Lu} = L_{arm} \frac{di_u}{dt}$, and $v_{Ll} = L_{arm} \frac{di_l}{dt}$.

The ADALINE algorithm produces a linear combination of its input vector X(k), which represents the switching-state vector $[S_{x1} \ S_{x2} \ \dots \ S_{xN}]^T$ at time k, where the suffix T refers to the transpose operation. In this technique, the input vector is multiplied by the weight vector W(k), which resembles the estimated capacitors voltages given in (4.1) and (4.2) to produce the predicted linear output $\hat{y} = X^T W(k)$. The next step is updating the weight vector using the Widro-Hoff delta rule given by [102]:

$$W(k+1) = W(k) + \alpha \frac{X(k)(y(k) - \hat{y}(k))}{X(k)^T X(k)},$$
(4.3)

where α is the reduction factor and y is the actual output, which is given as:

$$y(k) = \frac{v_{dc}}{2} - v_o - v_{Lu}, \tag{4.4}$$

$$y(k) = \frac{v_{dc}}{2} + v_o - v_{Ll}.$$
(4.5)

When the difference between the measured signal y(k) and the estimated signal $\hat{y}(k)$ converges to zero, the ADALINE algorithm decomposes the signal and estimates capacitor voltages. Figure 4.1 presents a detailed flowchart for the ADALINE algorithm. It is observed that increasing the reduction factor α increases the convergence speed due to losing the stability, as the prediction error may increase dramatically. This observation is a common behaviour of the Widrow-Hoff delta rule for all the study cases. A practical value for the reduction factor α is 0.002 for this application. This value is determined based on minimising the error to guarantee system stability [103]. It is worth mentioning that the proposed capacitor voltage across the arm reactors and the output phase voltage.


Figure 4.1: Flow chart of capacitor voltage estimation using ADALINE technique.

This action enables the implementation of a low-cost centralised controller for the MMC with a large number of submodules since the voltage measurement of submodules and their associated communication links are eliminated.

4.2.1.2 The capacitor voltage estimation using RLS technique

The RLS technique is a modified method of the least squares solution, which is reformulated into an iterative solution [104]. One of the main advantages of using the RLS as an estimation technique is the reduced dependency on historical measurements from previous iterations.

The fundamental RLS algorithm is formed by two basic equations. The first equation, given in (4.6), is to update the estimate vector $\hat{\theta}(k)$ of the submodule

capacitor voltages, while the second equation is used to update the weighting matrix P(k) as shown in (4.7).

$$\hat{\theta}(k) = \hat{\theta}(k-1) + P(k)\hat{\theta}(k) [y(k) - \phi(k)^T \hat{\theta}(k-1)], \qquad (4.6)$$

$$P(k) = \frac{1}{\lambda} \Big[P(k-1) - \frac{P(k-1)\phi(k)\phi(k)^{T}P(k-1)}{\lambda+\phi(k)^{T}P(k-1)\phi(k)} \Big],$$
(4.7)

where $\phi(k)$ is the connection vector that represents the switching-state vector $[S_{x1}, S_{x2}, \dots, S_{xN}]^T$ at time k, and λ is the forgetting factor that controls the speed of updating the weight matrix in (4.7). The result of multiplying $\phi(k)^T$ by $\hat{\theta}(k-1)$ gives the predicted output, \hat{y} [105]. As shown in Figure 4.2, the predicted signal is subtracted from the measured signal, and the error is used to update the estimated vector. When the estimation error approaches zero, convergence is reached, and the estimated submodule capacitor voltages are calculated.



Figure 4.2: Flow chart of capacitor voltage estimation using RLS technique.

It is clear that the calculation burden of the RLS is slightly higher than the ADALINE; however, the RLS is characterised by its fast and accurate dynamic response. Figure 4.3 illustrates the general block diagram of the capacitor voltage estimation and the process of updating the capacitor voltages.



Figure 4.3: Block diagram of capacitor voltage estimation unit.

A test is run to compare between the dynamic performance of ADALINE and the RLS techniques for the estimation of capacitor voltages of the MMC. Figure 4.4 illustrates that the RLS algorithm offers a faster dynamic response and more accurate performance than that of the ADALINE algorithm for estimating the timevarying capacitor voltage signal. Therefore, the RLS algorithm is used to provide the estimated capacitor voltages for the proposed control strategy of the MMC.



Figure 4.4: Comparison between the RLS and ADALINE performances: a) actual capacitor voltages versus RLS and ADALINE estimated signals, b) percentage error between actual voltage and RLS estimation, c) percentage error between actual voltage and ADALINE estimation.

4.2.2 The averaging control

The averaging control is responsible for controlling the average voltage across the complete leg. This controller is formed from two cascaded loops, as shown in Figure 4.5 [106].



Figure 4.5: Block diagram of the averaging controller.

The first loop is the voltage control loop where the estimated average voltage, $\hat{v}_{avg} = (\sum v_{cu_{est}} + \sum v_{cl_{est}})/2N$, is subtracted from the reference capacitor voltage, v_c^* , and the resultant error is processed through a PI controller to generate the reference signal for the differential current, i_{diff}^* . Another PI controller is utilised in the inner loop to regulate the differential current, calculated from (2.10), at its reference signal. The action of the inner loop controller is the average voltage signal, v_{avg}^* .

To determine the value of the gains for the inner and outer loops PI controllers, the mathematical representation of each loop should be introduced using Laplace transforms. Regarding the inner loop which controllers the differential current, the loop gain is given by:

$$l(s) = \frac{K_i(s)}{ls + r_a} \tag{4.8}$$

This means that the loop has a pole at $s = \frac{-r_a}{l}$, this pole reduces the magnitude and the phase of the loop gain at low frequencies. To cancel the effect of this pole, the PI controller is added to place a zero. This modifies $K_i(s)$ to be:

$$K_i(s) = k_{p1} + \frac{k_{i1}}{s} \tag{4.9}$$

Where k_{p1} and k_{i1} are the proportional and integral gains respectively which should be as follows:

$$k_{p1} = \frac{l}{\tau} \tag{4.10}$$

$$k_{i1} = \frac{r_a}{\tau} \tag{4.11}$$

Where τ is the time constant, based on this the loop gain is changed to:

$$l(s) = \frac{1}{\tau(s)} \tag{4.12}$$

The transfer function of the inner loop is given by:

$$G_i(s) = \frac{i_{diff}}{i_{diff}^*} = \frac{1}{\tau(s)+1}$$
(4.13)

To enable the controller from functioning with fast response, the τ should be by far lower than the switching frequency which is 2 kHz. The τ is selected to be 4 ms which is 1% from the value of the switching frequency. By substituting in (4.12) and (4.13), the values of k_{p1} and k_{i1} are 0.5 and 37.5 respectively. Figure. 4.6 demonstrates the bode plot for the inner loop transfer function with the selected PI gains showing a stable performance with fast response.



Figure 4.6: The bode plot of the inner loop of the averaging controller.

Moving to the outer loop which controls the average voltage across the whole leg, the steady state error is eliminated by adding the PI controller with the following gain:

$$K_{\nu}(s) = k_{p2} + \frac{k_{i2}}{s} \tag{4.14}$$

As shown in (4.14), the transfer function of the inner loop has a bandwidth which is by far higher than the bandwidth of the outer loop. Based on this fact, the gain of the inner loop is assumed to be 1 for the purpose of simplifying the calculations. As a result, the outer loop gain is given by:

$$l_2(s) = \frac{k_{p2}s + k_{i2}}{2 \, c_{Sm} \, s^2} \tag{4.15}$$

From (4.16), the transfer function of the outer loop can be expressed as a second order differential equation as follows:

$$G_{v}(s) = \frac{v_{avg}}{v_{avg}^{*}} = \frac{k_{p2}s + k_{i2}}{2 Csm s^{2} + k_{p2}s + k_{i2}}$$
(4.16)

Using MATLAB software, this equation is solved for ξ equals 0.7 for a better dynamic response. The gains of the PI dedicated for the outer loop is calculated as follows:

$$k_{p2} = 4 \, Csm \,\xi \,\omega_o \tag{4.17}$$

$$k_{i2} = 2Csm\,\xi\,\omega_o^{\ 2} \tag{4.18}$$

The natural frequency ω_o is kept very low compared to the frequency of the MMC AC voltage output. The k_{p2} and the k_{i2} are 1.5 and 150 respectively. Figure 4.7 shows the bode diagram of the outer loop.



Figure 4.7: The bode plot of the outer loop of the averaging controller.

4.2.3 Balancing control

The balancing control, presented in Figure 4.8, is a centralised control system that generates reference signals to balance the voltage across the submodules capacitors based on the estimated voltages from the ADALINE processing unit. First, the controller subtracts the estimated capacitor voltage of each submodule from the reference capacitor voltage, v_c^* . The resulting errors are passed to simple proportional (P) controllers. The outputs from the P-controllers are multiplied by the sign of their corresponding arm current to form reference signals for balancing the voltage across capacitors (v_{cul_bal} , ..., v_{cuN_bal} , v_{cll_bal} , ..., v_{clN_bal}). Finally, to form the modulating signal for a submodule, the balancing control signal for this submodule, the average voltage command, v_{avg}^* , and the reference phase voltage, v_o^* , are added. The PS-PWM technique is utilised to generate the switching signals for the submodules of the MMC [107]. As shown in this subsection, the proposed scheme eliminates the need for massive numbers of voltage sensors for the submodule capacitors and their associated communication systems with the central controllers. Consequently, the proposed strategy has succeeded in substantially decreasing the cost of implementation at a very good operating performance rendered in its application for an MMC system with a large number of submodules.



Figure 4.8: Block diagram of the balancing controller.

Following the same design concept introduced in the previous subsection, the gain of the balancing controller $k_{p \ b}$ is selected to be 0.35.

4.3 MMC Simulation Models

Two simulation models have been developed for the validation of all proposed techniques in the thesis. The first model demonstrates a DC voltage source connected to a three-phase load through a detailed five-level MMC. The second model demonstrates two wind farms connected to the grid through a HVDC link based on two MMCs with 77 voltage levels. The average model of the MMC is considered.

4.3.1 MMC Model 1

The first model considers a detailed simulation of a five-level MMC. Each submodule in the MMC is formed from two IGBTs and a capacitor. As shown in Figure 4.9, the MMC connects a DC voltage source to a three-phase load, which means that the MMC permanently acts as an inverter. The rated power of the MMC is selected to be 1 MW, while the rated DC voltage is 9,000 V.



Figure 4.9: The circuit diagram of the first MMC model.

To calculate the size of the capacitor, first the arm energy variation must be plotted according to (2.30), assuming that the power factor is 0.8. Figure 4.10 illustrates the energy variation across the arm. The maximum energy stored in the arm is 3.855 kJ. If the maximum allowable ripples for the capacitor voltage $k_{ripple max}$ are 10% and for the arm current Δi_{diff} are 20%, the value of the submodule capacitor C_{SM} is chosen to be 1900 µF to fit the criteria presented in (2.32).



Figure 4.10: The arm energy variation of the MMC simulated in the first model.

| MMC model parameters of the first model | |
|--|-------------|
| Rated power | 1 MW |
| Submodule rated voltage | 2,250 V |
| Rated DC voltage | 9,000 V |
| Arm inductance | 3.3 mH |
| Submodule capacitance | 1900 µF |
| Number of submodules per leg | 8 |
| Carrier frequency, <i>f</i> _c | 2 kHz |
| Load | 30 Ω, 10 mH |
| | |

Table 4.1: Summary of the MMC parameters of the first model.

The PS-PWM modulation technique is utilised for the switching of the MMC submodules. The frequency of carrier signals f_c is 2 kHz, which means that the switching cycle of the IGBT T_s is 0.5 ms. Thus, the arm inductor L_{arm} is selected to have a value of 3.3 mH to fit the rule presented in (2.34). Table 4.1 summarises the parameters of the simulated MMC.

4.3.2 MMC Model 2

Unlike the detailed model presented in the previous subsection, the average model of a 77-level MMC is considered to facilitate the calculations because of the relatively high number of submodules. The average model used in the simulation depends on converting the detailed model of the submodule to its Thevenin equivalent [57].

As shown in Figure 4.11, the model contains two wind farms, where each has a rating of 500 MW. The first wind farm delivers the energy through a permanent magnet synchronous generator (PMSG), and the output voltage of the PMSG is

0.69 kV. The terminals of the PMSG are connected to a back-to-back two-level VSC.



Figure 4.11: The circuit diagram of the second MMC model.

The generator used in the second wind farm is a double-fed induction generator (DFIG), which generates the power at 0.69 kV. The rotor winding of the DFIG is connected to a back-to-back two-level VSC. The output of this VSC and the stator winding of the DFIG are both connected to a three-core transformer. Since both windfarms are working on the same voltage level, a point of common coupling is formed, connecting the two wind farms together with the rectifier stage of the HVDC. The HVDC link is formed from two MMCs connected through a 200 kM cable. The output of the inverter stage of the HVDC system is connected to a 220 kV AC grid. Modelling details of wind turbines, PMSG, DFIG, and all control loops

for controlling the generated power from the wind farms and the injected power into the grid are illustrated in Appendix A.

The rated power of both MMCs is selected to be 1,000 MW, while the rated DC voltage is 640 kV. The size of the submodule capacitor is calculated in the same manner as presented in the previous subsection. First, the arm energy variation must be plotted according to (2.30), where 0.8 power factor is assumed. Figure 4.12 illustrates the energy variation across the arm. It can be seen from the plot that the maximum energy stored in the arm is 3.855 kJ. If the maximum allowable ripples for the capacitor voltage $k_{ripple max}$ are 10% and for the arm current Δi_{diff} is 20%, the value of the submodule capacitor C_{SM} is chosen to be 220 µF to fit the criteria presented in (2.32).



Figure 4.12: The arm energy variation of the MMC simulated in the second model.

Similar to Model 1, the PS-PWM modulation technique is used and the frequency of carrier signal f_c is 2 kHz. A 70 mH arm inductor is selected to fit the rule presented in (2.34). Table 4.2 summarises the parameters of the simulated MMC of Model 2. Details of simulation of both models are found in Appendix B.

| MMC Model Parameters of the Second Model | |
|--|-----------------|
| Rated power | 1,000 MW |
| Submodule rated voltage | 640/76=8.4 8 kV |
| Rated DC voltage | 640 kV |
| Arm inductance | 70 mH |
| Submodule capacitance | 220 µF |
| Number of submodules per | 152 |
| leg | |
| Carrier frequency, <i>f</i> _c | 2 kHz |
| Grid voltage | 220 kV |

Table 4.2: Summary of the MMC parameters of the second model.

4.4 Simulation Results

The proposed capacitor voltage balancing strategy is applied to the MMC, which is simulated using the PSCAD/EMTDC software package [108]. Three case studies were simulated to evaluate the performance of the MMC and the proposed capacitor voltage balancing technique under different operating conditions. Case 1 and Case 2 used the system parameters shown in Table 4.1, while Case 3 was applied to the MMC with parameters shown in Table 4.2. Primarily, the first task is dedicated to the performance evaluation of the proposed RLS algorithm in order to estimate capacitor voltages used for controlling the MMC under dynamic reference phase voltage changing conditions, while the second case is devoted to assessing the dynamic performance of the proposed control strategy during the boost operation of the MMC. Regarding the third case, it is conducted to evaluate the proposed capacitor voltage balancing technique when applied to a grid-connected MMC with a large number of voltage levels.

4.4.1 Case 1: Dynamic performance of the proposed capacitor voltage estimation-based control strategy

The purpose of this simulation case is to examine the performance of the proposed capacitor voltage estimation-based control algorithm for the MMC under the dynamic change of the reference phase voltage. The reference capacitor voltage v_c^* is set at vdc/N = 2.25 KV, while the reference phase voltage signal v_o^* is dynamically changed from 0.7 pu to 1 pu at t = 0.2 s. First, the estimation units for capacitor voltages are disabled, and the actual measurements are used in the control system. The three-phase voltages, currents, and capacitor voltages are shown in

Figure 4.13. Similar results to Figure 4.13 are obtained in Figure 4.14, where the proposed estimation unit for the capacitor voltages are enabled and utilised in the proposed control system of Figures 4.5 and 4.8. As displayed in Figure 4.14(a), the measured three-phase AC voltages follow their reference signals. As expected, only six submodules are utilised when $v_o^* = 0.7 \text{ pu}$, while all the submodules are involved when $v_o^* = 1$ pu. The three-phase load currents are displayed in Figure 4.14(b). Figure 4.14(c) traces the capacitor voltages, which are grouped into two main trajectories: one for the upper arm and the other for the lower arm. These two trajectories are out of phase, and they are identical to that presented in Figure 4.13(c), where the measured capacitor voltages are used instead of their estimated signals in the control of the MMC. The proposed MMC controller succeeds in balancing the capacitor voltages at their reference value of 2.25 KV. Moreover, the estimated capacitor voltages and their corresponding actual measurements for the upper and lower arms are presented in Figures 4.15 and 4.16, respectively. Fast tracking with accurate performance of the proposed RLS algorithm for estimating the submodules capacitor voltages are evident. Furthermore, the circulating current tightly tracks its reference signal, as demonstrated in Figure 4.17(a). Figures 4.17(b) and 4.17(c) illustrate the actions of the averaging controller of Figure 4.5 and the balancing controller of Figure 4.8 for the first submodule at the upper arm $v_{cu1 \, bal}^*$, respectively. These results reveal the efficient utilisation of the proposed capacitor voltage estimation technique for the control algorithm of the MMC.



Figure 4.13: Load waveforms of Case 1, while disabling the proposed estimation unit: a) three-phase voltages, b) three-phase load currents, c) voltages of Leg a submodules.



Figure 4.14: Load waveforms of Case 1, while enabling the proposed estimation unit: a) three-phase voltages, b) three-phase load currents, c) voltages of Leg a submodules.



Figure 4.15: Case 1 actual and estimated voltages of the upper arm submodules of Phase a.



Figure 4.16: Case 1 actual and estimated voltages of the lower arm submodules of Phase a.



Figure 4.17: Action of the different controllers for Leg a during Case 1: a) actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm.

4.4.2 Case 2: Dynamic performance during the boosting operation of the MMC

In this test scenario, the reference phase voltage signal v_o^* is kept at 1 pu and the capacitor reference voltage command is increased from 2.25 KV to 2.5 KV at t = 0.4 s. Setting v_c^* at values higher than *Vdc/N* results in boosting the operation of the MMC. This operation is useful to compensate for the load voltage during submodule faults. Figure 4.18(a) indicates the increase of the output of three-phase voltages from the MMC at t = 0.4 s when the proposed controller succeeds in boosting and balancing the capacitor voltages at the new set value of 2.5 KV, as illustrated in Figure 4.18(c). Yet again, fast dynamics with a tolerated error of the proposed capacitor voltage estimation units are revealed from Figures 4.19 and 4.20 for the submodules at the upper and lower arms, respectively. Boosting the capacitor voltages increases the circulating current. Hence, the balancing voltage signal is escalated, as demonstrated in Figure 4.21.



Figure 4.18: Load waveforms of Case 2: a) three-phase voltages, b) three-phase load currents, c) voltages of Leg a submodules.



Figure 4.19: Case 2 actual and estimated voltages of the upper arm submodules of Phase a during the boosting operation.



Figure 4.20: Case 2 actual and estimated voltages of the lower arm submodules of Phase a during the boosting operation.



Figure 4.21: Action of the different controllers for Leg a during the boosting operation (Case 2): a) actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm.

4.4.3 Case 3: Performance of capacitor voltage estimation algorithm in grid-connected MMC

This case is particularly developed to evaluate the proposed capacitor voltage estimation technique, when applied to an MMC-based HVDC system that integrates offshore wind power with the grid, and to investigate the accuracy of the proposed capacitor voltage estimation techniques for an MMC with a relatively high number of voltage levels. Furthermore, the dynamic performance of the proposed capacitor voltage estimation technique is assessed under external grid faults. The Model 2 MMC with the parameters shown in Table 4.2 is used. The simulated MMC has 152 submodules per leg. Unlike the previous cases, the capacitor voltage balancing technique used in this case is the reduced frequency-based technique presented in Subsection 3.2.3. The RLS capacitor voltage estimation technique provides it with the estimated capacitor voltages instead of using direct measurements. This is done to prove that the proposed RLS capacitor voltage estimation can be adopted for different capacitor voltage balancing techniques. In this case, the reference DC voltage signal is 640 kV. Then, at t = 6 s, a three-phase fault is applied at the grid side, which lasts for 0.2 s. The capacitor voltage estimation-based voltage balancing technique succeeds in balancing the capacitor voltages, as revealed from the balanced three-phase voltages presented in Figure 4.22. The RLS estimation algorithm provided the balancing control over the whole period even during the external fault period. Figures 4.23(a) and 4.23(b) illustrate the summation of capacitor voltages of the upper and lower arms. Moreover, the capacitor voltages of one submodule from the upper and lower arms are portrayed in Figures 4.23(c) and 4.23(d), respectively, which follows the traces of the actual capacitor voltages of the upper and lower arms, even during the grid fault. Results of this case proves that the proposed capacitor voltage estimation algorithm is not affected by external AC faults or the grid power controls. Therefore, the proposed capacitor voltage estimation algorithm is suitable for an MMC with a high number voltage levels, which are normally used in HVDC circuits.



Figure 4.22: Case 3 voltages and currents of grid: a) three-phase voltages, b) three-phase currents.



Figure 4.23: Case 3 actual and estimated signals of submodule capacitor voltages in Leg a: a) summation of capacitor voltages in the upper arm, b) summation of capacitor voltages in the lower arm, c) capacitor voltage of one submodule in the upper arm, d) capacitor voltage of one submodule in the lower arm.

4.5 Chapter Summary

This chapter has presented two capacitor voltage estimation units. The first one is based on the ADALINE algorithm, while the second is based on the RLS algorithm. Both are used for estimation submodule voltages of the MMC. For each leg of the MMC, the proposed estimation units require only three sensors: one for the phase voltage and the other two for the voltages across the arm reactors. The proposed estimation unit is then integrated into a control strategy to balance the voltages across the submodules of the MMC. A comparison between the two estimation techniques has been conducted, showing that the RLS has better performance than the ADALINE in terms of speed and accuracy. As a result, the RLS has been incorporated into the capacitor voltage balancing algorithm. The proposed technique eliminates the need for direct measurement of capacitor voltages and associated communication systems, which renders it suitable for implementing a low-cost centralised controller for an MMC with a large number of submodules. The fast response and accurate tracking of the estimation unit under different dynamic conditions are validated in simulation results. It is important that the proposed MMC controller successfully balances the capacitor voltages at their set values even in boosting operation mode. Furthermore, it could easily balance the capacitor voltages of a grid connected MMC with a high number of voltage levels without being affected by the grid controls or external faults.

Chapter 5

Proposed Fault Detection Technique Based on Capacitor Voltage Estimation Algorithms

5.1 Introduction

Faults that can be located inside submodules of the MMC which are due to failures in semi-conductor devices not only affect the quality of the output power, but also threaten the safety of the converter. Since MMCs have a large number of submodules, the probability of having a fault inside a submodule is relatively high. As a result, it is highly recommended that detection and localization of the fault is carried out quickly, in order to enable the MMC inner control system to deal with the fault, by either converter disconnection or applying a suitable fault tolerant technique.

This chapter presents a new fault detection technique for the diagnosis and localization of submodule faults that develop due to failures in switching devices. As opposed to other fault detection strategies that have been proposed in the literature (Section 3.5), the proposed fault detection technique does not need extra sensors nor special power circuits, as it depends on a hybrid ADALINE-RLS capacitor voltage estimation algorithm, which combines the two estimation algorithms described previously in Chapter 4. The rest of this chapter is organized as follows:

Firstly, the MMC arm currents and submodule voltages are mathematically analysed during submodule faults to determine the consequences of such faults, which is a mandatory step for designing the proposed fault FDU. The proposed unit is then explained in detail, focusing on its advantages compared to other techniques proposed in the literature. Finally, various case studies are applied to the PSCADsimulation model, in order to evaluate the performance of the FDU under various fault conditions.

5.2 The Proposed Fault Detection Strategy

This fault detection technique has been developed to solve all problems associated with other techniques found in the literature. As shown previously in section 3.5, most of these techniques require additional sensors to guarantee the robustness of the proposed technique, otherwise depending on software techniques such as the ANN-based detection method, which do not offer the required accuracy for this critical application. Thus, the proposed fault detection technique is classified as a model-based software technique. Figure 5.1 illustrates the block diagram of the proposed FDU. The proposed algorithm is built according to the electrical disturbances associated with submodule IGBT faults, without the need to incorporate any sensors or special power circuits, instead depending on an estimation of different signals, based on adaptive strategies.

As shown in Figure 5.2, the proposed FDU receives the values of the RLS estimated submodule capacitor voltages $v_{cxi RLS}$ (x = u, l) and (i = 1, 2, 3, ..., N), then these values are divided by the reference capacitor voltage v_c^* to obtain the per-unit values $v_{cxi RLS pu}$.



Figure 5.1. Block diagram of the proposed FDU.



Figure 5.2. Flowchart of the proposed fault detection technique.

At the same time, the error between the ADALINE-estimated capacitor voltage, $v_{cxi ADA}$ and the RLS-estimated voltage $v_{cxi RLS}$ of each submodule is calculated, and then passed through a high speed differentiator to calculate the rate of change of the error signals e_{xi} . The next step involves determining each per-unit value of the voltage estimated signal $v_{cxi RLS pu}$, whether it is higher than 85% or not. The reason for this check is based on the fact that the per-unit estimation of capacitor voltage of a faulty submodule is decreased during faults, this is simply due to the increased error between the actual signal and its estimation as the right and left hand sides of (4.1) and (4.2) being no longer equal. If the comparison result is

true, then the checked submodule is considered to be in a healthy condition, otherwise the algorithm starts to check the differentiated error signal e_{xi} and compares it with a predefined threshold, $v_{pu\,threshold}$, which is selected based on the required sensitivity of the FDU.

The reason for calculating this error is that the proposed FDU is based on the difference between the dynamic behaviours of the ADALINE and RLS algorithm results, which is due to their different natures. The FDU here depends on the difference of the behaviour between the two algorithms. As illustrated in the Widrohoff learning rule given in (4.3), which is used in the update of the ADALINE-estimated voltages, the weighting vector of the current iteration, W(k + 1), is updated by adding the error reduction component $\alpha \frac{X(k)(y(k)-\hat{y}(k))}{X(k)^T X(k)}$ to the weighting vector of the previous iteration, W(k). This means that the ADALINE algorithm has no memory, since the Widro-hoff equation is independent of the accumulative value of weighting vectors. This allows the ADALINE to be able to rapidly track sudden changes in capacitor voltage. However, it needs a relatively long time to reach convergence.

Unlike the ADALINE, updating the weighting vector of the RLS algorithm depends on the accumulation of historic values of previous capacitor voltages, as demonstrated in (4.6) and (4.7). This means that the RLS algorithm has a reduced convergence time when compared with the ADALINE. However, in the situation of having a sudden change in capacitor voltage (which occurs during submodule faults), the RLS has a slower response in tracking capacitor voltages.

If the error signal, e_{xi} is lower than the threshold, $e_{threshold}$, the algorithm considers the submodule as being in a healthy condition, otherwise it increases a counter by one and then starts a 100 µs timer. After this time period, the algorithm repeats the same steps and if the counter reaches 10, this implies that the submodule is faulty.

The proposed method proved its reliability and robustness, as it was tested under different healthy and faulty conditions. This is because it makes the decision based on two factors: the estimated per-unit voltage, and the rate of change of the estimated voltage error signals. Moreover, this method does not require any additional components or devices, as it is based on simple calculations, and can detect the fault in less than 5ms for short-circuit faults and less than 35 ms for opencircuit faults.

5.3 Simulation Results

The dynamic behaviour of the proposed FDU is tested under different fault conditions when applied to the PSCAD simulations of the MMC, presented in Table 4.1. The first case is devoted to assessing the dynamic performance of the proposed FDU during an open-circuit fault in a submodule, while the second case examines FDU behaviour for the MMC under a short-circuit fault in a submodule. Finally, the third case simulates two faults in two different arms, in order to assess the performance of the FDU during multiple faults.

5.3.1 Case 1: Performance under IGBT open circuit fault

This task is devoted to assessing the capabilities of the proposed FDU under an IGBT open-circuit fault in a submodule. The upper switch of the third submodule in the upper arm of phase *a* is intentionally opened at t = 0.5s. As a result, the voltage and current of phase *a* are distorted, as shown in Figures 5.3(a) and 5.3(b), respectively, due to the imbalance of submodule capacitor voltages. The main reason leading to the failure of the voltage-balancing strategy is the increased capacitor voltage of the faulty submodule. This is logical, since the capacitor is allowed only to be charged, with no path for discharging, due to the open circuit condition. One important concern in this case is that all capacitor voltages of the healthy submodules located in the arm are increased, as illustrated in Figure 5.3(c). The increase in capacitor voltages stresses the insulation of MMC components, and may lead to a cascading failure in other submodules.



Figure 5.3. FDU performance during an open-circuit fault on the third submodule of the upper arm: a) Three-phase voltages, b) three-phase load currents, c) voltages of leg a submodules.

Figures 5.4 and 5.5 clearly illustrate that the voltages of the healthy submodules adequately track their actual voltages, which proves the robustness of the estimation algorithm during faults, as errors between the actual and estimated signals have not reached 10%. Regarding the faulty submodule, there is a considerable error between the estimated voltage and its corresponding actual voltage, due to the decreased value of the estimated voltage, which enables the FDU algorithm to detect the fault. Figure 5.6 (a) indicates that the differential current is oscillating to twice its value during normal operating conditions. This means that the MMC is no more stable as the differential current, which flows inside the MMC, and the arm currents may exceed their capacities. In other words, the controller failed to balance the capacitor voltages during faults. Moreover, the heavy oscillations of the differential current are transferred to the averaging and balancing control signals, as demonstrated in Figures 5.6 (b) and (c), respectively. As a result, the MMC is no longer stable and must be deactivated.



Figure 5.4. Actual and estimated voltages of the upper arm submodules of phase a, under an open-circuit fault.



Figure 5.5. Actual and estimated voltages of the lower arm submodules of phase a, under an open-circuit fault.



Figure 5.6. Action of the different controllers for leg a during the open-circuit fault: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm.

As shown in Figure 5.7(a), the estimated voltage of the faulty submodule exceeded the FDU voltage threshold for a longer period than the other voltages. This is the first condition for the FDU to detect the fault. The estimated signals of all other submodules in the healthy arm are by far higher than the FDU voltage threshold. Ten ms after the fault, the derivative of the error signal between the RLS and ADALINE-estimated voltages for the faulty submodule by far exceeded the predefined threshold, and this is the second condition which the FDU needs to satisfy to successfully detect the fault. The FDU succeeds in detecting and localizing the fault 33 ms after being activated, as illustrated in Figure 5.7 (b). With the slow rate of increase of the capacitor voltages, the 33 ms period of detecting an open-circuit fault is enough to guarantee the safety of the MMC.



Figure 5.7: The performance of the FDU under IGBT open circuit fault: a) The RLS estimated voltages of the upper arm of phase a, b) The detection time consumed by the FDU to detect the open circuit fault.

5.3.2 Case 2: Performance under IGBT short circuit fault

This case is conducted in order to investigate the capabilities of the proposed FDU under IGBT short circuit fault in a submodule. A short-circuit fault is applied to the upper switch of the third submodule in the upper arm of phase *a* at t = 0.5s. Figures 5.8 (a) and 5.8(b) illustrate the increased level of harmonics and DC components on the voltage and current waveforms of phase *a*, respectively. The submodule capacitor voltages are not balanced, as indicated in Figure 5.8(c).

Figures 5.9 and 5.10 show the RLS estimated capacitor voltages. The voltages of the healthy submodules adequately track their actual voltages, which proves the robustness of the estimation algorithm during faults. Regarding the faulty submodule, there is an error between the estimated voltage and its corresponding actual voltage, which enables the FDU algorithm to detect the fault.



Figure 5.8: Operation during a short-circuit fault on the third submodule of the upper arm: a) Three-phase voltages, b) three- phase load currents, c) voltages of leg a submodules.



Figure 5.9: Actual and estimated voltages of the upper arm submodules of phase a, under a short-circuit fault.



Figure 5.10: Actual and estimated voltages of the lower arm submodules of phase *a*, under *a* short-circuit fault.

The voltage-balancing strategy fails due to short-circuiting the faulty submodule capacitor when activating its lower switch. The proposed balancing strategy fails to regulate the differential current at its increased reference signal, as shown in Figure 5.11 (a). The value of differential current approaches the level of the load current, which threatens the safety of the MMC as the arm currents may exceed the rated current of the IGBT switches. Consequently, the averaging and balancing control signals are not stable, and bounce between the upper and lower limits of their controllers, as exhibited in Figures 5.11 (b) and (c), respectively. These results demonstrate that it is extremely dangerous to continue operation of the MMC with an IGBT short circuit fault without either the disconnection of the MMC, or applying a suitable fault tolerant control strategy.



Figure 5.11: Action of the different controllers for leg a during the short-circuit fault: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm.

As shown in Figure 5.12 (a), the estimated voltage of the faulty submodule exceeded the FDU voltage threshold for a longer period than the other voltages. This is the first condition for the FDU to detect the fault. While the estimated signals of all other submodules in the healthy arm are by far higher than the FDU voltage threshold. Also the derivative of the error signal between the RLS and ADALINE-estimated voltages for the faulty submodule by far exceeded the pre-defined threshold, and this is the second condition which the FDU needs to satisfy to successfully detect the fault. The FDU succeeds in detecting and localizing the short-circuit fault after 2.2 ms from its incidence, as demonstrated in Figure 5.12. (c). This detection time is fast, as the required time for clearing short circuit faults must not exceed 5 ms [25].



Figure 5.12: The performance of the FDU under IGBT short circuit fault: a) The RLS estimated voltages of the upper arm of phase a, b) The detection time consumed by the FDU to detect the open circuit fault.

5.3.3 Case 3: Performance under multiple faults in two different arms

This case is conducted in order to investigate the capabilities of the proposed FDU under multiple faults in two different arms. An open-circuit fault is applied to the upper switch of the third submodule in the upper arm of phase *a* at t = 0.5s, while a short circuit fault is applied to the lower switch in the second submodule in the lower arm of phase *a* at t = 0.7s. As shown in Figures 5.13 (a) and 5.13(b), the voltage and current of phase *a* suffer from an increased level of harmonics and high DC component just after the occurrence of the open-circuit fault. After applying the short circuit fault, the harmonics increase, affecting the power quality of the voltage and current and lowering it to a very poor level. The submodule capacitor voltages start to lose their balance just after the first fault, and increase rapidly after the application of the second fault, as indicated in Figure 5.13(c).



Figure 5.13: Operation during multiple faults: a) Three-phase voltages, b) three-phase load currents, c) voltages of leg a submodules.

The RLS-estimated voltages of leg a are shown in Figures 5.14 and 5.15. These estimated voltages track their actual voltages until the occurrence of the two faults. Following each fault, the estimated voltages contain a steady-state error. Regarding the estimated voltages of the faulty submodules, it is apparent that these signals rapidly decrease, which enables the FDU to easily detect the two faults.

Since the capacitor voltage of the third submodule cannot be charged and discharged properly due to the first fault, the voltage-balancing strategy partially fails to achieve its function. However, after the second fault, the situation becomes worse, and the balancing algorithm totally fails to balance the capacitor voltages. This appears in the failure of regulating the differential current at its increased reference signal, as shown in Figure 5.16 (a).

The value of differential current approaches the level of the load current after the first fault, and exceeds it after the second fault. This condition threatens the safety of the MMC, as the arm currents may exceed the rated current of the IGBT switches. Consequently, the averaging and balancing control signals are not stable, and bounce between the upper and lower limits of their controllers, as exhibited in Figures 5.16
(b) and (c), respectively. These results demonstrate that it is dangerous to continue operation of the MMC under multiple faults without a fault tolerant control.

As shown in Figure 5.17 (a), the estimated voltage of the two faulty submodules exceeded the FDU voltage threshold for a longer period than the other voltages. This is the first condition for the FDU to detect the fault. While the estimated signals of all other submodules are by far higher than the FDU voltage threshold. Also the derivative of the error signal between the RLS and ADALINE-estimated voltages for the faulty submodules by far exceeded the pre-defined threshold, and this is the second condition which the FDU needs to satisfy to successfully detect the fault. The FDU succeeds in detecting and localizing the first fault after 18.8 ms, while for the second fault the detection time is lower than 2.2 ms, as demonstrated in Figure 5.17. (b). This case has proved that the proposed FDU is able to detect multiple faults very quickly, which is required in MMCs as the high number of submodules may lead to several failures concurrently.



Figure 5.14: Actual and estimated voltages of the upper arm submodules of phase a, under two faults.



Figure 5.15: Actual and estimated voltages of the lower arm submodules of phase a, under two faults.



Figure 5.16: Action of the different controllers for leg a during multiple faults: a) actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm.



Figure 5.17: The performance of the FDU under two faults: a) The RLS estimated voltages of the upper arm of phase a, b) The detection time consumed by the FDU to detect the open circuit fault.

5.4 Chapter Summary

In this chapter, a new fault detection technique for the detection and localization of MMC submodule switch faults has been presented. The proposed FDU is based on a hybrid ADALINE-RLS capacitor voltage estimation algorithm, and does not require any additional components as they are based on the estimation algorithms that are simultaneously utilized for the proposed control of the MMC (presented in Chapter 4). PSCAD simulations illustrated the technique's accuracy for detecting and localizing different faults, and is a result of its hybrid processing where the fault decision is not only based on reduction of the estimated per-unit voltage, but also on the rate of change of the difference between the estimated voltages from the proposed ADALINE and RLS algorithms. Further, the results show the necessity of having a fault tolerant control scheme to protect the MMC during submodule faults, which helps it to continue in a safe operation mode, which is discussed in Chapter 6.

Chapter 6

Proposed Fault Tolerant Control Strategy Based on a Sorting Algorithm

6.1 Introduction

FTC systems are required to provide the MMC with fault-handling capability without blocking the converter during faults. Thus, this concept enhances the reliability of the MMC by increasing its availability with high levels of security, even in the case of having internal faults. This chapter presents a new fault tolerant control technique, which is integrated with the proposed capacitor voltage balancing and fault detection strategies presented in Chapters 4 and 5, respectively, in order to provide a complete inner controller for the MMC which is capable of controlling the converter during normal and abnormal operations. The proposed fault tolerant control is based on updating the parameters of the proposed capacitor voltage balancing technique, and the PS-PWM switching algorithm. This means that it does not need any redundant components or special conditioning circuits, which makes it very cost effective when compared to other techniques proposed in the literature (Section 2.9).

Firstly, the proposed FTC strategy is explained. Then, the reliability of the MMC is assessed before and after applying the proposed fault tolerant control technique, in order to evaluate the amount of reliability enhancement. Finally, the

inner MMC control technique, which contains the proposed capacitor voltage balancing, fault detection and fault tolerant control techniques is tested using PSCAD simulations, in order to evaluate its performance under different operating conditions.

6.2 The Proposed Fault Tolerant Control Algorithm

The proposed FTC system is built to overcome the problems presented in the previous section. The developed technique offers a good alternative to others proposed in the literature, as it does not need redundant components nor special circuits, whilst at the same time it does not affect the quality of the output power.

The proposed fault tolerant control unit (FTCU) has two main tasks: to isolate faulty submodules, and to modify the capacitor balancing algorithm. As shown in Figure 6.1, the sorting algorithm starts with the acquisition of the fault flags F_{x1} , F_{x2} , ... and F_{xN} . where x can be either u for the upper arm or l for the lower arm.

It then calculates the number of faulty submodules in the arm, as per the following expression:

$$N_{xF=x_1+F_{x2}+\dots+F_{xN}} \tag{6.1}$$

The algorithm then compares the percentage of faulty submodules to the total number of submodules in each arm. If this percentage is higher than 25%, the FTC sorting algorithm considers this a cascading failure condition, which means that a critical problem has occurred in the whole arm, and will spread to other arms. Thus, the algorithm isolates all MMC submodules and blocks the converter. If the percentage of faulty submodules is less than 25%, the sorting algorithm isolates the faulty submodules, and compares the number of faulty submodules in the upper and lower arms of each leg. If this number is not equal, it then isolates some healthy submodules until it becomes equal, in order to balance the energy between the two arms. Moreover, the algorithm calculates the number of effective submodules that are still in the operation:

$$N_{eff} = N - 2N_{xi} \tag{6.2}$$

After this step, the algorithm updates the reference capacitor voltage v_c^* according to the new number of connected submodules N_{eff} , to compensate the energy lost due to the isolation of faulty submodules:

$$v_c^* = v_c^* \frac{N}{N_{eff}} \tag{6.3}$$

Finally, the algorithm modifies the phase shift of the PS-PWM presented in (2.23), in order to cope with the new number of submodules:



$$\theta = \frac{360}{N_{eff} - 1} \tag{6.4}$$

Figure 6.1: The proposed fault tolerant control algorithm.

It is important to mention that the threshold representing the percentage of failed submodules compared to the total number of arm submodules should be carefully selected according to the number of submodules per arm, the ratings of submodule components and the quality of the output power.

This strategy is designed to permit the converter to continue its operation, provided that the number of faulty submodules per arm does not reach a pre-defined value based on the number of submodules and the ratings of the devices. Otherwise, the control system considers the MMC to be suffering from cascading failure (which is selected to be 25% for the simulated MMC) and in this case, it isolates all submodules and deactivates the converter. The submodule capacitor must be carefully selected, taking into consideration the worst case condition at which the maximum number of faulty submodule can be tolerated. This can be done by calculating the maximum energy variation, the boosted capacitor voltage level and the allowable voltage ripples, and selecting the capacitor value so that it fits the capacitor selection rule presented in (2.32).

This strategy guarantees a stable performance for the MMC under internal abnormal operating conditions, without the need of any redundant components. Moreover, this technique can be applied to an MMC with more than 400 levels, as it is a software technique and can be implemented by simple instructions formed from the required sorting instructions.

Figure 6.2 illustrates a block diagram of the proposed integrated control system of the MMC, which achieves the capacitor voltage balancing technique presented in Chapter 4, the FDU proposed in Chapter 5 and the proposed FTCU presented here.



Figure 6.2: Block diagram of the proposed MMC control strategy.

It is apparent that the proposed scheme eliminates the need for significant numbers of voltage sensors for the submodule capacitors and their associated communication system with the central controllers. Consequently, the proposed strategy renders its application for a MMC system with a large number of submodules.

6.3 Reliability Assessment of the Proposed Fault Tolerant Control Technique

The main outcome of applying a fault tolerant control technique is to enhance the reliability of the entire system. For this purpose, the reliability of the MMC is assessed before and after the application of the proposed fault tolerant control technique, in order to address the amount of enhancement resulting from the proposed fault tolerant control algorithm.

6.3.1 MMC reliability definitions

A system failure can be represented by its probability to occur; this is performed considering the process behaviour as a random variable which receives its value from finite state space elements. To assess the reliability of the MMC, firstly the failure rate of the IGBT λ_0 should be assumed. The reliability function R(t) is the probability of success for the system, which is expressed as:

$$R(t) = 1 - F(t)$$
(6.5)

Where F(t) is the probability of system failure, which is complementary to the R(t), and is given by:

$$F(t) = \int_0^t f(\tau) d\tau \tag{6.6}$$

Where f(t) is rate of change for the failure probability of a certain component, thus (6.6) can be rewritten as [109]:

$$f(t) = \frac{dF(t)}{dt} \tag{6.7}$$

Another important definition is the mean time to failure (MTTF), which gives an indication for the expected life time of a component that cannot be repaired or maintained. The MTTF is expressed as [109]:

$$MTTF = \int_0^\infty R(t)dt \tag{6.8}$$

6.3.2 Reliability calculations for MMC without fault tolerant control

For a system without fault tolerant control, the MMC will fail if one submodule suffers from a fault in either the upper or lower legs. To model the reliability of the converter in this case, a Markov chain is developed, as illustrated in Figure 6.3.



Figure 6.3: Markov chain for a modular multilevel converter without fault tolerant control.

In this case the failure rate for one submodule in a certain leg should equal:

$$\lambda_1 = (2N - 1)\lambda_0 \tag{6.9}$$

Where λ_0 is the failure rate of each IGBT switch.

Based on the Markov chain, and from (6.9) substituting in (6.7), this gives:

$$\frac{dP_1}{dt} = -2\,\lambda_1\,P_1(t) \tag{6.10}$$

Solving the first order differential equation presented in (6.10), the reliability function can be expressed as:

$$R(t) = P_1(t) = e^{-2(2N-1)\lambda_0 t}$$
(6.11)

From (6.11) substituting in (6.8), the mean time to failure is:

$$MTTF_{MMC \ without \ FTC} = \frac{1}{2\lambda_0} \left(\frac{1}{2N-1}\right) \tag{6.12}$$

6.3.3 Reliability calculations for MMC with the redundancy based fault tolerant control

In this case, the reliability of the MMC can be figured out from Figure 6.4 which has two failure rates. The first failure rate when losing one sub-module, λ_1 . While the second failure rate, λ_2 indicates losing two sub-modules in the same leg.



Fig. 6.4. Markov chain of the MMC with the classical redundancy based FTC technique.

In the sub-module redundant based FTC, each leg is assumed to be equipped with one redundant sub-module. Thus, the sub-module failure rates are given by:

$$\lambda_1 = (\frac{2N-1}{2} + 1)\lambda_0 \tag{6.13}$$

$$\lambda_2 = \left(\frac{2N-1}{2}\right)\lambda_0 \tag{6.14}$$

Based on the Markov chain, the probability for the first state P1 will remain the same, as presented in (6.7).

Following the same way, the failure probability of state 2 gives:

$$\frac{dP_1}{dt} = \lambda_1 P_1(t) - \lambda_2 P_2(t)$$
(6.15)

Solving (6.15) gives:

$$P_2(t) = \left(\frac{2N-1}{2}\right) \left(e^{-\left(\frac{2N-1}{2}\right)\lambda_0 t} - e^{-\left(\frac{2N-1}{2}+1\right)\lambda_0 t}\right)$$
(6.16)

From (6.7) and (6.16), the probability of the fail state is the summation of $P_1(t)$ and $P_2(t)$:

$$R(t) = P_1(t) + P_2(t) = \left(\left(\frac{1-2N}{2}\right) e^{-\left(\frac{2N-1}{2}+1\right)\lambda_0 t} \right) + \left(\frac{2N-1}{2}\right) \left(e^{-\left(\frac{2N-1}{2}\right)\lambda_0 t} \right)$$
(6.17)

From (6.17) substituting in (6.12), the MTTF is:

$$MTTF_{MMC with FTC 1} = \frac{1}{\lambda_0} \left(\frac{1-N}{N+1} \right)$$
(6.18)

6.3.4 Reliability calculations for MMC with the proposed fault tolerant control

The reliability of the MMC has a root change after the application of the proposed fault tolerant control, which is due to the ability of the MMC to work under

faults until reaching the cascading failure condition in one arm. To simplify the calculations and give a sense of the results, the reliability calculations are performed for the simulation model, which incorporates four submodules per arm, as previously presented (Table 4.1). In this particular case, the Markov reliability chain is developed as shown in Figure 6.5.



Figure 6.5: Markov chain for a five-level modular multilevel converter with the proposed fault tolerant control.

The Markov reliability chain shows that the MMC, with the proposed FTCU, has four possible finite states before reaching the failure state, instead of one as discussed in the previous section. The possible states are:

- State 1: The healthy state (no faulty submodules)
- State 2: Losing a submodule in the upper arm
- State 3: Losing a submodule in the lower arm
- State 4: Losing two submodules (one in the upper arm and one in the lower arm)
- Fail state: Losing two submodules in the upper arm or two submodules in the lower arm, or both

In this case, the two failure rates should be presented as λ_1 and λ_2 . λ_1 is the failure rate of one submodule, and will be the same as calculated in (6.9). λ_2 is the failure rate for losing two submodules, and should equal:

$$\lambda_2 = (2N - 3)\lambda_0 \tag{6.19}$$

Based on the Markov chain, the probability for the first state P1 will remain the same, as presented in (6.7). Regarding the probability of the second state P2, it can be expressed as follows:

$$\frac{dP_2}{dt} = -(\lambda_1 + \lambda_2)P_2(t) + \lambda_1 P_1(t)$$
(6.20)

The solution of the differential equation given in (6.20) gives:

$$P_2(t) = \frac{2N-3}{2} \left(e^{-(\lambda_1 + \lambda_2)t} - e^{-2\lambda_1 t} \right)$$
(6.21)

 $P_3(t)$ has the same probability of $P_2(t)$, since the conditions are the same.

Moving to the forth state probability, this can also be emphasized from the Markov chain, as follows:

$$\frac{dP_4}{dt} = -(2\lambda_2)P_4(t) + \lambda_1 P_2(t) + \lambda_1 P_3(t)$$
(6.22)

The solution of this differential equation is expressed as:

$$P_4(t) = \frac{(2N-3)^2}{4} \left(e^{-2\lambda_1 t} + e^{-2\lambda_2 t} \right) - \frac{(2N-3)^2}{2} e^{-(\lambda_1 + \lambda_2)t}$$
(6.23)

The reliability function R(t) is the summation of the four probabilities, as follows $R(t) = P_1(t) + P_2(t) + P_3(t) + P_4(t)$ $\binom{(N-3)^2}{2} = (2N-3)^2 + (2N-3)^2$

$$= \left((2N-2) + \frac{(N-3)^2}{4} \right) e^{-2(2N-1)\lambda_0 t} + \left((2N-3) - \frac{(2N-3)^2}{4} \right) e^{-2(2N-2)\lambda_0 t} + \left(\frac{(2N-3)^2}{4} \right) e^{-2(2N-3)\lambda_0 t}$$
(6.24)

From (6.24), substituting in (6.8), the mean time to failure is:

$$MTTF_{MMC with FTC} = \frac{1}{2\lambda_0} \left(\frac{\left((2N-2) + \frac{(2N-3)^2}{4}\right)}{(2N-1)} + \frac{\left((2N-3) - \frac{(2N-3)^2}{4}\right)}{(2N-2)} + \frac{\left(\frac{(2N-3)^2}{4}\right)}{(2N-3)} \right) \quad (6.24)$$

To compare the reliability before and after the application of the proposed fault tolerant control, the reliability density functions presented in (6.11), (6.17) and (6.24) are plotted against time for a five-level MMC. The IGBT failure rate λ_0 is assumed to be $10^{-7}h^{-1}$, which is almost three years [110].

As shown in Figure 6.6, the reliability density function is considerably enhanced, which means that the rate of failure for the MMC is decreased.



Figure 6.6: Plot of reliability density functions before and after the application of the proposed fault tolerant control.

This is also apparent when the MTTF is calculated, which gives 1615 days when applying the proposed fault tolerant control and 998 days when applying the redundancy based fault tolerant control compared to 82 days when the FTCU is deactivated.

6.4 Simulation Results

6.4.1 The FTCU performance under IGBT open circuit fault

This task is devoted to assessing the capabilities of the proposed control strategy, the FDU and the FTCU under an IGBT open circuit fault in a submodule. The upper switch of the third submodule in the upper arm of phase *a* is intentionally opened at t = 0.5s, while enabling the proposed FDU and FTCU. The FTCU isolates the faulty submodule, and the third submodule in the lower arm, to balance the energy between the two arms. Furthermore, the FTCU increases the v_c^* from 2.25 kV to 3 kV (Figure 6.7), and adjusts the phase shift of the carrier waves from 45° to 60°, in order to compensate for the loss of two isolated submodules. These actions are sufficient to tolerate the fault and restore the stability of the MMC control system, as apparent from the results seen in Figures 6.8 (a) and (b), where the threephase output voltages and currents are balanced. The capacitor voltages of the remaining submodules restore their balance around the new desired value set by the FTCU, as indicated in Figure 6.9 (c). As expected, the capacitor voltages increase slightly at the moment of fault detection, t = 0.533s. It should be noted that the capacitor voltages of the isolated submodules freeze at their values when isolation occurs. Figures 6.10 and 6.11 demonstrate that the proposed RLS algorithm adequately estimates the capacitor voltages during the fault. The ADALINE and RLS algorithms provide the FDU and the FTCU with accurate data that result in precise performance during transient periods. With the help of the FTCU, the averaging and balancing control loops are able to quickly overcome the consequences of a fault, and forces the differential current to reach its rated value prior to the occurrence of the fault, as displayed in Figure 6.11 (a). Moreover, the averaging and balancing control signals retain their stability, as illustrated in Figures 6.11 (b) and (c), respectively.



Figure 6.7: Dynamic performance of the proposed FTCU under an open-circuit fault in a submodule.



Figure 6.8: Enabling the proposed FTCU for an open-circuit fault: a) Three-phase voltages, b) three-phase load currents, c) voltages of leg a submodules.



Figure 6.9: Actual and estimated voltages of the upper arm submodules of phase a, under an open-circuit fault while enabling the proposed FTCU.



Figure 6.10: Actual and estimated voltages of the lower arm submodules of phase *a*, under an open-circuit fault while enabling the proposed FTCU.



Figure 6.11: Action of the different controllers for leg a with enabling the FTCU for an open-circuit fault: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm.

6.4.2 The FTCU Performance under IGBT short circuit fault

This case is conducted in order to investigate the capabilities of the proposed control strategy, the FDU and the FTCU under an IGBT short circuit fault in a submodule. A short-circuit fault is applied to the upper switch of the third submodule in the upper arm of phase *a* at t = 0.5s, while enabling the proposed FDU and FTCU. At the instant of fault detection, the FTCU isolates the faulty submodule in the upper arm, and its corresponding third submodule in the lower arm, in order to balance the energy between the two arms. Furthermore, the FTCU increases the v_c^* from 2.25 kV to 3 kV, as displayed in Figure 6.12, and modifies the phase shift of the carrier waves from 45° to 60° to compensate for the loss of two submodules. Figures 6.13 (a) and (b) show the three-phase output voltages and currents, which are balanced. The capacitor voltages of the remaining submodules restore their balance at the new set value, as illustrated in Figure 6.13 (c). The proposed RLS scheme accurately estimates the capacitor voltages during a short-circuit fault, as indicated in Figures

6.14 and 6.15. Figure 6.16(a) shows that the FTCU stabilizes the operation of the averaging and balancing control loops, which succeed in limiting the magnitude of the differential current, and tightly track its reference. Additionally, the improved averaging and balancing control signals are portrayed in Figures 6.16 (b) and (c), respectively.



Figure 6.12: Dynamic performance of the proposed FTCU under a short-circuit fault in a submodule.



Figure 6.13: Enabling the proposed FTCU for a short-circuit fault: a) Three-phase voltages, b) three-phase load currents, c) voltages of leg a submodules.



Figure 6.14: Actual and estimated voltages of the upper arm submodules of phase *a*, under a short-circuit fault.



Figure 6.15: Actual and estimated voltages of the lower arm submodules of phase *a*, under a short-circuit fault.



Figure 6.16: Action of the different controllers for leg a with enabling the FTCU for a short-circuit fault: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm.

6.4.3 Performance of the FTCU under multiple faults

This case is devoted to investigating the capabilities of the proposed control strategy, the FDU and the FTCU under multiple IGBT faults in different submodules. Firstly, an open-circuit fault is applied to the upper switch of the third submodule in the upper arm of phase *a* at t = 0.5s. Then, at t = 0.7s, a short-circuit fault is applied to the second submodule in the lower arm of phase *a* while enabling the proposed FDU and FTCU. At the instant of the first fault detection, the FTCU isolates the faulty submodule in the upper arm, and its corresponding third submodule in the lower arm, to balance the energy between the two arms. Furthermore, the FTCU increases the v_c^* from 2.25 kV to 3 kV, and modifies the phase shift of the carrier waves from 45° to 60°, in order to compensate for the loss of two submodules. At the moment of detection of the second fault, the FTCU instantaneously isolates the second submodule in the lower arm of phase *a* which is faulty, and re-inserts the third submodule in the same arm which has been previously

bypassed as a response from the FTCU against the first fault, thus the energy between the two arms is always balanced. Figures 6.17 (a) and (b) show the three-phase output voltages and currents which are balanced during the entire period, and after the application of the two faults. The capacitor voltages of the remaining submodules restore their balance at the new set value, as illustrated in Figure 6.17(c). The proposed RLS scheme accurately estimates the capacitor voltages during the entire period, as indicated in Figures 6.18 and 6.19. Figure 6.20 (a) shows that the FTCU stabilizes the operation of the averaging and balancing control loops during the two faults, which succeeds in limiting the magnitude of the differential current and tightly tracking its reference. Additionally, the improved averaging and balancing control signals are portrayed in Figures 6.20 (b) and (c), respectively.



Figure 6.17: Enabling the proposed FTCU for multiple faults: a) Three-phase voltages, b) three-phase load currents, c) voltages of leg a submodules.



Figure 6.18: Actual and estimated voltages of the upper arm submodules of phase a, under multiple faults.



Figure 6.19: Actual and estimated voltages of the lower arm submodules of phase a, under multiple faults.



Figure 6.20: Action of the different controllers for leg a with enabling the FTCU for multiple faults: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm.

6.4.4 Performance of the FTCU under cascading failures

This case is performed in order to investigate the capabilities of the proposed control strategy, the FDU and the FTCU under cascading failure. Firstly, an opencircuit fault is applied to the upper switch of the third submodule in the upper arm of phase *a* at t = 0.5s. Then, at t = 0.55s, another short-circuit fault is applied to the forth submodule in the same arm whilst enabling the proposed FDU and FTCU. At the instant of the first fault detection, the FTCU isolates the faulty submodule in the upper arm, and its corresponding third submodule in the lower arm, in order to balance the energy between the two arms. Furthermore, the FTCU increases the v_c^* from 2.25 kV to 3 kV, and modifies the phase shift of the carrier waves from 45° to 60°, in order to compensate for the loss of two submodules. At the moment of the second fault, the FTCU considers this condition as a cascading failure condition, since the percentage of faulty submodules to the total number of submodules in the upper arm of phase *a* is higher than 25%. As a result, the FTCU isolates all submodules in the three legs and blocks the converter. Figure 6.21 shows the reference of capacitor voltages, which increased from 2.25 to 3 kV after the first fault, while after the second fault are decreased to 0 kV, in order to block the MMC. Figures 6.22 (a) and (b) show the three-phase output voltages and currents which are balanced after the application of the first fault, but after the second fault, the three phase voltages and currents progress towards zero. The capacitor voltages of the remaining submodules restore their balance at the new set value after the first fault, and then remain constant due to the isolation of the MMC, as a result of the second fault, as illustrated in Figure 6.22(c). The proposed RLS scheme accurately estimates the capacitor voltages until the application of the second fault, as indicated in Figures 6.23 and 6.24. Figure 6.25 (a) shows that the FTCU stabilizes the operation of the averaging and balancing control loops during the application of the first fault, which succeeds in limiting the magnitude of the differential current and tightly tracking its reference. Additionally, the improved averaging and balancing control signals are portrayed in Figures 6.25 (b) and (c), respectively. However, after the detection of the second fault, the output signals of the proposed controller are nulled, due to the MMC blocking order.



Figure 6.21: Dynamic performance of the proposed FTCU under cascading failure.



Figure 6.22: Enabling the proposed FTCU for cascading failures: a) Three-phase voltages, b) three-phase load currents, c) voltages of leg a submodules.



Figure 6.23: Actual and estimated voltages of the upper arm submodules of phase a, under cascading failures.



Figure 6.24: Actual and estimated voltages of the lower arm submodules of phase a, under cascading failures.



Figure 6.25: Action of the different controllers for leg a with enabling the FTCU for cascading failures: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first submodule of the upper arm.

6.5 Chapter summary

The chapter has presented a new fault tolerant control strategy for the MMC which is based on modifying the parameters of the capacitor voltage balancing technique presented in Chapter 4. The proposed method is integrated with the fault detection algorithm introduced in Chapter 5 to form a complete inner controller for the MMC. This action results in a significant reduction of the number of sensors and their associated wiring issues. Furthermore, the reliability enhancement of the MMC with the proposed FTCU is studied using Markov chains, and was significantly improved, which is reflected in an increased MTTF. Accurate performance of the proposed fault tolerant-based control strategy is revealed from the results presented and discussed in this chapter. In addition, these results prove that the proposed fault tolerant SMs at a new set value, and limits the circulating current during SM faults.

Chapter 7

Validation of Proposed Techniques Using Hardware in The Loop Real-Time Simulations

7.1 Introduction

This chapter presents the validation of the proposed capacitor voltage balancing, fault detection and tolerant control techniques using a HIL simulation concept, in order to practically evaluate the performance of the proposed control. This is done by using a real-time digital simulator connected to a real-time field programmable gate array (FPGA) based physical controller. Firstly, the concept of HIL simulation systems is presented with focus on their advantages over offline simulations and experimental setups. Then, the platform of the HIL used for testing the proposed techniques is described, showing its main parts. A comparative study is also conducted, in order to illustrate the practical assessment of the proposed techniques when applied to a real MMC project with a large number of voltage levels. Finally, different case studies are applied to the HIL system, and their corresponding results are analysed and compared with those captured from the offline PSCAD simulations.

7.2 HIL System Description

Although off-line simulations are accurate modelling techniques for power systems, they may suffer from various difficulties such as high computational time, particularly for an application such as MMC, which is composed of a large number of series connected submodules. Moreover, they cannot precisely illustrate the inoperation behaviour of real controllers under different operating conditions [111]. HIL simulations can be used as a suitable alternative for testing and prototyping newly developed control techniques that will be applied to power electronic devices. The HIL simulator is defined as a setup that emulates a system, by immersing faithful physical replicas of some of its subsystems within a closed-loop virtual simulation of the remaining subsystems[112]. This means that the basic task of the HIL simulator is to capture the bi-directional or closed loop interactions between a physical controller and the virtual simulation. As shown in Figure 7.1, the HIL simulator.



Figure 7.1: Structure of the HIL platform.

The main advantages of the HIL simulations can be summarized in the following points [113], [114]:

- Cost effective due to the reduced hardware when compared to experimental circuits
- Since it is a simulation-based system, rapid prototyping is available as it does not need hardware modifications
- The ability to run on a real-time basis due to its advanced computational structure
- HIL simulators are comprehensive and repeatable which means that they can be used in the testing of controllers that require a variety of operating conditions

• Finally, its most two important features are non-destructiveness and safety, which is very important in the research since it depends on applying faults on the MMC

7.3 HIL System Implementation

The MMC system is simulated using the RT-LAB [111], and downloaded to the OP-4510 real time digital simulator [115], while the proposed MMC control technique and its associated capacitor voltage estimation units are programmed using the LAB-VIEW software [116], and uploaded to an FPGA physical controller type cRIO-9024 [117], as shown in Figure 7.2. Appendix C shows the details of the HIL simulations.



(b)

Figure 7.2: The HIL platform system components: a) the block diagram of the HIL platform, b) a snapshot of the HIL system.

The system parameters are identical to the PSCAD-simulated systems that are presented in Table 4.1. The following sub-sections illustrate each part of the platform.

7.3.1 The real-time physical controller

The real-time controller shown in Figure 7.3 is a cRIO system type 9024 reconfigurable controller made by National Instruments [117]. This controller contains an internal real-time process and FPGA controller type virtex 5 [118], which combines high speed calculations with the ability to deal with various types of physical signals. Moreover, it has 8 slots where different modules can be fitted, offering a wide range of analogue and digital cards that can be utilized for both signal acquisition and measurements.



Figure 7.3: The cRIO-9024 real-time controller.

The real-time controller is responsible for controlling the MMC in very small time steps (25 ns) by applying the proposed control technique and then generating the submodules' switching signals. As shown in Figure 7.4, the main advantage of using FPGA controllers over processor-based systems is the ability to implement the application logic in hardware circuits, instead of using a complex operating system, which saves a significant amount of time and enables it to be competitive for real-time control applications. It is programmed using LABVIEW, in order to demonstrate 4 control loops through which the controller performs its desired functions. The tasks of these loops are:

- Submodule capacitor voltage estimation techniques (ADALINE and RLS)
- The proposed capacitor voltage-balancing algorithm
- The submodule fault detection algorithm

To achieve these functions, the real-time controller must receive from the real-time digital simulator the output voltage and the arm voltages. Further, it should output the firing signals required for switching the MMC submodules.



Figure 7.4. The application logic inside the FPGA-based controller [119], [120].

7.3.2 The real-time digital simulator

The real-time digital simulator used in the HIL platform is the OP4510, made by OPAL-RT (Figure 7.5).



Figure 7.5: The OP4510 real-time digital simulator.

It is responsible for simulating the MMC power system and exchanging data with the real-time controller through its hardware ports. As shown in Figure 7.6, the OP4510 consists of three main parts:

- 1. The KINTEX 7 FPGA carrier board is responsible for controlling the I/O ports, which enables the system to exchange signals with the physical controllers demonstrating different signal types.
- 2. The multi-core CPU, which builds the state space representation for the MMC system, and performs the real-time calculations that are necessary for system modelling

3. The Ethernet port, which connects the OP4510 with a host PC for the purpose of changing the system parameters



Figure 7.6: The OP4510 internal structure [115].

The OP4510 is programmed using the RT-LAB software package and MATLAB/SIMULINK [121]. The process starts with modelling the MMC and its attachments using SIMULINK, including the hardware signals that need to be inputted or outputted from the simulator. The RT-LAB is then responsible for synthesizing the system, building the bootable file which will be loaded into the simulator, and then running the real-time simulation. Looking inside the simulation, the real-time simulation model expects to receive the IGBT's firing signals from the cRIO controller while sending the arm inductor voltages and the AC output voltage. It is important to mention that the whole system is running at a very low time step (25 μ s), which makes it convenient for real-time testing.

7.4 Practical Assessment of the Proposed Techniques

The main aim of building a HIL platform is an ability to assess any proposed control technique by determining whether it can be implemented industrially or not.

Also the use of HIL simulations ascertains the feasibility of any developed algorithms. Regarding the proposed techniques for MMC inner control, fault detection and tolerant control algorithms, the great challenge in implementing these control functions is the computational volume required, as all tasks depend on the estimation of the capacitor voltages for all submodules, which may total at least 800 per leg. If the computational burden is too high, a computer with a very large processor is required, and in this case the control technique would need code optimization and several reductions in the calculations. For this purpose, the computational burden of the simulated MMC is presented, and based on this a conception is shown for a real MMC project with 800 submodules per arm. The computational burden can be calculated by the number of logical resources consumed inside the FPGA. The building unit of an FPGA is the lookup table (LUT). As shown in Figure 7.7, a 4-input LUT is a small piece of RAM which is connected in the form of combinational logic to perform a specific function. The relationship between the 4 inputs and the outputs are simply a truth table, which is built through the synthesizing process.



Figure 7.7: 4-way LUT.

The implementation of the proposed control on the cRIO virtex 5 based FPGA controller for controlling a MMC with 8 modules per arm consumed 19,872 LUTs, with the physical controller having an embedded FPGA system with a maximum number of 28800 LUTs, which means that only 70% of the FPGA's resources are utilized. This means that when selecting a controller for an MMC with 800 submodules per leg, applying the proposed control technique will require 1,987,200 LUTs. This can easily be achieved by using three FPGA controller type VIRTEX 7s, which have 2,000,000 LUTs inside their processing unit [119], [120]. Therefore, 99% of the logic capacity of the controller is utilized, which means that

the proposed control techniques can be implemented practically in terms of computational burden.

7.5 HIL Simulation Results

All case studies conducted using PSCAD simulations were repeated using HIL simulation, in order to examine the performance of the proposed techniques and compare the acquired results in both simulation systems.

7.5.1 Case 1: The performance of the proposed control under dynamic change of output voltage reference

This case is identical to that presented in sub-section 4.4.1, and is dedicated to examining the performance of the proposed capacitor voltage estimation-based control algorithm for the MMC under the dynamic change of the reference phase voltage. The reference capacitor voltage v_c^* is set at $v_{dc}/N = 2.25$ KV, while the reference phase voltage signal v_o^* is dynamically changed from 0.7pu to 1 pu.

The measured three-phase AC voltages follow their reference signals, as illustrated in Figure 7.8(a). Initially, when $v_o^* = 0.7$ pu, six submodules are utilized, and when $v_o^* = 1$ pu, all submodules are encompassed. Figure 7.8(b) displays the three-phase load currents. Figures 7.9(a) and 7.9(b) trace the capacitor voltages, which are grouped into two main trajectories, one for the upper arm and the other for the lower arm. The proposed MMC controller succeeds in balancing the capacitor voltages at their reference value of 2.25 KV. Moreover, Figure 7.10 (a) displays the estimated capacitor voltage of the first submodule, and compares it with its corresponding actual voltage. Fast tracking with accurate performance of the proposed scheme, based on the RLS algorithm for estimating submodule capacitor voltages, is evident. Furthermore, the circulating current tightly tracks its reference signal, as demonstrated in Figure 7.10(b). Figures 7.11(a) and 7.11(b) illustrate the actions of the averaging and balancing controllers for the first submodule at the upper arm $v_{cu1 \ bal}^*$, respectively. These results reveal the efficient utilization of the MMC.



Figure 7.8: Performance of the proposed inner controller under dynamic change of v_o^* : a) Three-phase voltages, b) three-phase load currents.



Figure 7.9: Performance of the proposed inner controller under dynamic change of v_o^* : a) submodules voltages of upper arm in leg a, b) submodules voltages of lower arm in leg a.



Figure 7.10: Action of the different controllers for leg a during dynamic change of v_o^* : a) Actual voltage of submodule 1 and its estimated signal, b) Actual and reference waveform of the circulating current.



Figure 7.11: Action of the different controllers for leg a during dynamic change of v_0^* : a) averaging voltage reference, b) modulating signal for the first submodule of the upper arm.

7.5.2 Case 2: The performance of the proposed under IGBT open circuit fault

This case is similar to that presented in sub-section 6.4.1, with the lower IGBT of the third submodule being opened to resemble an open-circuit fault condition. The results are in contrast with PSCAD simulations. As shown in Figure 7.12, the proposed FDU successfully detected the fault after 35.6 ms, compared to 33 ms in off-line simulations. This increase in detection time is logical, as the physical controller runs at a definite clock frequency. The FTCU succeeds in restoring the stability of the MMC directly after the detection of the fault. This is performed by isolating the faulty submodule, and one healthy submodule, from the other arm to balance the energy. The reference signal of the capacitor voltage, v_c^* , is then increased from 2.25 to 3 kV, which compensates the lost energy of the faulty submodule. Moreover, the phase shift between carrier signals is increased from 45° to 60° . This appears in the three phase voltages and currents shown in Figures 7.13 (a) and (b), respectively, since they are free from DC components and distortion. Also, the capacitor voltages have their reference signals, v_c^* , tracked in a balanced manner over the entire test period, as illustrated in Figure 7.14. The RLS unit estimates all healthy capacitor voltages, with an average error significantly lower than 5%, as shown in Figure 7.15 (a). Regarding the estimation of the faulty submodule, the estimated signal contains large amounts of steady state error, which helps the FDU quickly detect the fault, as demonstrated in Figure 7.15 (b). The proposed capacitor voltage balancing succeeds in controlling the MMC during the
fault with the aid of the FTCU. This appears in the reduced value of the differential current, i_{diff} , which strictly tracks its reference signal, as given in Figure 7.16 (a). Moreover, the averaging and balancing control signals are stable over the entire period, with only minor fluctuations when the fault is initially generated, as indicated in Figures 7.16 (b) and (c).



Figure 7.12: Dynamic performance of the proposed control under an open-circuit fault in a submodule.



Figure 7.13: Performance of the proposed control for an open-circuit fault: a) Three-phase voltages, b) three-phase load currents.



Figure 7.14: Performance of the proposed control for an open-circuit fault: a) submodule voltages of upper arm in leg a, b) submodule voltages of lower arm in leg a.



Figure 7.15: RLS-estimated signals in the upper arm of leg a under open-circuit fault: a) first submodule (healthy), b) third submodule (faulty).



Figure 7.16: Action of the different controllers for leg a with enabling the proposed control for an open-circuit fault: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) modulating voltage reference for the first submodule of the upper arm.

7.5.3 Case 3: The performance of proposed control under IGBT short circuit fault

This case is similar to that presented in sub-section 6.4.2, with the lower IGBT of the third submodule being opened to resemble an open-circuit fault condition. The results are in contrast with PSCAD simulations. The FDU is able to detect the fault in less than 3.2 ms, which is higher that the PSCAD simulations by

only 1 ms, due to the real-time calculations of the FPGA controller, as demonstrated in Figure 7.17. After fault detection, the proposed FTCU performed the same actions mentioned in the previous case to maintain stability. These actions led to stable performance of the MMC before and after fault application. The quality of the output voltage and current was not affected, as shown in Figure 7.18. Moreover, the capacitor voltages of MMC submodules tracked their reference with only minor differences between each other, as illustrated in Figure 7.19. Again, the RLS estimation unit was able to accurately estimate capacitor voltages of all healthy MMCs with a very small amount of error. However, this was not the case for faulty submodules, as there was a considerable error between the RLS and ADALINEestimated voltages, which assisted the FDU in its task. Figure 7.20 shows a sample for estimated voltages of one healthy submodule, and another one which is faulty. Furthermore, the proposed capacitor voltage balancing controller offers stable performance, which helps to reduce differential current, as given in Figure 7.21.



Figure 7.17: Dynamic performance of the proposed control under a short-circuit fault in a submodule.



Figure 7.18: Enabling the proposed control for a short-circuit fault: a) Three-phase voltages, b) three-phase load currents.



Figure 7.19: Enabling the proposed control for a short-circuit fault: a) submodule voltages of upper arm in leg a, b) submodule voltages of lower arm in leg a.



(a) (b) Figure 7.20. RLS-estimated signals in the upper arm of leg a under short circuit: a) first submodule (healthy), b) third submodule (faulty).



Figure 7.21: Action of the different controllers for leg a with enabling the proposed control for a short-circuit fault: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) modulating voltage reference for the first submodule of the upper arm.

7.5.4 Case 4: The performance of the proposed control under multiple submodule faults

This case is identical to the PSCAD-offline simulated case presented in subsection 6.4.3, and is devoted to studying the behaviour of the real-time physical behaviour of the proposed techniques in the case of having two faults. Firstly, an open-circuit fault is applied to the upper switch of the third submodule in. the upper arm of phase a, then after 0.3s a short-circuit fault is applied to the first submodule in the lower arm of phase *a* while enabling the proposed FDU and FTCU. The FDU succeeds in detecting the first fault after 35.4 ms, as shown in Figure 7.22(a). Moreover, the FDU identifies the second fault after 3.2 ms, as shown in Figure 7.22 (b). The detection time of the two faults was similar to that in PSCAD-offline simulations, with very small deviation. Figures 7.23(a) and (b) portray the threephase output voltages and currents, which are balanced even during the application of the two faults. The capacitor voltages of the remaining submodules of the upper and lower arms restore their balance at the new set value, as illustrated in Figures 7.24(a) and (b), respectively. Consequent to the detection of the first fault, the proposed FTCU isolates the faulty submodule in the upper arm, and its corresponding third submodule in the lower arm, in order to balance the energy between the two arms. Furthermore, the FTCU increases v_c^* from 2.25 kV to 3 kV, in order to compensate for the loss of two submodules. After the second fault is identified, the FTCU isolates the first submodule in the lower arm of phase a, which is faulty, and re-inserts the third submodule in the same arm, which is previously bypassed as a response from the FTCU against the first fault. Hence, the energy between the two arms is always balanced, and the capacitor voltages of the remaining submodules are regulated at the new set value, 3KV. The proposed RLS scheme accurately estimates the capacitor voltages during different operating conditions, as indicated in Figure 7.25(a). Figure 7.25(b) demonstrates that the FTCU succeeds in limiting the magnitude of the differential current, which tightly tracks its reference signal. Finally, the averaging and output voltage reference control signals are portrayed in Figures 7.26 (c) and (d), respectively. It is apparent that the proposed FTCU stabilizes the operation of the control loops during different types of faults. The results illustrate the satisfactory behaviour of the proposed control technique under different types of switch faults.



Figure 7.22: Dynamic performance of the proposed control under multiple faults: a) fault 1, b) fault 2.



Figure 7.23: The performance of the proposed control under multiple faults: a) Three-phase voltages, b) three-phase load currents.



Figure 7.24: The performance of the proposed control under multiple faults: a) submodule voltages of upper arm in leg a, b) submodule voltages of lower arm in leg a.



Figure 7.25: Action of the different controllers for leg a with enabling the proposed control for multiple failure: a) The capacitor voltage estimation of submodule Vcu1, b) actual and reference waveform of the circulating current.



Figure 7.26: Action of the different controllers for leg a with enabling the proposed control for multiple failure: a) averaging voltage reference, b) modulating voltage reference for the first submodule of the upper arm.

7.5.5 Case 5: The performance of the proposed control under cascading failures

This case is identical to that presented in sub-section 6.4.4 under PSCADoffline simulations for examining the real-time performance under cascading failures. A scenario of cascading failure is considered in this case study, in order to investigate the capabilities of the proposed control strategy. Firstly, an open-circuit fault is applied to the upper switch of the third SM in the upper arm of phase *a*. Then, after 0.5*s*, a short-circuit fault is applied to the fourth SM in the same arm. After 45 ms, the FDU successfully detects the open-circuit fault, as indicated in Figure 15(a). Instantaneously, the FTC sorting algorithm isolates the faulty SM in the upper arm, and its corresponding third SM in the lower arm, in order to balance the energy between the two arms. In addition, the proposed FTC increases v_c^* from 2.25 kV to 3 kV, and modifies the phase shift of the carrier waves from 45° to 60° to compensate for the loss of two SMs. The FDU detected the short-circuit fault after 3.3 ms.

The FTC sorting algorithm considers this condition as a cascading failure, since the percentage of faulty SMs in the upper arm of phase *a* reaches 25% of the total number of SMs. As a result, the FTC isolates the SMs in the three legs and blocks of the MMC. Figures 7.27 (a) and (b) portray the three-phase output voltages and currents, which are balanced even during the first fault. After the second fault, the three-phase voltages and currents are damped to zero, and the MMC is blocked. Figures 7.28 (a) and (b) illustrates that the capacitor voltages of the remaining SMs restore their balance at the new set value after the first fault. Following the second fault, the capacitors become floating, due to the converter blockage.

The proposed RLS scheme accurately estimates the capacitor voltages, as indicated in Figure 7.29 (a). It should be noted that after the second fault, where the MMC is blocked, the estimated capacitor voltages tend to zero, due to assigning zero switching signals to the SMs. Subsequently, the capacitors become floating and their voltages cannot be estimated. Figure 7.29 (b) shows that the proposed strategy succeeds in limiting the magnitude of the differential current, which tightly tracks its reference signal. Additionally, the averaging voltage reference, and the modulation signal for the first SM of the upper arm, are portrayed in Figures 7.30 (a) and (b), respectively.

The results demonstrate the accurate dynamic behaviour and fast response of the proposed FDU and FTC under cascading failure. The proposed strategy stabilizes the operation of the averaging and balancing control loops during SM faults.



Figure 7.27: The performance of the proposed strategy under cascading failure: a) three-phase voltages, b) three-phase load currents.



Figure 7.28: The performance of the proposed strategy under cascading failure: a) capacitor voltages of the upper arm in phase a, b) capacitor voltages of the lower arm in phase a.



Figure 7.29: Action of the different controllers for leg a: a) The capacitor voltage estimation of SM Vcu1, b) Actual and reference waveforms of the circulating current.



Figure 7.30: Action of the different controllers for leg a under cascading failures: a) averaging voltage reference, b) Modulation signal for the first SM of the upper arm.

7.6 Chapter Summary

This chapter has presented the validation of the proposed capacitor voltage balancing, fault detection and fault tolerant control techniques using HIL real-time simulation. Based on the memory consumed in the FPGA controller used in the HIL platform, a study for the controller design in case of applying the proposed techniques for a MMC with 401 voltage levels is calculated, including the sizing of memory resources to investigate the suitability of the proposed techniques when applied to a MMC with high number of voltage levels. This study confirms that the proposed techniques are highly suitable for industrial implementation. Finally, HIL real-time results illustrate the robust behaviour of the proposed techniques, which are similar to PSCAD-offline simulations.

Chapter 8

Conclusion

8.1 Conclusion

This thesis presents two estimation units based on ADALINE and RLS algorithms for capacitor voltages of the submodules of the MMC. For each leg of the MMC, the proposed estimation unit requires only three sensors; one for the phase voltage and the others for the voltages across the arm reactors. Moreover, the proposed RLS estimation unit is integrated with a control strategy, in order to balance the voltages across the submodules of the MMC. Furthermore, the estimated capacitor voltages from the two techniques are utilized to detect and localize the open-circuit and short-circuit faults of submodule switches. Finally, it proposes a FTCU depending on a sorting algorithm, which is attached to the proposed capacitor voltage balancing and fault detection techniques, in order to form an integrated MMC inner controller. After isolating the faulty submodules, the proposed FTCU boosts the capacitor voltage, and modifies the phase shift between the carrier signals for the rest of submodules in the leg, in order to cope with the reduced number of effective submodules. The proposed integrated control strategy eliminates the need for direct measurements of capacitor voltages and their associated communication systems, which renders it suitable for implementing a low-cost centralized controller for the MMC with a large number of submodules. In addition, the proposed FDU and FTCU do not require any extra components, as they are based on the estimation algorithms that are simultaneously utilized for the proposed control of the MMC.

The fast response and accurate tracking of the proposed scheme for submodule capacitor voltage estimation are revealed from the results of offline simulations, as well as HIL real-time simulations, where the proposed control is tested under different dynamic conditions. Moreover, the proposed MMC controller succeeds in balancing the capacitor voltages. The accurate performance of the proposed FDU for detecting and localizing different faults, results from its hybrid processing, where the fault decision is not only based on the reduction on the estimated per-unit voltage, but also on the rate of change of the difference between the estimated voltages from the proposed ADALINE and RLS algorithms. Furthermore, the proposed FTCU succeeds in stabilizing the operation of the averaging and balancing control loops of the MMC under different types of faults. In addition, the circulating current tightly tracks its reference signal.

8.2 Summary of Contributions

The summaries of contributions achieved in the thesis are listed below:

- Conducting a detailed review of the fault detection and tolerance control of MMCs (Chapters 2 and 3). This review includes the structure, operation and modulation of MMCs. Moreover, common capacitor voltage balancing techniques proposed in the past are investigated. In addition, all MMC submodule faults are mapped with a particular focus on the consequences associated with each fault type. Finally, popular fault diagnosis and tolerance control techniques used for MMCs are also investigated, in order to determine the research areas that need to be further investigated
- Developing a capacitor voltage balancing technique for an MMC that depends on capacitor voltage estimation, without the need to measure the actual voltage in each submodule (Chapter 4), the proposed technique gives a better performance compared to other capacitor voltage techniques, since capacitor voltages are balanced under different operating conditions, keeping the output power free of distortions at a very low circulating current.

- Developing an improved fault diagnosis technique for detecting submodule faults which is able to detect and localize submodule faults without adding extra sensors or special power circuits, and without affecting the power quality of the MMC (Chapter 5), the proposed technique succeeds in detecting different submodule faults with a rapid response. Furthermore, it differentiates between fault events and outer system disturbances
- Developing a software fault tolerant control technique for controlling the MMC in the case of occurrence of any submodule fault (Chapter 6), the proposed technique does not require any redundant components. This means that costs are minimized, keeping the MMC secured during the fault incident, leading to increased MMC reliability, since its availability is increased.

8.3 **Recommendations for Future Research**

The following research points could be investigated in future:

- The stability of MMC-based HVDC grids could be investigated after applying the proposed MMC inner control, in order to assess the effect of fault tolerance in power system stability
- The proposed capacitor voltage balancing technique and FTCU uses the PS-PWM modulation technique. Other modulation techniques could be studied, in order to determine the most suitable switching algorithm for this application
- The proposed FDU is designed to detect faults inside power electronic devices. A condition monitoring unit for health monitoring of submodules could be investigated for the purpose of detecting faults inside submodule capacitors
- Another FDU for the detection of external DC faults could be attached to the proposed FDU, in order to offer an integrated FDU for detection of internal as well as external faults
- The proposed MMC inner control could be tested industrially, by applying various industrial operating conditions (e.g. EMI, noise, etc.).

References

- R. L. Sellick and M. Åkerberg, "Comparison of HVDC Light (VSC) and HVDC Classic (LCC) site aspects, for a 500MW 400kV HVDC transmission scheme," in 10th IET International Conference on AC and DC Power Transmission (ACDC 2012), 2012, pp. 1–6.
- [2] N. Ahmed, A. Haider, D. V. Hertem, L. Zhang, and H. P. Nee, "Prospects and challenges of future HVDC SuperGrids with modular multilevel converters," in *Proceedings of the 2011-14th European Conference on Power Electronics and Applications (EPE 2011)*, 2011, pp. 1–10.
- [3] Peter Fairley, "New developments in high-voltage DC electronics could herald an epic shift in energy delivery?," *IEEE Spectrum*, 29-Apr-2013.
- [4] S. Debnath, J. Qin, B. Bahrani, M. Saeedifard, and P. Barbosa, "Operation, Control, and Applications of the Modular Multilevel Converter: A Review," *IEEE Trans. Power Electron.*, vol. 30, no. 1, pp. 37–53, Jan. 2015.
- [5] L. Ängquist, A. Haider, H. P. Nee, and H. Jiang, "Open-loop approach to control a Modular Multilevel Frequency Converter," in *Proceedings of the 2011-14th European Conference on Power Electronics and Applications (EPE 2011)*, 2011, pp. 1–10.
- [6] M. Winkelnkemper, A. Korn, and P. Steimer, "A modular direct converter for transformerless rail interties," in 2010 IEEE International Symposium on Industrial Electronics, 2010, pp. 562–567.
- [7] M. Spichartz, V. Staudt, and A. Steimel, "Modular Multilevel Converter for propulsion system of electric ships," in 2013 IEEE Electric Ship Technologies Symposium (ESTS), 2013, pp. 237–242.
- [8] M. Coppola, A. D. Pizzo, and D. Iannuzzi, "A power traction converter based on Modular Multilevel architecture integrated with energy storage devices," in *Electrical Systems for Aircraft, Railway and Ship Propulsion (ESARS), 2012*, 2012, pp. 1–7.
- [9] K. Meah and S. Ula, "Comparative Evaluation of HVDC and HVAC Transmission Systems," in *IEEE Power Engineering Society General Meeting*, 2007, 2007, pp. 1–5.
- [10] Y. Ye, M. Kazerani, and V. H. Quintana, "Current-source converter based STATCOM: modeling and control," *IEEE Trans. Power Deliv.*, vol. 20, no. 2, pp. 795–800, Apr. 2005.

- [11] N. Stretch, M. Kazerani, and R. E. Shatshat, "A Current-Sourced Converter-Based HVDC Light Transmission System," in 2006 IEEE International Symposium on Industrial Electronics, 2006, vol. 3, pp. 2001–2006.
- [12] S. Nomura and J. A. Wiik, "Feasibility Study on Current Source Power Conversion for Superconducting Magnets Using Series Compensated Thyristor Converters," *IEEE Trans. Appl. Supercond.*, vol. 22, no. 3, pp. 5400804– 5400804, Jun. 2012.
- [13] Y. Suh and C. Kim, "A study on high current rectifier systems with mitigated time-varying magnetic field generation at ac input and dc output busbars," in 2011 Twenty-Sixth Annual IEEE Applied Power Electronics Conference and Exposition (APEC), 2011, pp. 125–132.
- [14] E. W. Kimbark, *Direct current transmission*. Wiley-Interscience, 1971.
- [15] J. Arrillaga, *High Voltage Direct Current Transmission*. IET, 1998.
- [16] M. Khatir, Z. S. Ahmed, S. Hadjeri, F. M. Karim, and R. Amiri, "Performance Analysis of a Voltage Source Converter (VSC) based HVDC Transmission System under Faulted Conditions," *Leonardo J. Sci.*, no. 15, pp. 33–46, Dec. 2009.
- [17] T. Orłowska-Kowalska, F. Blaabjerg, and J. Rodríguez, Eds., Advanced and Intelligent Control in Power Electronics and Drives, vol. 531. Cham: Springer International Publishing, 2014.
- [18] S. Ruihua, Z. Chao, L. Ruomei, and Z. Xiaoxin, "VSCs based HVDC and its control strategy," in 2005 IEEE/PES Transmission Distribution Conference Exposition: Asia and Pacific, 2005, pp. 1–6.
- [19] A. J. Korn, M. Winkelnkemper, P. Steimer, and J. W. Kolar, "Capacitor voltage balancing in modular multilevel converters," in 6th IET International Conference on Power Electronics, Machines and Drives (PEMD 2012), 2012, pp. 1–5.
- [20] M. Hagiwara and H. Akagi, "Control and Experiment of Pulsewidth-Modulated Modular Multilevel Converters," *IEEE Trans. Power Electron.*, vol. 24, no. 7, pp. 1737–1746, Jul. 2009.
- [21] W. LI, L. GREGOIRE, J. BÉLANGER, "Control and Performance of a Modular Multilevel Converter System," presented at the PARIS CIGRÉ – 110 Conference on Power Systems, Halifax, Canada, 2011.

- [22] S. Yang, A. Bryant, P. Mawby, D. Xiang, L. Ran, and P. Tavner, "An Industry-Based Survey of Reliability in Power Electronic Converters," *IEEE Trans. Ind. Appl.*, vol. 47, no. 3, pp. 1441–1451, May 2011.
- [23] B. Lu and S. K. Sharma, "A Literature Review of IGBT Fault Diagnostic and Protection Methods for Power Inverters," *IEEE Trans. Ind. Appl.*, vol. 45, no. 5, pp. 1770–1777, Sep. 2009.
- [24] "BS EN 62501: VOLTAGE SOURCED CONVERTER (VSC) VALVES FOR HIGH-VOLTAGE DIRECT CURRENT (HVDC) POWER TRANSMISSION."
- [25] K. Tahata, S. E. Oukaili, K. Kamei, D. Yoshida, Y. Kono, R. Yamamoto, and H. Ito, "HVDC circuit breakers for HVDC grid applications," in 11th IET International Conference on AC and DC Power Transmission, 2015, pp. 1–9.
- [26] H. Liu, P. C. Loh, and F. Blaabjerg, "Review of fault diagnosis and faulttolerant control for modular multilevel converter of HVDC," in *IECON 2013 - 39th Annual Conference of the IEEE Industrial Electronics Society*, 2013, pp. 1242–1247.
- [27] J. Rodriguez, L. G. Franquelo, S. Kouro, J. I. Leon, R. C. Portillo, M. A. M. Prats, and M. A. Perez, "Multilevel Converters: An Enabling Technology for High-Power Applications," *Proc. IEEE*, vol. 97, no. 11, pp. 1786–1817, Nov. 2009.
- [28] L. G. Franquelo, J. Rodriguez, J. I. Leon, S. Kouro, R. Portillo, and M. A. M. Prats, "The age of multilevel converters arrives," *IEEE Ind. Electron. Mag.*, vol. 2, no. 2, pp. 28–39, Jun. 2008.
- [29] R. L. Naik and K. R. Y. Udaya, "A novel technique for control of cascaded multilevel inverter for photovoltaic power supplies," in 2005 European Conference on Power Electronics and Applications, 2005, p. 9 pp.-pp.P.9.
- [30] M. Calais, V. G. Agelidis, L. J. Borle, and M. S. Dymond, "A transformerless five level cascaded inverter based single phase photovoltaic system," in *Power Electronics Specialists Conference, 2000. PESC 00. 2000 IEEE 31st Annual*, 2000, vol. 3, pp. 1173–1178 vol.3.
- [31] Amirnaser Yazdani, Reza Iravani, Voltage-Sourced Converters in Power Systems : Modeling, Control, and Applications. Wiley-Interscience, 2010.
- [32] M. Marchesoni and P. Tenca, "Diode-clamped multilevel converters: a practicable way to balance DC-link voltages," *IEEE Trans. Ind. Electron.*, vol. 49, no. 4, pp. 752–765, Aug. 2002.

- [33] S. Ogasawara and H. Akagi, "Analysis of variation of neutral point potential in neutral-point-clamped voltage source PWM inverters," in , Conference Record of the 1993 IEEE Industry Applications Society Annual Meeting, 1993, 1993, pp. 965–970 vol.2.
- [34] C. Newton and M. Sumner, "Neutral point control for multi-level inverters: theory, design and operational limitations," in , *Conference Record of the 1997 IEEE Industry Applications Conference, 1997. Thirty-Second IAS Annual Meeting, IAS '97*, 1997, vol. 2, pp. 1336–1343 vol.2.
- [35] A. Bendre, G. Venkataramanan, D. Rosene, and V. Srinivasan, "Modeling and design of a neutral-point voltage regulator for a three-level diode-clamped inverter using multiple-carrier modulation," *IEEE Trans. Ind. Electron.*, vol. 53, no. 3, pp. 718–726, Jun. 2006.
- [36] C. Newton and M. Sumner, "Neutral point control for multi-level inverters: theory, design and operational limitations," in , *Conference Record of the 1997 IEEE Industry Applications Conference, 1997. Thirty-Second IAS Annual Meeting, IAS '97*, 1997, vol. 2, pp. 1336–1343 vol.2.
- [37] A. K. Gupta and A. M. Khambadkone, "A Simple Space Vector PWM Scheme to Operate a Three-Level NPC Inverter at High Modulation Index Including Overmodulation Region, With Neutral Point Balancing," *IEEE Trans. Ind. Appl.*, vol. 43, no. 3, pp. 751–760, May 2007.
- [38] A. Videt, P. L. Moigne, N. Idir, P. Baudesson, and X. Cimetiere, "A New Carrier-Based PWM Providing Common-Mode-Current Reduction and DC-Bus Balancing for Three-Level Inverters," *IEEE Trans. Ind. Electron.*, vol. 54, no. 6, pp. 3001–3011, Dec. 2007.
- [39] J. I. Leon, G. Guidi, L. G. Franquelo, T. Undeland, and S. Vazquez, "Simple Control Algorithm to Balance the DC-Link Voltage in Multilevel Four-Leg Four-Wire Diode Clamped Converters," in *Power Electronics and Motion Control Conference, 2006. EPE-PEMC 2006. 12th International*, 2006, pp. 228–233.
- [40] S. Busquets-Monge, J. Bordonau, D. Boroyevich, and S. Somavilla, "The nearest three virtual space vector PWM - a modulation for the comprehensive neutral-point balancing in the three-level NPC inverter," *IEEE Power Electron. Lett.*, vol. 2, no. 1, pp. 11–15, Mar. 2004.
- [41] N. Celanovic and D. Boroyevich, "A comprehensive study of neutral-point voltage balancing problem in three-level neutral-point-clamped voltage source PWM inverters," *IEEE Trans. Power Electron.*, vol. 15, no. 2, pp. 242–249, Mar. 2000.

- [42] B. P. McGrath, T. Meynard, G. Gateau, and D. G. Holmes, "Optimal Modulation of Flying Capacitor and Stacked Multicell Converters Using a State Machine Decoder," *IEEE Trans. Power Electron.*, vol. 22, no. 2, pp. 508–516, Mar. 2007.
- [43] B. P. McGrath and D. G. Holmes, "Analytical Modelling of Voltage Balance Dynamics for a Flying Capacitor Multilevel Converter," *IEEE Trans. Power Electron.*, vol. 23, no. 2, pp. 543–550, Mar. 2008.
- [44] C. Turpin, F. Forest, F. Richardeau, T. A. Meynard, and A. Lacarnoy, "Switching faults and safe control of an ARCP multicell flying capacitor inverter," *IEEE Trans. Power Electron.*, vol. 18, no. 5, pp. 1158–1167, Sep. 2003.
- [45] X. Kou, K. A. Corzine, and Y. L. Familiant, "A unique fault-tolerant design for flying capacitor multilevel inverter," *IEEE Trans. Power Electron.*, vol. 19, no. 4, pp. 979–987, Jul. 2004.
- [46] C. C. Davidson and D. R. Trainer, "Innovative concepts for hybrid multilevel converters for HVDC power transmission," in 9th IET International Conference on AC and DC Power Transmission, 2010. ACDC, 2010, pp. 1–5.
- [47] M. M. C. Merlin, T. C. Green, P. D. Mitcheson, D. R. Trainer, R. Critchley, W. Crookes, and F. Hassan, "The Alternate Arm Converter: A New Hybrid Multilevel Converter With DC-Fault Blocking Capability," *IEEE Trans. Power Deliv.*, vol. 29, no. 1, pp. 310–317, Feb. 2014.
- [48] A. M. Cross, D. R. Trainer, and R. W. Crookes, "Chain-link based HVDC Voltage Source Converter using current injection," in 9th IET International Conference on AC and DC Power Transmission, 2010. ACDC, 2010, pp. 1–5.
- [49] G. P. Adam, K. H. Ahmed, S. J. Finney, K. Bell, and B. W. Williams, "New Breed of Network Fault-Tolerant Voltage-Source-Converter HVDC Transmission System," *IEEE Trans. Power Syst.*, vol. 28, no. 1, pp. 335–346, Feb. 2013.
- [50] G. P. Adam, I. A. Abdelsalam, K. H. Ahmed, and B. W. Williams, "Hybrid Multilevel Converter With Cascaded H-bridge Cells for HVDC Applications: Operating Principle and Scalability," *IEEE Trans. Power Electron.*, vol. 30, no. 1, pp. 65–77, Jan. 2015.
- [51] Y. Zhang, G. P. Adam, T. C. Lim, S. J. Finney, and B. W. Williams, "Hybrid Multilevel Converter: Capacitor Voltage Balancing Limits and its Extension," *IEEE Trans. Ind. Inform.*, vol. 9, no. 4, pp. 2063–2073, Nov. 2013.

- [52] J. Qin and M. Saeedifard, "Predictive Control of a Modular Multilevel Converter for a Back-to-Back HVDC System," *IEEE Trans. Power Deliv.*, vol. 27, no. 3, pp. 1538–1547, Jul. 2012.
- [53] D. Jovcic and A. A. Jamshidifar, "Phasor Model of Modular Multilevel Converter With Circulating Current Suppression Control," *IEEE Trans. Power Deliv.*, vol. 30, no. 4, pp. 1889–1897, Aug. 2015.
- [54] A. Jamshidifar and D. Jovcic, "Small-Signal Dynamic DQ Model of Modular Multilevel Converter for System Studies," *IEEE Trans. Power Deliv.*, vol. 31, no. 1, pp. 191–199, Feb. 2016.
- [55] J. Xu, A. M. Gole, and C. Zhao, "The Use of Averaged-Value Model of Modular Multilevel Converter in DC Grid," *IEEE Trans. Power Deliv.*, vol. 30, no. 2, pp. 519–528, Apr. 2015.
- [56] J. Xu, C. Zhao, W. Liu, and C. Guo, "Accelerated Model of Modular Multilevel Converters in PSCAD/EMTDC," *IEEE Trans. Power Deliv.*, vol. 28, no. 1, pp. 129–136, Jan. 2013.
- [57] U. N. Gnanarathna, A. M. Gole, and R. P. Jayasinghe, "Efficient Modeling of Modular Multilevel HVDC Converters (MMC) on Electromagnetic Transient Simulation Programs," *IEEE Trans. Power Deliv.*, vol. 26, no. 1, pp. 316–324, Jan. 2011.
- [58] A. Beddard, M. Barnes, and R. Preece, "Comparison of Detailed Modeling Techniques for MMC Employed on VSC-HVDC Schemes," *IEEE Trans. Power Deliv.*, vol. 30, no. 2, pp. 579–589, Apr. 2015.
- [59] A. Ferreira, R. Teixeira-Pinto, O. Gomis-Bellmunt, and M. Teixidó, "Elimination of MMC differential currents via a feedback LTI control system," in 2015 9th International Conference on Power Electronics and ECCE Asia (ICPE-ECCE Asia), 2015, pp. 1535–1541.
- [60] W. Jiang, Q. Wang, Q. Chen, and C. Hu, "A simplified SVPWM control algorithm for multilevel VSI based two-level SVPWM control algorithm," in *International Conference on Electrical Machines and Systems, 2007. ICEMS*, 2007, pp. 43–49.
- [61] R. Yang, G. Wang, Y. Yu, D. Xu, and Z. Xu, "A fast generalized multilevel voltage SVPWM algorithm based on voltage decomposition," in *2010 5th IEEE Conference on Industrial Electronics and Applications*, 2010, pp. 1220–1224.
- [62] A. Hassanpoor, S. Norrga, H. P. Nee, and L. Ängquist, "Evaluation of different carrier-based PWM methods for modular multilevel converters for

HVDC application," in *IECON 2012 - 38th Annual Conference on IEEE Industrial Electronics Society*, 2012, pp. 388–393.

- [63] D. W. Kang, W. K. Lee, and D. S. Hyun, "Carrier-rotation strategy for voltage balancing in flying capacitor multilevel inverter," *IEE Proc. - Electr. Power Appl.*, vol. 151, no. 2, pp. 239–248, Mar. 2004.
- [64] M. M. C. Merlin and T. C. Green, "Cell capacitor sizing in multilevel converters: cases of the modular multilevel converter and alternate arm converter," *IET Power Electron.*, vol. 8, no. 3, pp. 350–360, 2015.
- [65] C. Oates, "Modular Multilevel Converter Design for VSC HVDC Applications," *IEEE J. Emerg. Sel. Top. Power Electron.*, vol. 3, no. 2, pp. 505– 515, Jun. 2015.
- [66] Q. Tu, Z. Xu, H. Huang, and J. Zhang, "Parameter design principle of the arm inductor in modular multilevel converter based HVDC," in 2010 International Conference on Power System Technology (POWERCON), 2010, pp. 1–6.
- [67] A. Lesnicar and R. Marquardt, "An innovative modular multilevel converter topology suitable for a wide power range," in *Power Tech Conference Proceedings*, 2003 IEEE Bologna, 2003, vol. 3, p. 6 pp. Vol.3-pp.
- [68] A. Antonopoulos, L. Angquist, and H. P. Nee, "On dynamics and voltage control of the Modular Multilevel Converter," in *13th European Conference on Power Electronics and Applications, 2009. EPE '09*, 2009, pp. 1–10.
- [69] K. Ilves, A. Antonopoulos, S. Norrga, L. Angquist, and H. P. Nee, "Controlling the ac-side voltage waveform in a modular multilevel converter with low energy-storage capability," in *Proceedings of the 2011-14th European Conference on Power Electronics and Applications (EPE 2011)*, 2011, pp. 1–8.
- [70] Q. Tu, Z. Xu, and L. Xu, "Reduced Switching-Frequency Modulation and Circulating Current Suppression for Modular Multilevel Converters," *IEEE Trans. Power Deliv.*, vol. 26, no. 3, pp. 2009–2017, Jul. 2011.
- [71] R. Darus, J. Pou, G. Konstantinou, S. Ceballos, R. Picas, and V. G. Agelidis,
 "A Modified Voltage Balancing Algorithm for the Modular Multilevel Converter: Evaluation for Staircase and Phase-Disposition PWM," *IEEE Trans. Power Electron.*, vol. 30, no. 8, pp. 4119–4127, Aug. 2015.
- [72] R. Chokhawala, J. Catt, and L. Kiraly, "A discussion on IGBT short circuit behavior and fault protection schemes," in *Applied Power Electronics Conference and Exposition, 1993. APEC '93. Conference Proceedings 1993., Eighth Annual*, 1993, pp. 393–401.

- [73] R. Wu, F. Blaabjerg, H. Wang, M. Liserre, and F. Iannuzzo, "Catastrophic failure and fault-tolerant design of IGBT power electronic converters - an overview," in *IECON 2013 - 39th Annual Conference of the IEEE Industrial Electronics Society*, 2013, pp. 507–513.
- [74] Z. Zhang and G.-Q. Lu, "Pressure-assisted low-temperature sintering of silver paste as an alternative die-attach solution to solder reflow," *IEEE Trans. Electron. Packag. Manuf.*, vol. 25, no. 4, pp. 279–283, Oct. 2002.
- [75] H. Wang and F. Blaabjerg, "Reliability of capacitors for DC-link applications #x2014; An overview," in *2013 IEEE Energy Conversion Congress and Exposition*, 2013, pp. 1866–1873.
- [76] Y. Tang, L. Ran, O. Alatise, and P. Mawby, "Capacitor selection for modular multilevel converter," in 2014 IEEE Energy Conversion Congress and Exposition (ECCE), 2014, pp. 2080–2087.
- [77] D. G. Shaw, S. W. Cichanowski, and A. Yializis, "A Changing Capacitor Technology - Failure Mechanisms and Design Innovations," *IEEE Trans. Electr. Insul.*, vol. EI-16, no. 5, pp. 399–413, Oct. 1981.
- [78] S. Shao, P. W. Wheeler, J. C. Clare, and A. J. Watson, "Fault Detection for Modular Multilevel Converters Based on Sliding Mode Observer," *IEEE Trans. Power Electron.*, vol. 28, no. 11, pp. 4867–4872, Nov. 2013.
- [79] "IEEE Draft Standard for Common Definitions for Switchgear," *IEEE PC371005D2 Febr. 2016*, pp. 1–37, Jan. 2016.
- [80] H. Hammouri, M. Kinnaert, and E. H. E. Yaagoubi, "Observer-based approach to fault detection and isolation for nonlinear systems," *IEEE Trans. Autom. Control*, vol. 44, no. 10, pp. 1879–1884, Oct. 1999.
- [81] E. Chow and A. Willsky, "Analytical redundancy and the design of robust failure detection systems," *IEEE Trans. Autom. Control*, vol. 29, no. 7, pp. 603– 614, Jul. 1984.
- [82] S. Sedghi, A. Dastfan, and A. Ahmadyfard, "Fault detection of a seven level modular multilevel inverter via voltage histogram and Neural Network," in 2011 IEEE 8th International Conference on Power Electronics and ECCE Asia (ICPE ECCE), 2011, pp. 1005–1012.
- [83] R. Patton and P. M. Frank, *Fault diagnosis in dynamic systems: theory and applications*. Prentice Hall, 1989.
- [84] Z. Wang, X. Shi, L. M. Tolbert, F. (Wang, and B. J. Blalock, "A di/dt Feedback-Based Active Gate Driver for Smart Switching and Fast Overcurrent

Protection of IGBT Modules," *IEEE Trans. Power Electron.*, vol. 29, no. 7, pp. 3720–3732, Jul. 2014.

- [85] M. A. Rodriguez-Blanco, A. Claudio-Sanchez, D. Theilliol, L. G. Vela-Valdes, P. Sibaja-Teran, L. Hernandez-Gonzalez, and J. Aguayo-Alquicira, "A Failure-Detection Strategy for IGBT Based on Gate-Voltage Behavior Applied to a Motor Drive System," *IEEE Trans. Ind. Electron.*, vol. 58, no. 5, pp. 1625– 1633, May 2011.
- [86] S. Haghnazari, M. Khodabandeh, and M. R. Zolghadri, "Fast fault detection method for modular multilevel converter semiconductor power switches," *IET Power Electron.*, vol. 9, no. 2, pp. 165–174, 2016.
- [87] M. Blanke, R. Izadi-Zamanabadi, S. A. Bøgh, and C. P. Lunau, "Faulttolerant control systems — A holistic view," *Control Eng. Pract.*, vol. 5, no. 5, pp. 693–702, May 1997.
- [88] M. Blanke, M. Staroswiecki, and N. E. Wu, "Concepts and methods in faulttolerant control," in *American Control Conference*, 2001. Proceedings of the 2001, 2001, vol. 4, pp. 2606–2620 vol.4.
- [89] "Introduction to diagnosis and fault-tolerant control," in *Diagnosis and Fault-Tolerant Control*, Springer Berlin Heidelberg, 2006, pp. 1–32.
- [90] K. Tsuda, D. Mignone, G. Ferrari-Trecate, and M. Morari, "Reconfiguration strategies for hybrid systems," in *American Control Conference*, 2001. *Proceedings of the 2001*, 2001, vol. 2, pp. 868–873 vol.2.
- [91] Youmin Zhang, Jin Jiang, "Bibliographical review on reconfigurable faulttolerant control systems," *ELSEVIER Annu. Rev. Control*, vol. 32, no. 2, pp. 229–252, Dec. 2008.
- [92] G.-H. Yang, S.-Y. Zhang, J. Lam, and J. Wang, "Reliable control using redundant controllers," *IEEE Trans. Autom. Control*, vol. 43, no. 11, pp. 1588– 1593, Nov. 1998.
- [93] R. J. Veillette, "Reliable linear-quadratic state-feedback control," *Automatica*, vol. 31, no. 1, pp. 137–143, Jan. 1995.
- [94] G.-H. Yang, J. L. Wang, Y. C. Soh, and K.-Y. Lou, "Reliable State Feedback Control Synthesis For Uncertain Linear Systems," *Asian J. Control*, vol. 5, no. 2, pp. 301–308, Jun. 2003.
- [95] W. Zhang, D. Xu, P. N. Enjeti, H. Li, J. T. Hawke, and H. S. Krishnamoorthy, "Survey on Fault-Tolerant Techniques for Power Electronic Converters," *IEEE Trans. Power Electron.*, vol. 29, no. 12, pp. 6319–6331, Dec. 2014.

- [96] G. T. Son, H. J. Lee, T. S. Nam, Y. H. Chung, U. H. Lee, S. T. Baek, K. Hur, and J. W. Park, "Design and Control of a Modular Multilevel HVDC Converter With Redundant Power Modules for Noninterruptible Energy Transfer," *IEEE Trans. Power Deliv.*, vol. 27, no. 3, pp. 1611–1619, Jul. 2012.
- [97] A. Benchaib, M. Tadjine, A. Rachid, and A. Tayebi, "Adaptive sliding mode observer for state estimation of an induction motor on DSP-based system," in *Electric Machines and Drives Conference Record, 1997. IEEE International*, 1997, p. MC2/3.1-MC2/3.3.
- [98] F. Chen and M. W. Dunnigan, "Comparative study of a sliding-mode observer and Kalman filters for full state estimation in an induction machine," *IEE Proc. - Electr. Power Appl.*, vol. 149, no. 1, pp. 53–64, Jan. 2002.
- [99] S. R. K. Hoseini, E. Farjah, T. Ghanbari, and H. Givi, "Extended Kalman filter-based method for inter-turn fault detection of the switched reluctance motors," *IET Electr. Power Appl.*, vol. 10, no. 8, pp. 714–722, 2016.
- [100] M. I. Marei, E. F. El-Saadany, and M. M. A. Salama, "Experimental evaluation of envelope tracking techniques for voltage disturbances," *Electr. Power Syst. Res.*, vol. 80, no. 3, pp. 339–344, Mar. 2010.
- [101] Mostafa I. Marei, Nour Elsayad, Ahmed A. El-Sattar, "PV interface system with LVRT capability based on a current controlled HFAC link converter," *Elsevier Sustain. Energy Technol. Assess.*, vol. 9, pp. 55–62, Mar. 2015.
- [102] B. Widrow and M. A. Lehr, "30 years of adaptive neural networks: perceptron, Madaline, and backpropagation," *Proc. IEEE*, vol. 78, no. 9, pp. 1415–1442, Sep. 1990.
- [103] S. Hui and S. H. Zak, "Robust stability analysis of adaptation algorithms for single perceptron," *IEEE Trans. Neural Netw.*, vol. 2, no. 2, pp. 325–328, Mar. 1991.
- [104] Maamar Bettayeb, Uvais Qidwai, "Recursive estimation of power system harmonics," *Elsevier Electr. Power Syst. Res.*, vol. 47, no. 2, pp. 143–152, Oct. 1998.
- [105] "System Modeling and Identification," in *Hard Disk Drive Servo Systems*, Springer London, 2006, pp. 21–35.
- [106] X. She and A. Huang, "Circulating current control of double-star choppercell modular multilevel converter for HVDC system," in *IECON 2012 - 38th Annual Conference on IEEE Industrial Electronics Society*, 2012, pp. 1234– 1239.

- [107] M. Saeedifard and R. Iravani, "Dynamic Performance of a Modular Multilevel Back-to-Back HVDC System," *IEEE Trans. Power Deliv.*, vol. 25, no. 4, pp. 2903–2912, Oct. 2010.
- [108] "PSCAD Home | PSCAD." [Online]. Available: https://hvdc.ca/pscad/. [Accessed: 28-Jun-2016].
- [109] N. RASHIDI-RAD, A. RAHMATI, A. ABRISHAMIFAR, "Comparison of Reliability in Modular Multilevel Inverters," PRZEGLĄD ELEKTROTECHNICZNY Electr. Rev.
- [110] "Datasheet of SCD IGBT Snubber Capacitor Modules.".
- [111] S. Dennetière, H. Saad, B. Clerc, E. Ghahremani, W. Li, and J. Bélanger, "Validation of a MMC model in a real-time simulation platform for industrial HIL tests," in 2015 IEEE Power Energy Society General Meeting, 2015, pp. 1– 5.
- [112] H. K. Fathy, Z. S. Filipi, J. Hagena, and J. L. Stein, "Review of hardware-inthe-loop simulation and its prospects in the automotive area," 2006, p. 62280E– 62280E–20.
- [113] C. Lin and L. Zhang, "Hardware-in-the-loop simulation and its application in electric vehicle development," in 2008 IEEE Vehicle Power and Propulsion Conference, 2008, pp. 1–6.
- [114] X. Guillaud, M. O. Faruque, A. Teninge, A. H. Hariri, L. Vanfretti, M. Paolone, V. Dinavahi, P. Mitra, G. Lauss, C. Dufour, P. Forsyth, A. K. Srivastava, K. Strunz, T. Strasser, and A. Davoudi, "Applications of Real-Time Simulation Technologies in Power and Energy Systems," *IEEE Power Energy Technol. Syst. J.*, vol. 2, no. 3, pp. 103–115, Sep. 2015.
- [115] "OP4510_Data_Sheet_May_14_2015.pdf.".
- [116] "LabVIEW System Design Software National Instruments." [Online]. Available: http://www.ni.com/labview/. [Accessed: 28-Jun-2016].
- [117] "cRIO-9024 National Instruments." [Online]. Available: http://sine.ni.com/nips/cds/view/p/lang/en/nid/207371. [Accessed: 28-Jun-2016].
- [118] "Space-grade Virtex-5QV FPGA." [Online]. Available: http://www.xilinx.com/products/silicon-devices/fpga/virtex-5qv.html. [Accessed: 28-Jun-2016].
- [119] "NI CompactRIO Developers Guide National Instruments." [Online]. Available: https://www.ni.com/compactriodevguide/. [Accessed: 28-Jun-2016].

- [120] "FPGA Fundamentals National Instruments." [Online]. Available: http://www.ni.com/white-paper/6983/en/. [Accessed: 01-Jul-2016].
- [121] "Simulink Simulation and Model-Based Design MathWorks United Kingdom." [Online]. Available: http://uk.mathworks.com/products/simulink/. [Accessed: 28-Jun-2016].

Appendix A

Modelling of Wind Energy Conversion Systems

This appendix describes the modelling process of wind energy conversion systems used in the simulation model which is presented in sub-section 4.3.2.

A.1 Modelling of the Wind Turbine

The wind turbine model is built according to the steady state characteristics of it. The power output is expressed as:

$$P_m = 0.5\rho A C_p v_w^3 \tag{A.1}$$

Where ρ is the air density, A is the effective area of the blade, C_p is the power coefficient of the turbine and v_w is the wind speed. The C_p needs to be modelled accurately as it is considered to be the governing factor in (1) since it determines the efficiency of the power converted from wind to mechanical [9]. The C_p is calculated from:

$$C_p = 0.22(116\lambda_1 - 0.4\beta - 5)e^{-12.5\lambda_1}$$
(A.2)

$$\lambda_1 = \frac{1}{\lambda + 0.08} - \frac{0.035}{1 + \beta^3} \tag{A.3}$$

where β is the pitch angle and λ is the tip speed ratio. The output mechanical torque is given by:

$$T_m = \frac{P_m}{\omega_m} \tag{A.4}$$

where ω_m is the rotor angular speed.

A.2 Modelling of PMSG

A third order model is used to represent the PMSG on PSCAD/EMTDC software package. The mathematical equations of the PMSG model in the rotor frame are:

$$\frac{d}{dt} \begin{bmatrix} i_{ds}^{e} \\ i_{qs}^{e} \end{bmatrix} = \begin{bmatrix} -\frac{R_s}{L_d} & \frac{L_q P \omega_m}{L_d} \\ -\frac{L_d P \omega_m}{L_q} & -\frac{R_s}{L_q} \end{bmatrix} \begin{bmatrix} i_{ds}^{e} \\ i_{qs}^{e} \end{bmatrix} - \begin{bmatrix} \frac{v_{ds}^{e}}{L_d} \\ \frac{v_{qs}^{e} - P \omega_m \psi_{pm}}{L_q} \end{bmatrix}$$
(A.5)

Where:

 i_{ds}^{e}, i_{qs}^{e} are the stator current components.

 v_{ds}^e , v_{qs}^e are the stator voltage components.

 R_s is the stator resistance while L_d and L_q are the direct and quadrature components of the inductance.

P is the number of pole pairs and ψ_{pm} is the magnetic flux inside the PMSG.

The electromagnetic torque is given by:

$$T_e = \frac{3}{2} P(i_{qs}^e \psi_{pm} + (L_d - L_q) i_{qs}^e i_{ds}^e)$$
(A.6)

From (A1.4) and (A1.6) the electromechanical equation is written as:

$$T_m - T_e = J \frac{d\omega_m}{dt} + B\omega_m \tag{A.7}$$

A.3 Modelling of the DFIG

Induction machine dynamic operation modes have been described by means of a system of voltage differential equations for the stator and rotor coils respectively. The DFIG mathematical model expressed in unit values and $\alpha\beta$ coordinate system are as follows:

$$\frac{d\psi_{s\alpha}}{dt} = -\frac{1}{T_s'}\psi_{s\alpha} + \frac{k_r}{T_s'}\psi_{r\alpha} + u_{s\alpha}, \qquad (A.8)$$

$$\frac{d\psi_{s\beta}}{dt} = -\frac{1}{T_s}\psi_{s\beta} + \frac{k_r}{T_s}\psi_{r\beta} + u_{s\beta}, \qquad (A.9)$$

$$\frac{d\psi_{r\alpha}}{dt} = \frac{k_s}{T_r} \psi_{s\alpha} - \frac{1}{T_r} \psi_{r\alpha} - \omega \psi_{r\beta} + u_{r\alpha}$$
(A.10)

$$\frac{d\psi_{r\beta}}{dt} = \frac{k_s}{T_r'}\psi_{s\beta} + \omega\psi_{r\alpha} - \frac{1}{T_r'}\psi_{r\beta} + u_{r\beta}, \qquad (A.11)$$

Rotor and stator feed voltage vector components are:

$$u_{s\alpha} = u_{s\alpha}, \qquad (A.12)$$

$$u_{s\beta} = \frac{1}{\sqrt{3}} (u_{sb} - u_{sc}), \tag{A.13}$$

$$u_{r\alpha} = u_{ra}, \qquad (A.14)$$

$$u_{r\beta} = \frac{1}{\sqrt{3}} (u_{rb} - u_{rc}).$$
 (A.15)

The stator and rotor current vector components, expressed by means of the known magnetic flux components, are as follows:

$$i_{s\alpha} = \frac{1}{L_s} \psi_{s\alpha} - \frac{k_s}{L_r} \psi_{r\alpha}, \qquad (A.16)$$

$$i_{s\beta} = \frac{1}{L_s} \psi_{s\beta} - \frac{k_s}{L_r} \psi_{r\beta}, \qquad (A.17)$$

$$i_{r\alpha} = -\frac{k_r}{L_s'} \psi_{s\alpha} + \frac{1}{L_r'} \psi_{r\alpha}, \qquad (A.18)$$

$$i_{r\beta} = -\frac{k_r}{L_s} \psi_{s\beta} + \frac{1}{L_r} \psi_{r\beta} \,. \tag{A.19}$$

The generator electromagnetic moment, at the same time representing the input value of the wind turbine two-mass model, is:

$$m_e = \psi_{s\alpha} i_{s\beta} - \psi_{s\beta} i_{s\alpha} \,. \tag{A.20}$$

The following are the parametres as occurring in the equations from (A.8) to (A.19):

$$L'_{s} = \sigma L_{s}, L'_{r} = \sigma L_{r}, \sigma = 1 - \frac{L^{2}_{m}}{L_{s}L_{r}}, k_{s} = \frac{L_{m}}{L_{s}}, k_{r} = \frac{L_{m}}{L_{r}}, T'_{s} = \frac{L'_{s}}{R_{s}} \text{ and } T'_{r} = \frac{L'_{r}}{R_{r}}.$$

The induction generator stator active and reactive power momentary value is obtained by multiplying the stator voltage vector by conjugated-complex value of the stator current vector, as illustrated below:

$$p_{sa} = u_{s\alpha} i_{s\alpha} + u_{s\beta} i_{s\beta} , \qquad (A.21)$$

$$p_{sr} = u_{s\beta} i_{s\alpha} - u_{s\alpha} i_{s\beta} \,. \tag{A.22}$$

The DFIG rotor active power momentary value may be calculated in a similar way:

$$p_{ra} = u_{r\alpha} i_{r\alpha} + u_{r\beta} i_{r\beta} \,. \tag{A.23}$$

The momentary value of the rotor reactive power equals zero.

The momentary values of the induction generator stator and rotor current may be obtained from the $\alpha\beta$ coordinate system components by means of the following equations:

$$i_s = \sqrt{i_{s\alpha}^2 + i_{s\beta}^2}, \qquad (A.24)$$

$$i_r = \sqrt{i_{r\alpha}^2 + i_{r\beta}^2}$$
 (A.25)

Losses in the generator coils stator and rotor DFIG are:

$$p_{Cu} = R_s i_s^2 + R_r i_r^2 \,. \tag{A.26}$$

A.4 Control of Active and Reactive Power Injected into the Grid

Controlling the flow of active and reactive powers in the grid is done by controlling both the DC and AC voltage at each terminal. The control system at each terminal is formed from five units as described in the following sub-sections.

A.4.1 Phase locked loop

The phase locked loop technique is responsible for synchronizing the AC voltage of the MMC with the grid voltage. This is done using a feed forward PI control loop which receives the quadrature axis voltage of the MMC AC terminals V_{tq} . The output of the PI controller is then added to the initial angular frequency ω_0 . The output of the addition is passed through integrator resulting into the grid phase θ .

A.4.2 AC and DC voltage control loops

The active and reactive power control loops are very simple as they utilize two PI controllers to decide the values of the MMC direct and quadrature components of the desired MMC output current i_d^* and i_q^* .

A.4.3 Inner current loop

The inner current control loop is used to regulate the MMC output current based on the received references from the DC and AC control loops, the loop is designed based on the MMC dynamics of the AC side:

$$L_{grid} \frac{di_d}{dt} = i_q \omega L_{grid} - i_d R_{grid} + V_{td} - V_{sd}$$
(A.27)

$$L_{grid}\frac{di_q}{dt} = -i_d \omega L_{grid} - i_q R_{grid} + V_{tq} - V_{sq}$$
(A.28)

Where L_{grid} and R_{grid} are the grid inductance and resistance. V_t is the MMC terminal voltage while V_s is the grid voltage.

According to (A.27) and (A.28), two PI controllers are added to regulate the i_d and i_q . The output of the two controllers is then added to the measured components of the grid voltage V_{sd} and V_{sq} forming two feed forward loops. The decoupled components of i_d and i_q are then subtracted from the two loops resulting in the reference terminal voltages V_{td}^* and V_{tq}^* . Figure A.1 shows the block diagram of the active and reactive power control.



Figure A.1: Active and reactive power control.

Appendix B

Details of PSCAD Simulation Models

This appendix shows all details of the MMC simulation models, this include the following:

- The MMC submodule model
- The block of PS-PWM.
- The ADALINE programming function
- The RLS programming function
- The block of the proposed capacitor voltage balancing algorithm.
- The block of the proposed FDU.
- The block of the proposed FTCU.







Figure B.2: The submodule model.



Figure B.3: The RLS programming block.

The following code is the RLS capacitor voltage balancing algorithm which is used in the RLS block (shown in Figure B.3):

```
! try
12345678
        #LOCAL REAL auphi 4
        #LOCAL REAL auphip 4
        #LOCAL REAL aupphi 4
        #LOCAL REAL aup1 4
        #LOCAL REAL aup2 4
        #LOCAL REAL aup3 4
        #LOCAL REAL aup4 4
        #LOCAL REAL auth 4
       IF (TIME .EQ. 0) THEN
         cunter=1
          i=0
          DO AUK= 1, 4
           i=i+1
           auth(i) = 0
          aupl(i) = 0
           aup2(i) = 0
           aup3(i) = 0
          aup4(i) = 0
          END DO
          aup1(1) = 0.5
           aup2(2) = 0.5
           aup3(3) = 0.5
           aup4(4) = 0.5
       ENDIF
       auphi(1) = $s1
auphi(2) = $s2
       auphi(3) = \$s3
       auphi(4) = $s4
!IF (cunter .EQ. 1) THEN
                     ----p=(p-(p*phi*phi'*p/(la+phi'*p*phi)))/la;------
! ---
          i=0
                          ! uphip = phi'*p
          DO AUK= 1, 4
          i=i+1
auphip(i) = (auphi(1) * aup1(i)) + (auphi(2) * aup2(i)) + (auphi(3) * aup3(i)) + (auphi(4) * aup4(i
))
         END DO
          i=0
                          ! under = la+uphip*phi
         auunder=$landa
          DO AUK= 1, 4
          i=i+1
```

```
auunder= auunder + (auphip(i)*auphi(i))
        END DO
         IF (auunder .GT. 0) THEN
!******
             ******
        * * * *
        i=0
                    ! pphi=zero
        DO AUK= 1, 4
         i=i+1
         aupphi(i)=0
        END DO
         i = 0
                     ! pphi=p*phi
        DO AUK= 1, 4
         i=i+1
         aupphi(1) = aupphi(1) + (aup1(i) * auphi(i))
         aupphi(2) = aupphi(2) + (aup2(i) * auphi(i))
         aupphi(3) = aupphi(3) + (aup3(i) * auphi(i))
         aupphi(4) = aupphi(4) + (aup4(i) * auphi(i))
        END DO
         i=0
                     ! p=(p-(upphi*uphip/under))/la;
         DO AUK= 1, 4
         i=i+1
         aup1(i)=(aup1(i)-(aupphi(1)*auphip(i)/auunder))/$landa
         aup2(i) = (aup2(i) - (aupphi(2) * auphip(i) / auunder)) / $landa
         aup3(i) = (aup3(i) - (aupphi(3) *auphip(i) /auunder)) /$landa
         aup4(i) = (aup4(i) - (aupphi(4) *auphip(i) /auunder)) /$landa
        END DO
   ____
                     _____
                                                _____
   ----- th=th+p*phi*(y-phi'*th) ------
         i=0
                     ! c = y - phi'*th
         cau = ($vd)-($va)-($vlu)
         DO AUK= 1, 4
         i=i+1
         cau = cau - (auphi(i)*auth(i))
        END DO
         i=0
                     ! th = th + pphi*c
        DO AUK= 1, 4
         i=i+1
         auth(i) = auth(i) + (aupphi(i)*cau)
        END DO
                                 _____
!-----
        ENDIF
cunter=0
!
       ENDIF
      cunter=cunter+1
      vc1e = auth(1)
      vc2e = auth(2)
      $vc3e = auth(3)
$vc4e = auth(4)
                                     A2
                                           A3
                                                 A4
                               A1
                                             s3
                                s1
                                      s2
                                                   s4
                                                     vc1e
                          vlu
                   *
                                     ADALINE
                                                               Vcea1
            vlau
                 1000
                                                     vc2e
                   *
                          va
                                   for Upper Vcap
                 1000
                                                               /cea2
                                                     vc3e
                          vd
                                     of Phase a
                 1000
            vď
                                                               Vcea3
                                                      c4e
                                                               Vcea4
```

Figure B.4: The ADALINE programming block.

The following code is the ADALINE capacitor voltage balancing algorithm which is used in the ADALINE block (shown in Figure B.4):

```
! try
!2345678
       IF (TIME .EQ. 0) THEN
         cunter=1
         w1 = 0
         w^2 = 0
w^3 = 0
         w4 = 0
       ENDIF
!IF (cunter .EQ. 1) THEN
         yc = w1*(\$s1) + w2*(\$s2) + w3*(\$s3) + w4*(\$s4)
          E = (\$s1)*(\$s1) + (\$s2)*(\$s2) + (\$s3)*(\$s3) + (\$s4)*(\$s4)
          y = (\$vd) - (\$va) - (\$vlu)
          w1 = w1 + c*(\$s1)
           w^2 = w^2 + c^* (\$s^2)
           w3 = w3 + c* ($s3)
w4 = w4 + c* ($s4)
          ENDIF
          cunter=0
!ENDIF
       cunter=cunter+1
       $vcle = w1
       vc2e = w2
       $vc3e = w3
       vc4e = w4
```



(a)



(b)



Figure B.5: The block of the proposed capacitor voltage algorithm: a) The averaging controller, b) The balancing controller, c) The PS-PWM modulation technique.


Figure B.6: The block of the proposed FDU.

The following code performs the necessary comparisons and decisions needed for the FDU block (shown in Figure B.6):

```
IF (($cs1 .GT. 200) .and. ($vsig1 .LT. 96)) THEN
fsig1 = 1
sreset1 = 0
ELSEIF ($vsig1 .GT. 96) THEN
fsig1 = 0
sreset1 = 1
ENDIF
IF (($cs2 .GT. 200) .and. ($vsig2 .LT. 96)) THEN
fsig2 = 1
sreset2 = 0
ELSEIF ($vsig2 .GT. 96) THEN
fsig2 = 0
$reset2 = 1
ENDIF
IF (($cs3 .GT. 200) .and. ($vsig3 .LT. 96)) THEN
fsig3 = 1
$reset3 = 0
ELSEIF ($vsig1 .GT. 96) THEN
sig3 = 0
$reset3 = 1
ENDIF
IF (($cs4 .GT. 200) .and. ($vsig4 .LT. 96)) THEN
fsig4 = 1
$reset4 = 0
ELSEIF ($vsig4 .GT. 96) THEN
\$fsig4 = 0
$reset4 = 1
ENDIF
IF (($cs5 .GT. 200) .and. ($vsig5 .LT. 96)) THEN
fsig5 = 1
$reset5 = 0
ELSEIF ($vsig5 .GT. 96) THEN
fsig5 = 0
$reset5 = 1
ENDIF
IF (($cs6 .GT. 200) .and. ($vsig6 .LT. 96)) THEN
fsig6 = 1
$reset6 = 0
ELSEIF ($vsig1 .GT. 96) THEN
fsig6 = 0
$reset6 = 1
ENDIF
IF (($cs7 .GT. 200) .and. ($vsig7 .LT. 96)) THEN
fsig7 = 1
$reset7 = 0
ELSEIF ($vsig7 .GT. 96) THEN
```



Figure B.7: The block of the proposed FDU.

The following code is the fault tolerant control sorting algorithm which is installed inside the FTCU (shown in Figure B.7):

```
IF($fcas .EQ. 0) THEN
$ps1 = 0
$ps2 = 90
$ps3 = 180
ps4 = 270
ps5 = 45
$ps6 = 135
ps7 = 225
$ps8 = 315
f1 = 0
f2 = 0
\$f3 = 0
\$f4 = 0
\$f5 = 0
f6 = 0
\$f7 = 0
$f8 = 0
$vcreft = 2250
snosm = 8
ELSEIF($fcas .EQ. 18) THEN
ps1 = 0
$ps2 = 120
ps3 = 240
ps4 = 270
$ps5 = 60
$ps6 = 180
$ps7 = 225
$ps8 = 300
\$f1 = 0
f2 = 0
```

```
$f3 = 0
$f4 = 1
$f5 = 0
f6 = 0
f7 = 1
$f8 = 0
$vcreft = 3000
snosm = 6
ELSEIF($fcas .EQ. 20) THEN
$ps1 = 0
$ps2 = 120
$ps2 = 240
$ps4 = 270
$ps5 = 60
$ps6 = 180
ps7 = 180
$ps8 = 300
\$f1 = 0
\$f2 = 0
\$f3 = 0

\$f4 = 1

\$f5 = 0
ff6 = 1
ff7 = 0
f8 = 0
$vcreft = 3000
$nosm = 6
ELSEIF($fcas .EQ. 24) THEN
$ps1 = 0
ps2 = 120
$ps3 = 240
$ps4 = 270
$p$4 = 270
$p$5 = 60
$p$6 = 60
$p$7 = 180
$p$8 = 300
$f$1
$f1 = 0
\$f2 = 0
f3 = 0
f4 = 1
f5 = 1
f6 = 0
$f7 = 0
$f8 = 0
$vcreft = 3000
$nosm = 6
ELSEIF($fcas .EQ. 33) THEN
$ps1 = 0
$ps2 = 120
$ps3 = 240
ps4 = 240
$ps5 = 60
$ps6 = 180
$ps7 = 300
$ps8 = 300
f1 = 0
f2 = 0
f3 = 1
\$f4 = 0
\$f5 = 0
$f6 = 1
\$f7 = 0
$f8 = 0
$vcreft = 3000
snosm = 6
ELSEIF($fcas .EQ. 36) THEN
$ps1 = 0
$ps2 = 120
$ps2 = 240
$ps4 = 240
$ps5 = 60
$ps6 = 180
ps7 = 180
$ps8 = 300
f1 = 0
f2 = 0
$f3 = 1
```

```
\$f4 = 0
$f5 = 0
f6 = 1
f7 = 0
$f8 = 0
$vcreft = 3000
snosm = 6
ELSEIF($fcas .EQ. 40) THEN
$ps1 = 0
$ps2 = 120
$ps3 = 240
$ps3 = 240
$ps4 = 240
$ps5 = 60
$ps6 = 60
$ps7 = 180
$ps8 = 300
f1 = 0
f2 = 0
$f3 = 1
\$f4 = 0

\$f5 = 1

\$f6 = 0
\$f7 = 0
\$f8 = 0
$vcreft = 3000
snosm = 6
ELSEIF($fcas .EQ. 65) THEN
ps1 = 0
$ps2 = 120
$ps3 = 120
ps4 = 240
ps5 = 60
$ps6 = 180
$ps7 = 300
$ps8 = 300
f1 = 0
f2 = 1
\$f3 = 0
\$f4 = 0
f5 = 0
f6 = 0
\$f7 = 0
$f8 = 1
$vcreft = 3000
$nosm = 6
ELSEIF($fcas .EQ. 66) THEN
$ps1 = 0
$ps2 = 120
$ps3 = 120
ps4 = 240
$ps5 = 60
ps6 = 180
$ps7 = 300
$ps8 = 300
$ps8 = .
$f1 = 0
$f2 = 1
$f3 = 0
\$f4 = 0
\$f5 = 0
\$f6 = 0
f7 = 1
\$f8 = 0
$vcreft = 3000
nosm = 6
ELSEIF($fcas .EQ. 72) THEN
$ps1 = 0
$ps2 = 120
$ps3 = 120
$ps3 = 240
$ps5 = 60
$ps6 = 60
$ps7 = 180
$ps8 = 300
\$f1 = 0
$f2 = 1
$f3 = 0
\$f4 = 0
```

\$f5 = 1 f6 = 0f7 = 0f8 = 0\$vcreft = 3000 nosm = 6ELSEIF(\$fcas .EQ. 129) THEN \$ps1 = 0 \$ps2 = 0 \$ps3 = 120 \$ps4 = 240 \$ps5 = 60 \$ps6 = 180 \$ps7 = 300 \$ps8 = 300 f1 = 1f2 = 0f3 = 0\$f4 = 0f5 = 0\$f6 = 0\$f7 = 0\$f8 = 1 \$vcreft = 3000 nosm = 6ELSEIF(\$fcas .EQ. 130) THEN \$ps1 = 0 \$ps2 = 0 \$ps3 = 120 ps4 = 240\$ps5 = 60 \$ps6 = 180 \$ps7 = 300 ps = 300\$f1 = 1 \$f2 = 0 \$f3 = 0f4 = 0f5 = 0f6 = 0f7 = 1f8 = 0\$vcreft = 3000 snosm = 6ELSEIF(\$fcas .EQ. 132) THEN \$ps1 = 0
\$ps2 = 0
\$ps3 = 120 ps4 = 240\$ps5 = 60 \$ps6 = 180 ps7 = 180\$ps8 = 300 f1 = 1f2 = 0\$f3 = 0 \$f4 = 0 $\frac{1}{5} = 0$ f6 = 1f7 = 0\$f8 = 0\$vcreft = 3000 nosm = 6 $\texttt{ELSEIF((\$fcas}\ .\texttt{EQ.}\ 1)$.or. ($\$fcas}\ .\texttt{EQ.}\ 16)$.or. ($\$fcas}\ .\texttt{EQ.}\ 17)) THEN$ ps1 = 0\$ps2 = 120 \$ps3 = 240 \$ps4 = 240 \$ps5 = 60 \$ps6 = 180 \$ps7 = 300 \$ps8 = 300 f1 = 0f2 = 0\$f3 = 1 \$f4 = 0\$f5 = 0

```
$f6 = 0
f7 = 0
$f8 = 1
$vcreft = 3000
snosm = 6
\texttt{ELSEIF((\$fcas}\ .\texttt{EQ.}\ 2) .or. (\$fcas}\ .\texttt{EQ.}\ 32) .or. (\$fcas}\ .\texttt{EQ.}\ 34)) THEN
ps1 = 0
$ps2 = 120
$ps3 = 240
$ps4 = 240
$ps5 = 60
$ps6 = 180
$ps7 = 180
$ps8 = 300
$f1 = 0
f2 = 0
f3 = 1
\$f4 = 0
$f5 = 0
f6 = 0
f7 = 1
f8 = 0
$vcreft = 3000
$nosm = 6
ELSEIF(($fcas .EQ. 4) .or. ($fcas .EQ. 64) .or. ($fcas .EQ. 68)) THEN
$ps1 = 0
$ps2 = 120
$ps3 = 120
ps4 = 240
ps5 = 60
$ps6 = 180
$ps7 = 180
$ps8 = 300
f1 = 0
f2 = 1
$f3 = 0
\$f4 = 0
$f5 = 0
f6 = 1
f7 = 0
$f8 = 0
$vcreft = 3000
snosm = 6
ELSEIF(($fcas .EQ. 8) .or. ($fcas .EQ. 128) .or. ($fcas .EQ. 136)) THEN
ps1 = 0
$ps2 = 0
$ps3 = 120
ps4 = 240
$ps5 = 60
$ps6 = 60
$ps7 = 180
ps8 = 300
$f1 = 1
\$f2 = 0
\$f3 = 0
\$f4 = 0
f5 = 1
$f6 = 0
\$f7 = 0
$f8 = 0
$vcreft = 3000
snosm = 6
ELSE
ps1 = 0
ps2 = 0
$ps3 = 120
$ps4 = 240
$ps5 = 60
$ps6 = 60
$ps7 = 180
$ps8 = 300
f1 = 1
f2 = 1
f3 = 1
$f4 = 1
$f5 = 1
f6 = 1
```

```
$f7 = 1
$f8 = 1
$vcreft = 0
$nosm = 0
ENDIF
```

Appendix C

Details of HIL Real-Time Simulations

This appendix shows all details of the MMC HIL real-time simulations. As mentioned in Chapter 7, the HIL real-time simulations is done through incorporating a physical FPGA controller (programmed using Labview) with a real-time digital simulator (programmed using RT-LAB/Simulink environment). Thus, the appendix is divided into two sections: the first shows the details of the RT-LAB/Simulink Model which has the MMC model, the fault simulation block, the proposed FTCU, the PS-PWM switching algorithm and the MMC voltages and currents monitoring console. While the second part shows the LABVIEW program which shows the ADALINE and RLS estimation blocks, the proposed capacitor voltage balancing algorithm and the proposed FDU.



C.1 The RT-LAB/Simulink Real-time Simulation Model

Figure C.1: The block of the proposed FDU for one leg.



Figure C.2: The block of submodule including the fault simulation unit.



Figure C.3: The blocks of the proposed FTCU.

The following code is the fault tolerant control sorting algorithm which is installed inside a MATLAB function (as shown in Figure C.3):

```
function
[ps1,ps2,ps3,ps4,ps5,ps6,ps7,ps8,f1,f2,f3,f4,f5,f6,f7,f8,vcreft,no
sm] = FTC(fcas)

if fcas==0
    ps1 = 0;
    ps2 = 90;
    ps3 = 180;
    ps4 = 270;
    ps5 = 45;
    ps6 = 135;
    ps7 = 225;
```

```
ps8 = 315;
    f1 = 0;
    f2 = 0;
    f3 = 0;
    f4 = 0;
    f5 = 0;
    f6 = 0;
    f7 = 0;
    f8 = 0;
       vcreft = 2250;
    nosm = 8;
elseif fcas == 18
    ps1 = 0;
    ps2 = 120;
    ps3 = 240;
    ps4 = 270;
    ps5 = 60;
    ps6 = 180;
    ps7 = 225;
    ps8 = 300;
    f1 = 0;
    f2 = 0;
    f3 = 0;
    f4 = 1;
    f5 = 0;
    f6 = 0;
    f7 = 1;
    f8 = 0;
    vcreft = 3000;
    nosm = 6;
elseif fcas == 20
    ps1 = 0;
    ps2 = 120;
    ps3 = 240;
    ps4 = 270;
    ps5 = 60;
    ps6 = 180;
    ps7 = 180;
    ps8 = 300;
    f1 = 0;
    f2 = 0;
    f3 = 0;
    f4 = 1;
    f5 = 0;
    f6 = 1;
    f7 = 0;
    f8 = 0;
    vcreft = 3000;
    nosm = 6;
elseif fcas == 24
    ps1 = 0;
    ps2 = 120;
    ps3 = 240;
    ps4 = 270;
    ps5 = 60;
    ps6 = 60;
    ps7 = 180;
    ps8 = 300;
    f1 = 0;
    f2 = 0;
    f3 = 0;
```

```
f4 = 1;
    f5 = 1;
    f6 = 0;
    f7 = 0;
    f8 = 0;
    vcreft = 3000;
    nosm = 6;
elseif fcas == 33;
    ps1 = 0;
    ps2 = 120;
    ps3 = 240;
    ps4 = 240;
    ps5 = 60;
    ps6 = 180;
    ps7 = 300;
    ps8 = 300;
    f1 = 0;
    f2 = 0;
    f3 = 1;
    f4 = 0;
    f5 = 0;
    f6 = 1;
    f7 = 0;
    f8 = 0;
    vcreft = 3000;
    nosm = 6;
elseif fcas == 36
    ps1 = 0;
    ps2 = 120;
    ps3 = 240;
    ps4 = 240;
    ps5 = 60;
    ps6 = 180;
    ps7 = 180;
    ps8 = 300;
    f1 = 0;
    f2 = 0;
    f3 = 1;
    f4 = 0;
    f5 = 0;
    f6 = 1;
    f7 = 0;
    f8 = 0;
    vcreft = 3000;
    nosm = 6;
elseif fcas == 40;
    ps1 = 0;
    ps2 = 120;
    ps3 = 240;
    ps4 = 240;
    ps5 = 60;
    ps6 = 60;
    ps7 = 180;
    ps8 = 300;
    f1 = 0;
    f2 = 0;
    f3 = 1;
    f4 = 0;
    f5 = 1;
    f6 = 0;
    f7 = 0;
```

```
f8 = 0;
    vcreft = 3000;
    nosm = 6;
elseif fcas == 65
    ps1 = 0;
    ps2 = 120;
    ps3 = 120;
    ps4 = 240;
    ps5 = 60;
    ps6 = 180;
    ps7 = 300;
    ps8 = 300;
    f1 = 0;
    f2 = 1;
    f3 = 0;
    f4 = 0;
    f5 = 0;
    f6 = 0;
    f7 = 0;
    f8 = 1;
    vcreft = 3000;
    nosm = 6;
elseif fcas == 66
    ps1 = 0;
    ps2 = 120;
    ps3 = 120;
    ps4 = 240;
    ps5 = 60;
    ps6 = 180;
    ps7 = 300;
    ps8 = 300;
    f1 = 0;
    f2 = 1;
    f3 = 0;
    f4 = 0;
    f5 = 0;
    f6 = 0;
    f7 = 1;
    f8 = 0;
    vcreft = 3000;
    nosm = 6;
elseif fcas == 72
    ps1 = 0;
    ps2 = 120;
    ps3 = 120;
    ps4 = 240;
    ps5 = 60;
    ps6 = 60;
    ps7 = 180;
    ps8 = 300;
    f1 = 0;
    f2 = 1;
    f3 = 0;
    f4 = 0;
    f5 = 1;
    f6 = 0;
    f7 = 0;
    f8 = 0;
    vcreft = 3000;
    nosm = 6;
elseif fcas == 129
```

```
ps1 = 0;
    ps2 = 0;
    ps3 = 120;
    ps4 = 240;
    ps5 = 60;
    ps6 = 180;
    ps7 = 300;
    ps8 = 300;
    f1 = 1;
    f2 = 0;
    f3 = 0;
    f4 = 0;
    f5 = 0;
    f6 = 0;
    f7 = 0;
    f8 = 1;
    vcreft = 3000;
    nosm = 6;
elseif fcas == 130
    ps1 = 0;
    ps2 = 0;
    ps3 = 120;
    ps4 = 240;
    ps5 = 60;
    ps6 = 180;
    ps7 = 300;
    ps8 = 300;
    f1 = 1;
    f2 = 0;
    f3 = 0;
    f4 = 0;
    f5 = 0;
    f6 = 0;
    f7 = 1;
    f8 = 0;
    vcreft = 3000;
    nosm = 6;
elseif fcas == 132
    ps1 = 0;
    ps2 = 0;
    ps3 = 120;
    ps4 = 240;
    ps5 = 60;
    ps6 = 180;
    ps7 = 180;
    ps8 = 300;
    f1 = 1;
    f2 = 0;
    f3 = 0;
    f4 = 0;
    f5 = 0;
    f6 = 1;
    f7 = 0;
    f8 = 0;
    vcreft = 3000;
    nosm = 6;
elseif (fcas == 1) || (fcas == 16) || (fcas == 17)
    ps1 = 0;
    ps2 = 120;
    ps3 = 240;
    ps4 = 240;
```

```
ps5 = 60;
    ps6 = 180;
    ps7 = 300;
    ps8 = 300;
    f1 = 0;
    f2 = 0;
    f3 = 1;
    f4 = 0;
    f5 = 0;
    f6 = 0;
    f7 = 0;
    f8 = 1;
    vcreft = 3000;
    nosm = 6;
elseif (fcas == 2) || (fcas == 32) || (fcas == 34)
    ps1 = 0;
    ps2 = 120;
    ps3 = 240;
    ps4 = 240;
    ps5 = 60;
    ps6 = 180;
    ps7 = 180;
    ps8 = 300;
    f1 = 0;
    f2 = 0;
    f3 = 1;
    f4 = 0;
    f5 = 0;
    f6 = 0;
    f7 = 1;
    f8 = 0;
    vcreft = 3000;
    nosm = 6;
elseif (fcas == 4) || (fcas == 64) || (fcas == 68)
    ps1 = 0;
    ps2 = 120;
    ps3 = 120;
    ps4 = 240;
    ps5 = 60;
    ps6 = 180;
   ps7 = 180;
   ps8 = 300;
   f1 = 0;
   f2 = 1;
    f3 = 0;
    f4 = 0;
    f5 = 0;
    f6 = 1;
    f7 = 0;
    f8 = 0;
    vcreft = 3000;
    nosm = 6;
elseif (fcas == 8) || (fcas == 128) || (fcas == 136)
    ps1 = 0;
    ps2 = 0;
    ps3 = 120;
    ps4 = 240;
    ps5 = 60;
    ps6 = 60;
    ps7 = 180;
    ps8 = 300;
```

f1 = 1;f2 = 0;f3 = 0;f4 = 0;f5 = 1;f6 = 0;f7 = 0;f8 = 0;vcreft = 3000; nosm = 6;else ps1 = 0; ps2 = 0;ps3 = 120;ps4 = 240;ps5 = 60;ps6 = 60;ps7 = 180;ps8 = 300; f1 = 1;f2 = 1;f3 = 1;f4 = 1;f5 = 1;f6 = 1;f7 = 1;f8 = 1;vcreft = 0;nosm = 1000000;end



Figure C.4: The blocks of the PS-PWM switching algorithm.



Figure C.5: The Monitoring console.

C.2 The cRIO/LABVIEW Programming Blocks



Figure C.6: The LABVIEW block of the ADALINE capacitor voltage estimation algorithm for one arm.





Figure C.7: The LABVIEW block of the RLS capacitor voltage estimation algorithm for one arm.



Figure C.8: The LABVIEW block of the averaging control for one leg.



Figure C.9: The LABVIEW block of the balancing control for one submodule.



Figure C.10: The LABVIEW block of the proposed FDU (for one submodule).

| | ADALINE Estimation | | RLS Estimation | |
|-------|-----------------------|-----|-------------------|------|
| Landa | Vc1 | Vc5 | Vc 1 | Vc 5 |
| | Ju | JU | JU | 0 |
| Alpha | Vc2 | Vc6 | Vc 2 | Vc 6 |
| 0 | JO | 0 | 0 | 0 |
| Vdc | Vc3 | Vc7 | Vc 3 | Vc 7 |
| | 0 | 0 | 0 | 0 |
| | Vc4 | Vc8 | Vc 4 | Vc 8 |
| | 0 | 0 | 0 | 0 |

Figure C.11: The LABVIEW graphical user interface of the capacitor voltage estimation.

| Fault Detection Unit Capacitor voltage estimation | Capacitor voltage balancing control |
|--|-------------------------------------|
| Vc ref I diff ref + 0 Kp Averaging Kp Averaging + 0 Ki Averaging + + 0 Ki Averaging I upper 0 0 I differential I lower 0 0 | Mod. Index 2 |
| MMC Inner Control | |

Figure C.12: The LABVIEW graphical user interface of the capacitor voltage balancing control.

| Capacitor voltage estimation | Capacitor voltage balancing control |
|------------------------------|-------------------------------------|
| radic Detection onic | |
| Voltage threshold (PU) | F1 F5 |
| Deravitive | F3 F7 |
| Threshold 410 | |
| | F4 F8 |
| | |
| | |

Figure C.13: The LABVIEW graphical user interface of the proposed FDU.

Appendix D

Selected Publications

Capacitor Voltage Balancing Strategy Based on Sub-module Capacitor Voltage Estimation for Modular Multilevel Converters

Mahmoud Abdelsalam, Mostafa Marei, Senior Member, IEEE, Sarath Tennakoon, Member, IEEE, and Alison Griffiths

Abstract—The modular multilevel converter (MMC) is ex-pected to be used extensively in high-voltage direct current (HVDC) transmission networks because of its superior charactersitis over the line-commutated converter (LCC). A key issue of concern is balancing sub-module capacitor voltages in the MMCs, which is critical for the correct operation of these converters. The majority of voltage balancing techniques proposed thus far require that the measurement of the capacitor voltages use a reliable measuring system. This can increase the capital cost of reliable measuring system. This can increase the capital cost of the converters. This paper presents a voltage balancing strategy based on capacitor voltage estimation using the adaptive linear neuron (ADALINE) algorithm. The proposed estimation unit requires only three voltage sensors per phase for the arm reactors and the output phase voltages. Measurements of sub-module capacitor voltages and associated communication links with the central controller are not needed. The proposed strategy can be applied to MMC systems that contain a large number of sub-modules. The method uses PSCAD/EMTDC, with particular focus on dynamic performance under a variety of operating conditions.

Index Terms-Capacitor balancing, HVDC, MMC.

I. INTRODUCTION

VOLTAGE source converters (VSCs) are being used increasingly in high-voltage direct current (HVDC) transmission systems over line-commutated converters (LCCs), particularly for connecting off shore wind farms. The advantages of using VSC include the ability to control active and reactive power independently, the ability to supply weak or passive networks, and VSC's lower space footprint. Although VSCs have many topologies, the modular multilevel converter (MMC) is considered the most suitable topology for high voltage applications, especially for power transmission and distribution. Their advantages include potential to eliminate harmonic filters with

Manuscript received September 4, 2015; revised November 23, 2015 and January 23, 2016; accepted January 25, 2016. Date of publication March 30, 2016; date of current version February 22, 2016. M. Abdelsalam (corresponding author), S. Tennakoon, and A. Grif-fiths are with the Centre of Energy Efficient Systems, Faculty of Com-puting, Engineering and Sciences, Staffordshire University, Stoke on trent, ST4 2DE, U.K. (e-mail: mahmoud.abdelsalam@research.staffs.ac.uk; s.b.tennakoon@staffs.ac.uk; a.l.Griffiths@staffs.ac.uk). M. Marei is with the Electrical Power and Machines Department, Faculty of Engineering, Ain Shams University, Cairo 11517, Egypt (e-mail: mostafa. marei@yahoo.ca) DOI: 10.17775/CSEEJPES.2016.00010

2096-0042 (c) 2016 CSEE

a sufficient number of voltage levels, elimination of the DClink capacitor, and low switching losses [1], [2]. The MMC consists of sub-modules connected in a series,

and forming a leg in each phase. The sub-module can be a halfbridge or a full-bridge and each sub-module has a capacitor that buffers the energy from the DC to the AC side and vice versa, thus eliminating the DC-link capacitor. As shown in Fig. 1, each phase leg is divided into two arms-upper and lower. Each arm has an identical number of sub-modules to generate balanced voltage between the two arms of each phase. Inductors are installed in arms to smoothen and filter the currents [3], [4].



Fig. 1. Three-phase MMC topology and internal sub-module.

This paper presents a new voltage balancing control technique based on capacitor voltage estimation in place of traditional direct measurement. The main contribution of the proposed technique is the elimination of the communication burden used to send the voltage measurements from submodules to the central controller. As a result, not only the cost is reduced, but the wiring and associated problems are also minimized, which renders the proposed strategy attractive for the practical realization of a MMC with a large number of submodules. The proposed control technique has been evaluated using simulations where a number of case studies were conducted to test the dynamic performance under a variety of operating conditions. Section $\ensuremath{\mathrm{II}}$ shows the modeling of the

MMC, including the mathematical representation and the switching algorithm. Section III shows the proposed balancing algorithm while presenting the capacitor voltage estimation technique. Section IV presents the simulation results with some discussions. Finally, a conclusion for the conducted work is presented in section V.

II. MODELING OF MMC

Generally, the operation of the MMC depends on the switching of the sub-module IGBTs. The switching is performed by a switching algorithm, which generates the firing signals to each sub-module [5].

A. MMC Operating Modes

The MMC half bridge sub-module has four operation modes during which energy is transferred:

- Mode 1: When S₁ is closed, S₂ is opened, and the arm has positive polarity. The current flows into the capacitor, charging it. The sub-module is then inserted. See Fig. 2(a).
- Mode 2: When S_1 is opened, S_2 is closed, and the arm has positive polarity. The sub-module is bypassed, and the current flows towards the next sub-module, keeping the capacitor charge constant. The submodule is then bypassed. See Fig. 2(b).
- Mode 3: S₁ is closed, S₂ is opened, and the arm has negative polarity. The capacitor starts to discharge and then the sub-module is inserted. See Fig. 2(c).
- Mode 4: When S₁ is opened, S₂ is closed, and the arm has negative polarity. The current flows towards the next module, keeping the capacitor charge constant, after which and the sub-module is bypassed. See Fig. 2(d).



Fig. 2. Different operating modes of MMC sub-modules. (a) Mode 1. (b) Mode 2. (c) Mode 3. (d) Mode 4.

B. Mathematical Model of MMC

For the purpose of designing an inner control system for the MMC, it is necessary to mathematically analyze the converter. The converter arms, shown in Fig. 3, are represented as variable capacitors connected in a series with arm resistance and inductance [6]. The number of sub-modules per arm and the switching frequency is assumed as infinite to simplify the analysis because of the perfect sinusoidal output voltage and total voltage balancing between arms. The arm capacitance insertion is represented with a modulation value $m_{\rm arm x}$, where x may be u for upper arm or l for lower arm. The $m_{\rm arm x}$

is varied from 0 to 1. For example, when $m_{\rm arm\,u} = 0$, this indicates that all sub-modules in the upper arm are bypassed and when $m_{\rm arm\,l} = 1$, this indicates that all sub-modules in the lower arm are inserted. The value Σv_{Cx} is the sum of arm capacitor voltages. The arm voltage is given by:

$$v_x(t) = m_{\operatorname{arm} x}(t) \cdot \Sigma v_{\operatorname{C} x}.$$
 (1)

The inserted arm capacitance is given by:

$$C_{\operatorname{ins} x} = \frac{C_{\mathrm{SM}}}{N.m_{\operatorname{arm} x}\left(t\right)} \tag{2}$$

where $C_{\rm SM}$ is the capacitance of one sub-module and N is the number of sub-modules per arm. If the arm current is $i_x(t)$, the total capacitor voltage dynamics can be expressed by:

$$\frac{d\Sigma v_{\mathrm{C}x}}{dt} = \frac{i_x\left(t\right)}{C_{\mathrm{ins}\,x}}.$$
(3)

Substituting (2) in (3), the capacitor voltage dynamics can be expressed for the upper and lower arms by:

$$\frac{d\Sigma v_{\rm Cu}}{dt} = \frac{N.m_{\rm arm\,u}.i_{\rm u}}{C_{\rm SM}} \tag{4}$$

The current i_{diff} is the differential current, which circulates between the phase legs. The circulating current of any phase

$$i_{\text{diff}} = \frac{i_{\text{u}} + i_{\text{l}}}{2}.$$
 (6)

By performing the necessary circuit analysis, the output phase voltage v_o is given by:

$$v_{\rm o} = \frac{v_{\rm dc}}{2} - L_{\rm arm} \frac{di_{\rm u}}{dt} - m_{\rm arm \, u} \Sigma v_{\rm Cu} \tag{7}$$

$$v_{\rm o} = -\frac{v_{\rm dc}}{2} + L_{\rm arm} \frac{di_{\rm l}}{dt} + m_{\rm arm\,l} \Sigma v_{\rm Cl}.\tag{8}$$



Fig. 3. The MMC average model.

is described as follows:

C. MMC Switching Algorithm

186

Many switching algorithms have been proposed in the literature [7]–[10], and the modulation techniques that have been used can be divided into two main categories: 1) reference signal based and 2) carrier based. The phase shifted pulse width modulation (PS-PWM), which is one of the well-known ABDELSALAM et al: CAPACITOR VOLTAGE BALANCING STRATEGY BASED ON SUB-MODULE CAPACITOR VOLTAGE ESTIMATION FOR MODULAR MULTILEVEL CONVERTERS 67

advantages [11]:

- 1) Ease of implementation;
- 2) lower losses due to reduced switching frequency; and
- 3) stable performance during dynamic changes.

In the PS-PWM technique, illustrated in Fig. 4, each submodule has a dedicated triangular carrier waveform with the same magnitude, but with a different phase shift.



Fig. 4. The PS-PWM concept.

The switching signal for a sub-module results from comparing its corresponding carrier wave with the sinusoidal voltage reference signal v_{o}^{*} . The phase shift between each consecutive carrier signal is given by:

$$\theta = \frac{360}{N-1}.\tag{9}$$

III. PROPOSED CAPACITOR VOLTAGE ESTIMATION TECHNIOUE

The voltage balancing process is vital for keeping the power quality at the standard level, as it significantly reduces the voltage ripples of the sub-module capacitors [12]–[14]. The proposed algorithm consists of three main stages:

- 1) Capacitor voltage estimation;
- 2) averaging control;
- 3) balancing control for capacitors' voltages.

A. Capacitor Voltage Estimation Technique

The capacitor voltage estimation is performed using the Adaptive Linear Neuron (ADALINE) algorithm. ADALINE is known for its efficiency and its rapid on-line tracking technique for dynamically changing voltage signals. It has been utilized in many applications because of its robust performance, low calculation burden, and accurate results [15], [16]. The ADALINE algorithm is formed by simple calculations that do not consume large computing time, which is very important in the application of capacitor voltage estimation. To estimate the capacitor voltages, the MMC governing equations given

carrier based techniques, is utilized because of the following by (7) and (8) are rearranged and rewritten in the vector form as follows:

$$\frac{v_{\text{dc}}}{2} - v_{\text{o}} - v_{\text{Lu}} = \begin{bmatrix} S_{\text{u1}} & S_{\text{u2}} & \dots & S_{\text{uN}} \end{bmatrix} \begin{bmatrix} v_{\text{cul_est}} \\ v_{\text{cul_est}} \\ \vdots \\ v_{\text{cuN_est}} \end{bmatrix}$$
(10)

$$\frac{v_{dc}}{2} + v_{o} - v_{LI} = \begin{bmatrix} S_{I1} & S_{I2} & \dots & S_{IN} \end{bmatrix} \begin{bmatrix} v_{cI1_ost} \\ v_{cI2_ost} \\ \vdots \\ v_{cIN_ost} \end{bmatrix}$$
(11)

where S_{xi} is the switching state for the *i*th sub-module, $v_{Lu} =$ $L_{\operatorname{arm}} \frac{di_{u}}{dt}$, and $v_{\operatorname{LI}} = L_{\operatorname{arm}} \frac{di_{1}}{dt}$.

The ADALINE algorithm produces a linear combination of its input vector X(k), which represents the switching state of each sub-module whether it is inserted or by-passed at time k, and it is written as follows:

$$\mathbf{X}(k) = \begin{bmatrix} S_{x1} & S_{x2} & \dots & S_{xN} \end{bmatrix}^{\mathrm{T}}$$
(12)

where the suffix T refers to the transpose operation. As shown in Fig. 5, the input vector is multiplied by the weight vector W, which resembles the estimated capacitor voltages, given in (10) and (11):

$$\boldsymbol{W}(k) = \begin{bmatrix} v_{\text{cx1_est}} & v_{\text{cx2_est}} & \dots & v_{\text{cxN_est}} \end{bmatrix}.$$
(13)

This multiplication is performed to produce the predicted linear output $\hat{y}(k) = \mathbf{X}^{\mathrm{T}} \mathbf{W}(k)$. The next step is updating the weight vector using an adaptation algorithm called the Widrow-Hoff delta rule given by [17]:

$$\boldsymbol{W}(k+1) = \boldsymbol{W}(k) + \alpha \frac{\boldsymbol{X}(k)(\boldsymbol{y}(k) - \hat{\boldsymbol{y}}(k))}{\boldsymbol{X}(k)^{\mathrm{T}} \boldsymbol{X}(k)}$$
(14)

where α is the reduction factor and $y(k) = \frac{v_{x}}{2} \pm v_{0} - v_{Lx}$. The adaptation algorithm is responsible for adjusting the weighting vector, which represents the estimated capacitor voltages so that the linear output of the ADALINE, $\hat{y}(k)$ is equal to its target value y(k). When the error between the measured signal y(k) and the estimated signal $\hat{y}(k)$ converges to zero, the ADALINE algorithm decomposes the signal and estimates capacitor voltages.

It is observed that increasing the reduction factor α increases the convergence speed on account of losing stability as the prediction error may increase dramatically. This observation is a common behavior of the Widrow-Hoff delta rule for all the study cases. A practical value for the reduction factor α is 0.002 for this application. This value is determined based on minimizing the error to guarantee system stability [18]. It is worth mentioning that the proposed capacitor voltage estima-tion unit is based on three voltage sensors per phase to measure the voltage across the arm reactors and the output phase voltage. This action enables the implementation of a low-cost centralized controller for the MMC with large numbers of submodules, since voltage measurements of sub-modules and their associated communication links are eliminated.



Fig. 5. Block diagram of the proposed capacitors voltages estimation unit based on the ADALINE algorithm.

B. Averaging Control

C. Balancing Control for Capacitor Voltages

The concept of averaging control is responsible for controlling the average voltage across a complete leg. Averaging control is formed from two cascaded loops as shown in Fig. 6 [19]. The first loop is the voltage control loop where the estimated average voltage, $\hat{v}_{avg} = (\sum v_{cu_{st}} + \sum v_{cl_{st}})/2N$, is subtracted from the reference capacitor voltage, $v_{d_{st}}^*$, and the resultant error is processed through a PI controller to generate the reference signal for the differential current, i_{stff}^* . Another PI controller is utilized in the inner loop to regulate the differential current calculated from (6) at its reference signal. The action of the inner loop controller is the averaging voltage signal, v_{avg}^* .

The balancing control method, presented in Fig. 7, is a centralized control system that generates reference signals for balancing the voltage across sub-module capacitors based on estimated voltages from the ADALINE processing unit First, the controller subtracts the estimated capacitor voltage of each sub-module from the reference capacitor voltage, v_c^* . The resulting errors are passed to simple proportional (P) controllers. The outputs from the P-controllers are multiplied by the sign of their corresponding arm current to form reference signals for balancing the voltage across capacitors, v_{cuLbal} ..., $v_{cuN-bal}$, $v_{clL-bal}$, ..., $v_{clN-bal}$. Finally, to form the modulating signal for a sub-module, which is



Fig. 6. Block diagram of the averaging controller.



Fig. 7. Block diagram of the balancing controller.



Fig. 8. Block diagram of the proposed MMC controller.

the balancing control signal for this sub-module, the average voltage command, $v_{\rm avg}^*$, and the reference phase voltage, $v_{\rm o}^*$, are all added. The PS-PWM technique is then utilized to generate the switching signals for the sub-modules of the MMC [20]. Fig. 8 illustrates a block diagram of the proposed integrated control system of the MMC. It is obvious that the proposed scheme eliminates the need for massive numbers of voltage sensors for the sub-module capacitors and their associated communication system with the central controllers. Consequently, the proposed strategy renders its application for an MMC system with a large number of sub-modules.

IV. SIMULATION RESULTS AND DISCUSSIONS

The proposed control scheme for the MMC is simulated using PSCAD/EMTDC software package. The system parameters are shown in Table I. Different simulation cases are considered to examine the dynamic performance of the proposed control strategy for the MMC under different operating conditions. Primarily, the first task is dedicated to performance evaluation of the proposed ADALINE algorithm in order to estimate capacitor voltages used for controlling the MMC under dynamic reference phase voltage changing conditions. The second case is devoted to assessing the dynamic performance of the proposed control strategy during the boost operation of the MMC. Finally, the third case examines the capabilities of the proposed strategy for estimating capacitor voltages under fault conditions in a sub-module, and for determining the faulty sub-module.

> TABLE I MMC MODEL PARAMETERS

| Parameter | Value |
|-------------------------------|------------|
| Rated power | 1 MW |
| Sub-module rated voltage | 2,250 V |
| Rated DC voltage | 4,500 V |
| Arm inductance | 3 mH |
| Sub-module capacitance | 1,900 µF |
| Number of sub-modules per leg | 8 |
| Load impedance | 30 Ω, 6 mH |

A. Dynamic Performance of the Proposed Capacitor Voltage Estimation Based Balancing Strategy

The purpose of this simulation case is to examine the performance of the proposed capacitor voltage estimation based control algorithm for the MMC under dynamic changes occuring in reference phase voltages. The reference capacitor voltage v_c^* is set at $v_{dc}/N = 2.25$ kV while the reference phase voltage signal v_o^* is dynamically changed from 0.7 p.u. to 1 p.u. at t = 0.2 s. First, the estimation units for capacitor voltages are disabled and the actual measurements are used in the control system. The three-phase voltages, currents, and capacitor voltages are shown in Fig. 9. A similar result to Fig. 9 is obtained in Fig. 10 where the proposed estimation unit for the capacitor voltages is enabled and utilized in the proposed control system shown in Fig. 8. As displayed in Fig. 10(a), the measured three-phase AC voltages follow their reference signals. As expected, only six sub-modules are utilized when = 0.7 p.u., while all the sub-modules are involved when v_0^* = 1 p.u. The three-phase load currents are displayed in Fig. 10(b). Fig. 10(c) traces the capacitor voltages that are grouped in two main trajectories, one for the upper arm and the other for the lower arm. These two trajectories are out of phase and are identical to that presented in Fig. 9(c), where the measured capacitor voltages are used instead of their estimated signals in controlling the MMC. The proposed MMC controller succeeds



Fig. 9. Disabling the proposed estimation unit. (a) Three-phase voltages. (b) Three-phase load currents, (c) Voltages of leg a sub-modules.



Fig. 10. Enabling the proposed estimation unit. (a) Three-phase voltages. (b) Three-phase load currents. (c) Voltages of leg a sub-modules.

in balancing the capacitor voltages at their reference value of 2.25 kV. Moreover, the estimated capacitor voltages and their corresponding actual measurements for the upper and lower arms are presented in Fig. 11 and Fig. 12, respectively.



Fig. 11. Actual and estimated voltages of the upper arm sub-modules of phase a.

Fast tracking with accurate performance of the proposed ADALINE algorithm for estimating the sub-module capacitor voltages are evident. Furthermore, the circulating current is tightly tracking its reference signal, as demonstrated in Fig. 13(a). Fig. 13(b) and Fig. 13(c) illustrate the actions of the averaging controller in Fig. 6 and the balancing controller in Fig. 7 for the first sub-module at the upper arm $v_{\rm cu1\,bal}^*$, respectively. These results reveal efficient utilization of the proposed capacitor voltage estimation techniques for the MMC's control algorithm.







Fig. 13. Different controllers action for leg a. (a) Actual and reference waveform of the circulating current. (b) Averaging voltage reference. (c) Balancing voltage reference for the first sub-module of the upper arm.

B. Dynamic Performance During Boost Operation of MMC

In this test scenario, the reference phase voltage signal v_o^* is kept at 1 p.u., and the capacitor reference voltage command is increased from 2.25 kV to 2.5 kV at t = 0.4 s. Setting v_o^* at values higher than v_{dc}/N results in boost operation of the MMC, which is useful in compensating for the load voltage during sub-module faults.

Fig. 14(a) indicates increase in output three-phase voltages from the MMC at t = 0.4 s when the proposed controller succeeds in boosting and balancing the capacitor voltages at the new set value of 2.5 kV, as illustrated in Fig. 14(c). Yet again, the fast dynamics that tolerates error of the proposed capacitor voltage estimation units are revealed in Fig. 15 and Fig. 16 for the sub-modules at both upper and lower arms, respectively. Boosting the capacitor voltages increases the circulating current, and hence the balancing voltage signal escalates, as demonstrated in Fig. 17.



ABDELSALAM et al.: CAPACITOR VOLTAGE BALANCING STRATEGY BASED ON SUB-MODULE CAPACITOR VOLTAGE ESTIMATION FOR MODULAR MULTILEVEL CONVERTERS

Fig. 14. Boost operation of the MMC. (a) Three-phase voltages. (b) Three-phase load currents. (c) Voltages of leg a sub-modules.



Fig. 15. Actual and estimated voltages of the upper arm sub-modules of phase a during the boost operation.



Fig. 16. Actual and estimated voltages of the lower arm sub-modules of phase a during boost operations.



71

Fig. 17. Different controller actions for leg a during the boost operation. (a) Actual and reference waveform of the circulating current. (b) Averaging voltage reference. (c) Balancing voltage reference for the first sub-module of the upper arm.

C. Performance Under Sub-module Fault

This task is devoted to assessing the capabilities of the proposed strategy for estimating capacitor voltages under fault conditions in a sub-module. A short-circuit fault is programmed to strike the upper switch of the third sub-module in the upper arm of phase a at t = 0.8 s. As a result, the voltage and current of phase a are distorted, as shown in Fig. 18(a) and Fig. 18(b), respectively. Fig. 18(c) indicates the collapse of the DC voltage of the faulty sub-module to zero at the fault instance. Consequently, the voltages of the



Fig. 18. Operation during a fault on the third sub-module of the upper arm. (a) Three-phase voltages. (b) Three-phase load currents. (c) Voltages of leg a sub-modules.

upper arm capacitors increase due to voltage drop in the faulty sub-module, while the voltages of the lower arm capacitors oscillate around the set value.

Fig. 19 and Fig. 20 demonstrate the fast dynamic response of the proposed ADALINE algorithm for estimating the DC two ltages of different sub-modules. This result reveals that the proposed ADALINE algorithm succeeds in determining the faulty sub-module where the greatest deviation of the estimated capacitor voltage occurs. As a result, the proposed estimation unit is a candidate for sub-module fault detection and localization, which are essential tasks for fault-tolerant control of the MMC. During the fault, the circulating current, averaging, and balancing control signals oscillate with the power frequency component, as illustrated in Fig. 21.



Fig. 19. Actual and estimated voltages of the upper arm sub-modules of phase a during a fault on the third sub-module of the upper arm.



Fig. 20. Actual and estimated voltages of the lower arm sub-modules of phase a during a fault on the third sub-module of the upper arm.



Fig. 21. Different controller actions for leg a during the faulty sub-module. (a) Actual and reference waveform of the circulating current. (b) Averaging voltage reference. (c) Balancing voltage reference for the first sub-module of the upper arm.

V. CONCLUSION

This paper presents an estimation unit based on the ADA-LINE algorithm for the sub-module DC voltages of the MMC. For each leg of the MMC, the proposed estimation unit requires only three sensors: one for the phase voltage and the other two for the voltages across the arm reactors. The proposed estimation unit is then integrated into a control strategy to balance the voltages across the sub-modules of the MMC. The proposed technique eliminates the need for direct measurement of capacitor voltages and associated communica-tion systems, which renders it suitable for implementing a lowcost centralized controller for the MMC with a large number of sub-modules. The fast response and accurate tracking of the estimation unit under different dynamic conditions are validated in simulation results. Furthermore, the proposed MMC controller successfully balances the capacitor voltages at their set values even in boost operation mode. Finally, the circulating current is able to tightly track its reference signal, and the results demonstrate that the proposed ADALINE algorithm is able to successfully determine the faulty sub-module, which is required for implementing fault-tolerant control of the MMC.

REFERENCES

- M. Perez, S. Bernet, J. Rodriguez, S. Kouro and R. Lizana, "Circuit topologies, modeling, control schemes, and applications of modular multilevel converters," *IEEE Transactions on Power Electronics*, vol. 30, no. 1, pp. 4-17, 2015.
 W. Li, L. Gregoire, and J. Bélanger, "Control and performance of a modular multilevel converter system," presented at CIGRE Canada Conference on Power Systems, September 2011.
 L. Hui, L. C. Poh, and F. Blaabjerg, "Review of fault diagnosis and fault-tolerant control for modular multilevel converter of HVDC," in proceed-ings of *IECON 2013, Industrial Electronics Society*, pp. 1242–1247, 2013.

- [4] W. Zhang, D. Xu, P. Erjeti, H. Li, J. Hanke, and H. Kristeamoor-thy, "Survey on fault-tolgant techniques for power electronic convert-ers," *IEEE Transactions on Power Electronics*, vol. 29, no. 12, pp. 6519-6534, 2014.

- thy, "Survey on full-tolized techniques for points electronic convertings," IEEE Transmittions a Power Electronics, vol. 29, no. 12, pp. 6319-6354, 2014.
 [3] L. G. Banquado, J. Rodrigues, J. I. Leon, S. Kouro, R. Papillo, and M. A. M. Rus, "The age of multilevel converting arrive," IEEE Industrial Electronics Magazine, vol. 2, no. 2, pp. 38, 39, 2008.
 [6] A. Antooropoulos, L. Vorguitz, and H. R. Hus, "Che dynamics and voltage control of the modular multilevel converting arrive," IEEE Industrial Electronics Magazine, vol. 2, no. 2, pp. 38, 39, 2008.
 [6] A. Antooropoulos, L. Vorguitz, and H. R. Hus, "Che dynamics and voltage control of the modular multilevel converts," in *Proceedings of 13th European Conference on Power Electronics and Applications IEEE (Massimulation of half-tridge consends of multilevel converts, " in <i>Proceedings of 13th European Conference on Power Electronics and Applications IEEE 2009*, 2009, pp. 1242-1243.
 [7] G. Konstantinou and V. Agalidi, "Parlamane swaliation of half-tridge consends mathing connecting and *Applications IEEE 2009*, 2009, pp. 3390-2404.
 [8] A. Hasanpoor, S. Horzy, H. Hee, and L. Angguitz, "Buulation of difficure angrity-based WM methods for modular multilevel converts," *Electronics Conference on IEEE Industrial Bestronics Society IEO/M 2012*, 2012, pp. 3583-393.
 [9] C. I. Odda, "Modular multilevel converts," *Electronics Power Congeneents and Systems*, vol. 43, no. 1, pp. 1-9, 2015.
 [10] B. S. Tin, W. Kio, Le, T. J. Kim, D. W. Kang, and D.-S. Hyun, "A study on the multi-capity PVM methods for wolking balancing of Hying equation multil-vel investige." in *Proceedings of SLA Annual Conference on IEEE Industrial Bestronics Society IEEOM 2012*, 2012, pp. 1-26.
 [11] C. S. Koretantinou and V. G. Agalidi, "Parlamanes evaluation of half-tridge on the tripica," in *Proceedings of Rel IEEE Conference on IEEE Industrial Electronics and Applications IEEE Transmistance on Industry in the Netronica,*

- 200. [15] M. I. Marci, E. F. El-Saadany, and M. M. A. Salama, "Experimental summation of envelope tracking techniques for voltage disturbances," *Electric Power Justen Research* (Elsevier), vol. 80, no. 3, pp. 339–344, 2009. 2040.
- [16] M. I. Marsi, N. El-Sayad, and A. A. El-Saita; "PV interface system with IV/RT capability based on a outpart controlled HEAC link converte," *Social analyle Energy Technologies and Accessments*, vol. 9, pp. 53-62, 2016. 2045
- 2015
 [17] B. Wickow and M. A. Lehr, "30 years of adaptive neural networks: perceptore, machine and backpropagation," *IEEE Proceedings*, vol 78, pp. 1415–1442, 1990.
 [18] H. Stefn, "Robust stability analysis of adaptation algorithms for single perceptors," *IEEE Transactions on Neural Networks*, vol. 2, pp. 325– 328, 1994.
 [10] M. H. Stefn, "Manual M. And M. Stefn," and M. Stefn, and M. Stefn, "Stefner, and M. Andre, "Stefner, and M. Stefner, and St
- 328, 1994.
 [19] M. Hagiwas and H. Akagi, "Control and experiment of pube with-modulated modular multilevel convertes," *IEEE Transmissions on Power Electronetics*, vol. 24, no. 7, pp. 1737–1746, 2009.
 [20] H. Akagi, S. Inoue, and T. Yoshii, "Control and performance of a transfer melless cascade PWM STATOOM with star configuration," *IEEE Transmissions on Industrial Applications*, vol. 43, no. 4, pp. 1041–1049, 2007.

techniques.



Mahmoud Abdekalam received the B.Sc. (Hores) and M.Sa. degrees from *ixab* Academy for Science and Technology, Cairo, Egypter, in 2008 and 2011, repetitively. He is superity working on his Ph.D. degree at Staffordbirle University, Solies on: Dark U.K. Bom 2008 to 2013, he worked as a teaching assistant in the Department of Electrical Engineering and Control, College of Engineering and Technology (arto, Egypt His research interests include power better protection, renewald energy technologies, and application of power electronics in power systems and control systems technologies.



Mosta L Marsi (SM'2014) received the BSe. (Krcs.) and MSe degree from Ain Shame Uris westy, Caino, Eppt, in 1971 and 2000, respec-tor westy, Caino, Eppt, in 1971 and 2000, respec-tor Warsho, Watsho, CH, Canada, in 2004, all in decinial engineering from 2004 to 2006 its was a postdotraal fellow with the University of Warsho, Ourarily, he is an Associate Rodeser Warsho, Currently, he is an Associate Rodeser with the Dopartment of Electrical Power and Ma-chine, Ani Shame University. His research interests artificial intelligent applications in power yeathers, D. Marci hold the State Iouritie Award in Engineering Sciences, Academy of Scientific Research and Technology, Egpt. His biography is listed in Marguis Who's Who in the Wald

World



Satath B. Tanakoon (M'87) gassing the BSa degree in electrical engineering from the Univer-sity of Magauean, Magauean, Magauean, Si Lanka, the MSa degree from the University of Actor, Ehrmingham, K., and the Ph.D. degree in electrical engineering and electronics from the University of Central Lar-cashire, Lancashire, U.K. He search interests are DC Bregge Efficient Systems, Statfackkire University engine and HVDC, power system protection, and power electronics.



Alison Griffiths received the B.Eng (Hons.), M.Eng, and Ph.D. degrees from Stating during the versity Stoks-on-Text, UK., in 1996, 1999, and 2004, respectively. She is expensibly working as a full-time senior lecturg within the Faculty of Com-puting Breginsering and Sciences, Stating during Uni-versity Stoks-on-Text, UK. She also worked with several comparise on industrial collaborations. Hy rescarch interests include control system technolo-gies, estimation and optimization technologies, with a solution less senior networks and battary life optimization

An Integrated Control Strategy with Fault Detection and Tolerant Control Capability Based on Capacitor Voltage Estimation for Modular Multi-Level Converters

Mahmoud Abdelsalam, Member, IEEE, Mostafa I. Marei, Senior Member, IEEE and Sarath Tennakoon, Member, IEEE

Abstract—Modular Multilevel Converter (MMC) will be extensively used in High-Voltage Direct Current (HVDC) transmission networks because of its superior characteristics over Line Commutated Converter (LCC). Increasing the reliability of the MMC is directly related to the balancing of the MMC sub-module capacitors voltages which guarantees the proper operation of the converter and lowers the stress on the sub-modules. This paper presents an adaptive voltage balancing strategy based on capacitor voltage estimation utilizing a hybrid ADALINE-RLS scheme. The proposed strategy eliminates the need of measuring sub-modules capacitor voltages and associated communication link with the central controller. Furthermore, the estimated capacitor voltages are utilized to detect and localize different types of sub-module faults. After isolating the faulty sub-modules, the proposed fault tolerant control unit (FTCU) modifies the parameters of the voltage balancing strategy to overcome the reduction of the active submodules. The dynamic performance of the proposed strategy is investigated, using PSCAD/EMTDC simulations and hardware in the loop (HIL), real-time simulations, under different normal and faulty operating conditions. The accuracy and the time response of the proposed fault detection and tolerant control units result in stabilizing the operation of the MMC under different types of faults. Consequently, the proposed integrated

Index Terms—Balancing control, fault detection, fault tolerant control, modular multilevel converter, voltage estimation.

I. INTRODUCTION

 V^{OLTAGE} Source Converters (VSC) are increasingly being used in high-voltage direct current (HVDC)

transmission systems, in particular for the grid connection of off-shore wind farms, due to their advantages over the line commutated converters (LCC). These include: the ability to control active and reactive power independently, the ability to supply weak or passive networks and the lower footprint. Although the VSCs have many topologies, the modular multilevel converter (MMC) is considered to be the most suitable topology for high voltage applications, in particular for power transmission and distribution due to their features such as: the possibility to eliminate harmonic filters with a sufficient number of voltage levels, elimination of the de-link capacitor and low switching losses [1][2].

The voltage balancing process is vital for proper operation of the MMC and to keep the power quality at the standard level [3-5]. Many capacitor voltage balancing techniques have been proposed in literature. However, few of them are considered to be practical and suitable for the control of industrial MMCs. One of the promising techniques is the arm energy control technique. The philosophy behind this technique is to control the energy stored in the whole leg and the difference of stored energy between the upper and lower arms [6]. Despite the accurate performance of this method, it suffers from many drawbacks such as distortion of the output current and the complexity of implementation due to large number of calculations. In addition, the arm energy control technique needs continuous measurements of large number of signals. Another capacitor voltage balancing strategy, proposed in [7], is based on regulating the capacitor voltage in each sub-module independently at the set value through monitoring the capacitor voltage of each sub-module [7]. Although the performance of this method is more attractive compared to the energy control, it has a main drawback of high value of differential current. Moreover, it needs the continuous measurements of capacitor voltages.

The reliability of MMC can be improved by implementing a voltage balancing strategy with fault detection and fault tolerant control capabilities that enable the converter to be operated under fail-to-safe operation principle with satisfactory performance [8][9]. Classifying the faults, which the MMC sub-modules may be subjected to, is considered to be the first step in implementing the fault detection and diagnosis algorithm. The sub-module is considered to be building unit in the MMC and it may be exposed to different fault types, that threaten the safety and reliability of the MMC such as open-circuit or short-circuit faults of switches or capacitors [10] [11]. A survey presented in [12] states that 30% of failures caused in power electronic converters are due to the switching devices such as IGBTs and diodes. Based on this statement, the quick detection and localization of IGBT faults is a necessity and can significantly increase the reliability of the MMC. In [12], a hardware based technique is proposed for detecting IGBTs faults. This technique depends on measuring the rate of change of the collector current and comparing it with a pre-defined threshold. The cost of this method is extremely high since each IGBT has a sophisticated monitoring circuit [12]. Another method presented in literature depends on the charge rate of the gate voltage which offers a better sensitivity but still has a high cost impact due to the need of having a dedicated hardware circuit [13]. The artificial neural network (ANN) is used to analyze the historical data of the MMC output voltage and current. Although this method is based on computations and does not need extra physical components, its accuracy is not guaranteed if the voltage and current levels are changed due to dynamic changes. Moreover, the training of the ANN consumes a lot of effort and does not cover all cases [14].

The objective of the fault tolerant control system is to overcome the effect of the presence of faulty sub-module and restoring the stability of the power electronics converter with acceptable performance level. The concept of fault tolerant control is gaining a lot of research interest especially for the MMC, due to the vulnerability of the converter to various types of faults. Additionally, it is not accepted to disconnect the MMC during the operation to assure the security of supply [15]. The fault tolerant control system can be conducted by either using redundant sub-modules or redundant IGBTs as described in [16] and [17], respectively. In addition, other fault tolerant control techniques based on hardware circuits to modify the MMC control system are presented in [18].

The majority of the voltage balancing, fault detection and fault tolerant control techniques proposed in literature require reliable measurements of the capacitor voltages which increases the MMC cost. Moreover, there is no much research has been performed to provide centralized control systems for the MMC in normal and abnormal conditions and utilizes the concept of fault tolerance. This paper presents a new voltage balancing control technique with fault detection and fault tolerant control capability based on capacitors voltage estimation instead of direct measurement. The main contribution of the proposed technique is the elimination of the communication burden used to send the voltage measurements from sub-modules to the central controller. As a result, not only the cost is reduced, but the wiring and its associated problems is minimized which renders the proposed strategy attractive for the practical realization of a MMC with a large number of sub-modules. Another important contribution is that the central MMC controller has fault detection and tolerant control capability without the need of redundant sub-modules or adding special power circuit. Various case studies are conducted and presented to evaluate the dynamic performance of the proposed integrated control strategy under different operating conditions. The rest of this paper is organized as follows. The operation concept, the mathematical analysis and the switching algorithms of the MMC are described in section II. Section III presents two capacitor voltage estimation techniques that are used in the proposed MMC control system. The proposed capacitor voltage balancing technique is explained in section IV Section V clarifies the proposed fault detection and tolerant control strategies. The results obtained from simulation and Hardware In the Loop (HIL) are provided with discussions in sections VI and VII, respectively. Finally, Section VIII $\,$ concludes the paper.

II. THE OPERATION PRINCIPLE OF MMC

Generally, the operation of the MMC depends on the switching of the sub-module IGBTs. The switching is performed through a switching algorithm which generates the firing signals to each sub-module [19].

A. The structure of the MMC

The MMC consists of sub-modules connected in series forming a leg in each phase. The sub-module can be a halfbridge or a full-bridge and each sub-module has a capacitor which buffers the energy from the dc to the ac side and vice versa, thus the dc-link capacitor is eliminated. A high speed bypass switch is connected to the output port of the submodule to isolate it in case of fault occurrence. As shown in Fig.1, each phase leg is divided into two arms upper and lower. Each arm has an identical number of sub-modules to

generate balanced voltage between the two arms of each phase. Inductors are installed in arms to smooth and filter the currents [20] [21].



Fig. 1 Three-phase MMC topology and internal sub-module

B. The MMC mathematical model

For the purpose of designing an inner control system for the MMC, it is necessary to mathematically analyze the converter. The converter arms are represented as variable capacitors connected in series with the arm resistance and inductance [6]. The number of sub-modules per arm and the switching frequency are assumed infinite to simplify the analysis because of the perfect sinusoidal output voltage and total voltage balancing between arms. The arm capacitance insertion is represented with a modulation value $m_{arm x}$, where x may be u for upper arm or l for lower arm. Them_{arm x} is varied from 0 to 1. For example, when $m_{arm u} = 0$, this indicates that all sub-modules in the upper arm are bypassed and when $m_{arm l} = 1$, this indicates that all sub-modules in the lower arm are inserted. The value Σv_{Cx} is the sum of arm capacitors voltages. The arm voltage is given by:

$$v_x(t) = m_{armx}(t) \cdot \Sigma v_{Cx} \tag{1}$$

 $C_{insx} = \frac{c_{SM}}{N.m_{arm_x}(t)}$ where C_{SM} is the capacitance of one sub-module and N is the the total capacitor voltage dynamics can be expressed by: $d\Sigma$

$$\frac{dt}{dt} = \frac{i_x(t)}{C}$$
(3)

Substituting (2) in (3), the capacitor voltage dynamics can be expressed for the upper and lower arms by: $d\Sigma v_{0}$

$$\frac{c_u}{c_{SM}} = \frac{N.m_{arm\,u}\cdot l_u}{C_{SM}} \tag{4}$$

$$\frac{d\Sigma v_{Cl}}{dt} = \frac{N.m_{arm\,l.\,l_l}}{C_{cru}}$$
(5)

The current \vec{i}_{diff} is the differential current which circulates between the phase legs. The circulating current of any phase is described as follows: (6)

 $i_{diff} = \frac{i_u + i_l}{2}$ By performing the necessary circuit analysis, the output phase voltage v_o is given by:

$$v_o = \frac{v_{dc}}{2} - L_{arm} \frac{dt_u}{dt} - m_{arm\,u} \Sigma v_{Cu} \tag{7}$$

$$v_o = -\frac{v_{dc}}{2} + L_{arm} \frac{dt_l}{dt} + m_{arm\,l} \Sigma v_{Cl} \tag{8}$$

197

dt

C. The MMC switching algorithm

Many switching algorithms have been proposed in literature [22-25]. The modulation techniques that has been used can be divided to two main categories: reference signal based and carrier based. The phase shifted pulse width modulation (PS-PWM), which is one of the well-known carrier based technique, is utilized due to its case of implementation, reduced switching frequency and hence switching losses. Moreover, the PS-PWM is characterized by stable performance during dynamic changes [26].

In the PS-PWM technique, each sub-module has a dedicated triangular carrier waveform with the same magnitude but with a different phase shift. The switching signals for a sub-module result from comparing its corresponding carrier wave with the sinusoidal voltage reference signal \mathbf{v}_{o}^{*} . The phase shift between each consecutive carrier signals is given by:

 $\theta = \frac{360}{N-1}$

111. ESTIMATION OF CAPACITOR VOLTAGES

(9)

The capacitor voltage estimation is the extremely important step of the proposed control strategy. Although the capacitor voltage estimation eliminates the use of direct measurements, it is a must to assure the estimation accuracy to avoid wrong control actions. Two capacitor voltage estimation techniques are presented. The first is achieved by using the adaptive linear neuron (ADALINE) algorithm which is needed in the proposed fault detection algorithm, while the second algorithm is the recursive least squares (RLS) algorithm which is used in the proposed capacitor voltage balancing control strategy and the proposed fault detection algorithm as well.

A. The capacitor voltage estimation using ADALINE technique

The capacitors voltages estimation is performed by utilizing the Adaptive Linear Neuron (ADALINE) algorithm. The ADALINE is known as being a very efficient and quick on-line tracking technique for the tracking of dynamically changing voltage signals. It is utilized in many applications because of its robust performance, low calculation burden and accurate results [27] [28]. The ADALINE algorithm is formed by simple calculations that do not consume large computing time which is very important in the application of capacitor voltage estimation. To estimate the capacitors voltages, the MMC governing equations given by (7) and (8) are rearranged and rewritten in the vector form as follows:

$$\frac{v_{dc}}{2} - v_o - v_{l,u} = [S_{u1} \quad S_{u2} \quad \dots \quad S_{uN}] \begin{bmatrix} v_{cul, est} \\ v_{cul, est} \\ \vdots \\ v_{cul, est} \end{bmatrix}$$
(10)
$$\frac{v_{dc}}{2} + v_o - v_{Ll} = [S_{l1} \quad S_{l2} \quad \dots \quad S_{lN}] \begin{bmatrix} v_{cl1, est} \\ v_{cl2, est} \\ \vdots \\ v_{cl2, est} \\ \vdots \end{bmatrix}$$
(11)

where S_{xi} is the switching state for the i^{th} sub-module, $v_{Lu} = L_{arm} \frac{di_u}{dt}$, and $v_{Ll} = L_{arm} \frac{di_l}{dt}$

To eliminate the effect of the arm inductance on the estimation accuracy, the arm inductor voltages are measured and used in (10) and (11) for the proposed capacitor voltage estimation techniques. In turn, the proposed capacitor voltage estimation unit does not require the knowledge of the circuit parameters.

The ADALINE algorithm produces a linear combination of its input vector X(k) which represents the switching state vector $[S_{x1} \quad S_{x2} \quad \dots \quad S_{xN}]^T$ at time k, where the suffix Trefers to the transpose operation. In this technique, the input vector is multiplied by the weight vector W(k) which resembles the estimated capacitors voltages, given in (10) and (11), to produce the predicted linear output $\hat{y} = X^T W(k)$. The next step is computing the error signal and adjusting the weights to eliminate the error using the delta rule, also known as Widrow-Hoff learning rule, which is given by [29]: $W(k + 1) = W(k) + \alpha \frac{X(k)(y(k) - \hat{y}(k))}{k!}$ (12)

$$Y(k+1) = W(k) + \alpha \frac{W(k) - V(k)}{\chi(k)^T \chi(k)}$$
(12)

where α is the reduction factor, it is observed that increasing this factor increases the convergence speed on account of losing stability as the prediction error may increase dramatically. The reduction factor α is set at 0.002, for this application, to minimize the error and hence to guarantce system stability [30]. When the difference between the measured signal y(k) and the estimated signal $\hat{y}(k)$ converges to zero, the ADALINE algorithm decomposes the signal and estimates the capacitors voltages. It is worth mentioned that the proposed capacitors voltages estimation unit is based on three voltage sensors per phase to measure the voltage across the arm reactors and the output phase voltage. This action enables the implementation of a low cost centralized controller for the MMC with a large number of sub-modules since the voltage measurement of sub-modules and their associated communication link are eliminated.

B. The capacitor voltage estimation using RLS technique

The fundamental RLS algorithm is formed by two basic equations. The first equation, given in (13), is to update the estimate vector $\hat{\theta}(k)$ of the sub-modules capacitor voltages, while the second equation is used to update the weighting matrix P(k) as shown in (14).

$$\hat{\theta}(k) = \hat{\theta}(k-1) + P(k)\hat{\theta}(k)[y(k) - \phi(k)^T \hat{\theta}(k-1)]$$
(13)

$$P(k) = \frac{1}{\lambda} \Big[P(k-1) - \frac{P(k-1)\phi(k)\phi(k)^T P(k-1)}{\lambda + \phi(k)^T P(k-1)\phi(k)} \Big]$$
(14)

$$\begin{split} P(k) &= \frac{1}{\lambda} \Big[P(k-1) - \frac{P(k-1)\phi(k)\phi(k)}{\lambda + \phi(k)^{T}P(k-1)\phi(k)} \Big] \quad (14) \\ \text{where } \phi(k) \text{ is the connection vector which represents the switching state vector } \begin{bmatrix} S_{x1} & \dots & S_{xy} \end{bmatrix}^{T}$$
 at time k and λ is the forgetting factor that controls the speed of updating the weight matrix in (14). The result of multiplying $\phi(k)^{T}$ by $\hat{\theta}(k-1)$ gives the predicted output, \hat{y} [31]. This predicted signal is then subtracted from the measured signal and the error is used to update the estimate vector. When the estimation error approaches zero, the convergence is reached and the estimated sub-module capacitor voltages are ready. It is clear that the calculation burden of the RLS is slightly more than the ADALINE; however, the RLS is characterized by its fast dynamic response. Fig. 2 illustrates the general block diagram of the capacitor voltages. \end{split}



. . .

IV. THE PROPOSED CONTROL STRATEGY FOR THE MMC

One of the main advantages of the proposed control strategy for the MMC is that it has the same structure for normal condition operation and for the proposed fault tolerant control. The proposed control strategy for the MMC is divided to two main controllers, the averaging controller and the balancing controller for the capacitors' voltages. The following subsections detail each controller.

A. The averaging controller

The averaging controller is formed from two cascaded loops that are responsible of controlling the average voltage across one leg [7]. As shown in Fig. 3, the outer loop sets the reference signal for the differential current, i_{atff}^* , using a simple PI controller to process the error between the estimated average voltage using the RLS algorithm, $\hat{v}_{avg} = (\sum v_{cul\,RLS} + \sum v_{cli\,RLS})/2N_{eff}$, and the desired reference capacitor voltage v_c^* , where N_{eff} is the number of effective sub-modules connected in the arm. The inner loop restrains the differential current, calculated from (6), at its reference signal by utilizing another PI controller that generates the averaging voltage signal, v_{avg}^* .



Fig.3: Block diagram of the averaging control sys

B. The balancing controller

The balancing controller is a centralized control system that generates reference signals to balance the voltage across the sub-modules capacitors based on the estimated voltages from the ADALINE processing unit as demonstrated in Fig.



Fig.4: Block diagram of the balancing control loops.

Firstly, the error between the estimated capacitor voltage and the reference capacitor voltage is calculated for each submodule. Simple proportional (P) controllers process these errors. The actions of the P-controllers are multiplied by the sign of their corresponding arm current to form reference signals for balancing the voltage across capacitors, $v_{cul,bal}$,... $v_{cul,bal}$, v_{cul

Finally, the modulating signal for a sub-module is formed by adding the balancing control signal for this sub-module, the average voltage command, v_{avg}^* , and the reference phase voltage, v_{o}^* , together. The PS-PWM technique is utilized to generate the switching signals for the sub-modules of the MMC [32]. Fig. 5illustrates a block diagram of the proposed integrated control system of the MMC. It is obvious that the proposed scheme eliminates the need of massive numbers of voltage sensors for the sub-module capacitors and their associated communication system with the central controllers. Consequently, the proposed strategy renders its application for a MMC system with a large number of submodules.



Fig.5: Block diagram of the proposed MMC control strategy.

V. THE PROPOSED FAULT DETECTION AND DIAGNOSIS ALGORITHM

The proposed fault detection algorithm, illustrated in Fig. 6, is designed based on the reduction percentage of the estimated per-unit voltage and the rate of change of the difference between the estimated voltages from the ADALINE and the RLS algorithms for each sub-module.

The proposed fault detection unit (FDU) receives the RLS estimated sub-module capacitor voltages $v_{cxi RLS}$ from the RLS algorithm, where x = u or l and i = 1, 2, 3, ..., N. The estimated values are divided by the reference capacitor voltage v_c^* to obtain the per-unit values $v_{cxiRLS pu}$. Instantaneously, the error between the ADALINE and RLS estimated capacitor voltages of each sub-module is calculated and passed to a high speed differentiator to calculate the rate of change of the error signals $\frac{de_{xi}}{dt}$. It is worthy mentioned that the proposed FDU is based on the difference between the dynamic behaviors of the ADALINE and RLS algorithms results from their different natures. Consider the Widro-Hoff rule, (12), which is used for the ADALINE to update estimated voltages. At each iteration, the weighting vector W(k + 1) is updated by adding the error reduction component $\alpha \frac{X(k)(y(k)-\hat{y}(k))}{X(k)^T X(k)}$ to the previous value, W(k). This action indicates that the ADALINE algorithm has no memory because of the independence of the Widro-Hoff rule on the accumulative value of weighting vector. Moreover, the reduction factor α controls the convergence speed; a high value of α increases the convergence period. Therefore, the ADALINE algorithm is characterized by fast tracking under sudden changes of the capacitor voltages. However, it needs relatively long time until reaching convergence. Unlike the ADALINE algorithm, the process of updating the weighting vector of the RLS algorithm depends on the accumulation of historic values of the capacitor voltages as demonstrated in equations (13) and (14). As a result, the RLS algorithm is characterized by lower convergence time when compared with the ADALINE algorithm. However, in the incident of having a sudden change in the capacitor voltage, which occurs during sub-module faults, the dynamic response of the RLS algorithm is slower than that of the ADALINE algorithm.

The next step is to check the estimated per-unit voltage signal, for each sub-module, whether it is higher than 85% or

not. If the comparison result is true, the sub-module is considered to be in healthy condition. Otherwise, the FDU compares the rate of change of the error signal with a predefined threshold which is selected based on the required sensitivity. If the rate of change of the error signal exceeds the threshold, the FDU considers the corresponding sub-module in faulty condition. However, a counter is used to ensure that the steadiness of the estimated signals for a continuous ten counts before indicating a faulty sub-module by setting its corresponding flag, f_i . It is worthy mentioned that the execution time of the proposed FDU is assumed 100 µs; hence the minimum detection time is 1 ms.

The reliability and robustness of the proposed FDU result from its hybrid operation where the fault decision is not only based on the reduction on the estimated per-unit voltage but also on the rate of change of the difference between the estimated voltages from the proposed ADALINE and RLS techniques. The difference between the dynamic behaviors of the two proposed estimation techniques results from their different adaption rules. Moreover, this method does not need any extra components or devices as it is based on the estimation algorithms that are simultaneously utilized in the proposed control of the MMC.



Fig.6: Flowchart of the proposed fault detection algorithm

VI. THE PROPOSED FAULT TOLERANT CONTROL STRATEGY

The proposed fault tolerant control unit (FTCU) has two main tasks: the first is to isolate the faulty sub-modules while the second is to modify the capacitor balancing algorithm. The proposed FTCU, presented in Fig. 7, utilizes the flags of the faulty sub-modules set by the FDU. Assume that there is a fault in one sub-module in the upper arm, the fault tolerant control strategy isolates the faulty sub-module and another one in the lower arm by activating the bypass switches and disabling their gating signals. The reason of isolating a healthy sub-module in the adjacent arm is to retrieve the energy balance between the two arms by equating the number of the activated sub-modules in the upper and lower arms.



As a result, the capacitor reference voltage v_c^z is boosted, by the gain $\frac{N}{N_{eff}}$, to compensate for the reduced number of submodules, while keeping the desired output voltage. At the same time, the proposed FTCU modifies the phase shift between the carrier signals for the rest of the sub-modules in the leg according to (9). This strategy is designed to permit the converter to continue its operation provided that the number of faulty sub-modules per arm does not reach a predefined value based on the number of sub-modules and the ratings of the devices. Otherwise, the control system considers the MMC suffering from cascading failure and in this case, it isolates all the sub-modules and deactivates the converter. It is important to mention that the sub-module capacitor must be carefully selected taking into consideration the worst case condition at which the maximum number of faulty sub-module can be tolerated. This can be done by calculating the maximum energy variation, the boosted up capacitor voltage level and the allowable voltage ripples and selecting the capacitor value from [33]:

$$C_{SM} > \frac{2 \log p \max}{N_{eff} V_c^2 \left(1 + k_{rinnle} \max\right)^2}$$
(15)

where V_c is the rated capacitor voltage, $k_{ripple \ max}$ is the maximum allowable ripple percentage and $\Delta e_{p \ max}$ is the maximum energy variation.

VII. SIMULATION RESULTS AND DISCUSSIONS

The proposed control scheme for the MMC is simulated using PSCAD/EMTDC software package. The system parameters are shown in Table 1. Different simulation cases are considered to examine the dynamic performance of the proposed control strategy for the MMC under different operating conditions. Moreover, the dynamic behavior of the proposed FDU is tested under different healthy and faulty condition. Primarily, the first task is dedicated to evaluate the performance of the proposed scheme based on the RLS
algorithm to estimate the capacitors voltages used to control the MMC under the dynamic reference phase voltage changing condition. The second case is devoted to assessing the dynamic performance of the proposed fault detection and fault tolerant control units during open-circuit fault in a submodule. Finally, the third case examines the capabilities of the proposed FDU and FTCU to stabilize the operation of the proposed control strategy for the MMC under short-circuit fault na sub-module.

| TABL MMC MODEL B | E I ARAMETERS | | | |
|-------------------------------|------------------|--|--|--|
| MMC model parameters | | | | |
| Rated power | 1 MW | | | |
| Sub-module rated voltage | 2250 V | | | |
| Rated dc Voltage | 9000 V | | | |
| Arm inductance | 3 mH | | | |
| Sub-module capacitance | 1900 µF | | | |
| Number of sub-modules per leg | 8 | | | |

A. Performance under IGBT open circuit fault

This task is devoted to assess the capabilities of the proposed control strategy and FDU under IGBT open circuit fault in a sub-module. The upper switch of the third submodule in the upper arm of phase *a* is intentionally opened at t = 0.5s. The fault detection and fault tolerant control units are enabled. The FDU succeeds to detect and localize the fault after 33 ms from its instant of activation as illustrated in Fig 8(a). With the slaw rate of increase of the capacitors voltages, the 33 ms period of detecting open-circuit fault is enough to guarantee the safety of the MMC. Once the fault of the third sub-module in the upper arm is detected, the FTCU isolates it and the third sub-module in the lower arm to balance the energy between the two arms. Furthermore, the FTCU increases the v_c^* from 2.25 kV to 3 kV, as shown in Fig. 8 (b), and adjusts the phase shift of the carrier waves from 45° to 60° to compensate for the loss of two isolated sub-modules. These actions are sufficient to tolerate the fault and restore the stability of the MMC control system as appear from the results of Fig. 9 (a) and (b) where the three-phase output voltages and currents are balanced. The capacitor voltages of the remaining sub-modules restore their balance around the new desired value set by the FTCU as indicated in Fig 9 (c). As expected, the capacitor voltages are slightly increased at the moment of fault detection, t = 0.533s. It is worth mentioned that the capacitor voltages of the isolated submodules freeze at their values when isolation occurs



Fig. 8: Dynamic performance of the proposed FDU and FTCU under an open-circuit fault in a sub-module.

Fig. 10 and 11 indicate the actual and estimated capacitor voltages of the upper and lower arms of phase a, respectively. Both the ADALINE and RLS algorithms provide accurate estimate of the capacitor voltage, even during transient

periods. It is obvious that the performance of the RLS algorithm is more accurate than that of the ADALINE algorithm for estimating the time-varying capacitor voltage signal. As indicated from (10) and (11), the estimation of the capacitor voltages is based on the switching signals. In turn, the proposed estimation algorithms cannot estimate the voltage of faulty or disabled sub-modules as indicated in Fig. 10(c) and Fig. 11(c). At the instant of open-switch fault, the difference between the estimated voltages from the ADALINE and the RLS algorithm increased and excites the proposed FDU to detect the fault quickly.

With the help of the FTCU, the averaging and balancing control loops are able to quickly overcome the consequences of fault and forces the differential current to reach its rated value prior the fault instant, as displayed in Fig. 12(a). Moreover, the averaging and balancing control signals retain their stability as illustrated in Fig. 12(b) and (c), respective



Fig. 9: Enabling the proposed FTCU for an open-circuit fault: a) Three-phase voltages, b) three-phase load currents, c) voltages of leg a sub-modules.



Fig. 10: Actual and estimated voltages of the upper arm sub-modules of phase *a*, under an open-circuit fault.



Fig. 11: Actual and estimated voltages of the lower arm sub-modules of phase a, under an open-circuit fault.

201



current, b) averaging voltage reference, c) balancing voltage reference for the first sub-module of the upper arm

B. Performance under IGBT short circuit fault

This case is conducted to investigate the capabilities of the proposed control strategy and the FDU under IGBT short circuit fault in a sub-module. A short-circuit fault is applied to the upper switch of the third sub-module in the upper arm of phase a at t=0.5s and the proposed FDU and FTCU are enabled. The FDU succeeds to detect and localize the shortcircuit fault after 2.2 ms from its incidence as demonstrated in Fig 13 (a). This detection time is fast as the needed time for clearing short circuit faults must not exceed 5 ms [34]. At the instant of fault detection, the FTCU has isolated the faulty sub-module in the upper arm and its corresponding third submodule in the lower arm to balance the energy between the two arms. Furthermore, the FTCU increases the v_c^* from 2.25 kV to 3 kV as displayed in Fig. 13 (b), and modifies the phase shift of the carrier waves from 45° to 60° to compensate for the loss of two sub-modules. Fig. 14 (a) and (b) show the three-phase output voltages and currents which are balanced. The capacitor voltages of the remaining sub-modules restore their balance at the new set value as illustrated in Fig 14(c).



Fig. 13: Dynamic performance of the proposed FDU and FTCU under a short-circuit fault in a sub-module

The proposed RLS and ADALINE algorithms are accurately estimating the capacitor voltages during the short-circuit fault for all healthy sub-modules as indicated in Fig. 15 and 16. Regarding the faulty sub-module, the RLS and ADALINE estimated signals are different and at the same time the RLS estimated voltage is by far below the voltage threshold. This helped the FDU to quickly detect the shortcircuit fault condition. Fig. 17(a) shows that the FTCU stabilizes the operation of the averaging and balancing

control loops which succeed to limit the magnitude of the differential current and to tightly track its reference. Additionally, the improved averaging and balancing control signals are portrayed in Fig. 17(b) and (c), respectively.



Fig. 14: Enabling the prope ed FTCU for a short-circuit fault: a) Three-phase voltages, b) three-phase load currents, c) voltages of leg a sub







Fig. 16: Actual and estimated voltages of the lower arm sub-modules of phase a, under a short-circuit fault.



FTCU for a short-circuit fault: a) Actual and reference waveform of the circulating current, b) averaging voltage reference, c) balancing voltage reference for the first sub-module of the upper arm.

VIII. HARDWARE IN THE LOOP RESULTS

The HIL test platform is used to validate the proposed control method. This platform is based on the RT-LAB realtime simulation software. As shown in Fig. 18 (a), the platform consists of two main parts: the real-time digital simulator and the real-time physical controller.



(b) Fig.18: The HIL platform system components: a) the block diagram of the HIL platform, b) the HIL test system.

The MMC with the parameters given in table 1 is simulated using the OP-4510 real-time digital simulator made by OPAL-RT with a CPU that operates at a time step of 25 µs. In addition, the real-time digital simulator provides the physical real time controller with needed measurements such as the ac-side voltages and currents, dc voltage, and arm currents. On the other hand, it receives gating signals for IGBTs from the physical controller. The real-time physical controller is a Virtex-5 LX based CRIO- 9024 FPGA controller, which operates at a time step of 25 ns, made by National Instruments and programmed using the LABVIEW software. The physical controller performs the proposed capacitor balancing, fault detection and tolerant control techniques, and the switching algorithm of the MMC. Fig 18 (b) illustrates the HIL test system where the gating signals and system measurements are exchanged through copper wires.

Firstly, an open-circuit fault is applied to the upper switch of the third sub-module in the upper arm of phase *a*, then after 0.3s a short-circuit fault is applied to the first sub-module in the lower arm of phase *a* while enabling the proposed FDU and FTCU. The FDU succeeds to detect the first fault after 35.4ms as shown in Fig. 19(a). Moreover, the FDU identifies the second fault after 3.2ms as shown in Fig. 19 (b). Fig. 20(a) and (b) portray the three-phase output voltages and currents which are balanced even during the application of the two faults.

The capacitor voltages of the remaining sub-modules of the upper and lower arms restore their balance at the new set value as illustrated in Fig 21(a) and (b), respectively. Consequent to the detection of the first fault, the proposed FTCU isolates the faulty sub-module in the upper arm and its corresponding third sub-module in the lower arm to balance the energy between the two arms. Furthermore, the FTCU isolates v_c^a from 2.25 kV to 3 kV to compensate for the loss of two sub-modules. After the second fault is identified, the FTCU isolates the first sub-module in the lower arm of phase a, which is faulty, and re-inserts the third sub-module in the same arm, which is previously bypassed as a response from

the FTCU against the first fault. Hence, the energy between the two arms is always balanced and the capacitor voltages of the remaining sub-modules are regulated at the new set vale, 3KV.

The proposed RLS scheme accurately estimates the capacitor voltages during the different operating conditions as indicated in Fig. 22(a). Fig. 22(b) demonstrates that the FTCU succeeds to limit the magnitude of the differential current which is tightly tracking its reference signal. Finally, the averaging and output voltage reference control signals are portrayed in Fig. 23 (a) and (b), respectively. It is obvious that the proposed FTCU stabilizes the operation of the control loops during the different types of faults. The results explore the perfect behavior of the proposed control technique under different types of switch faults.



Fig. 19: Dynamic performance of the proposed FDU and FTCU under multiple faults: a) fault 1, b) fault 2.



Fig. 20: The performance of the proposed control under multiple faults: a) Three-phase voltages, b) three- phase load currents.

201



(b) Fig. 21: The performance of the proposed control under multiple faults: a) sub-modules voltages of upper arm in leg a, b) sub-modules voltages of lower arm in leg a.



(b) Fig. 22: Action of the different controllers for leg *a* with enabling the FTCU for multiple failure: a) The capacitor voltage estimation of sub-module *Vcu1*, b) Actual and reference waveform of the circulating current.



Fig. 23: Action of the different controllers for leg *a* with enabling the FICU for multiple failure: a) averaging voltage reference, b) modulating voltage reference for the first sub-module of the upper arm.

IX. CONCLUSION

The paper presents two estimation units based on ADALINE and RLS algorithms for capacitor voltages of the sub-modules of the MMC. For each leg of the MMC, the proposed estimation unit requires only three sensors; one for the phase voltage and the others for the voltages across the arm reactors. Moreover, the proposed RLS estimation unit is integrated with a control strategy to balance the voltages across the sub-modules of the MMC. Furthermore, the estimated capacitor voltages from the two techniques are utilized to detect and localize the open-circuit and shortcircuit faults of sub-module switches. After isolating the faulty sub-modules, the proposed FTCU boosts the capacitor voltage and modifies the phase shift between the carrier signals for the rest of sub-modules in the leg to cope with the reduced number of effective sub-modules. The proposed integrated control strategy eliminates the need for direct measurements of capacitors voltages and their associated communication system which renders it suitable for implementing a low cost centralized controller for the MMC with a large number of sub-modules. In addition, the proposed FDU and FTCU do not need any extra components as they are based on the estimation algorithms that are simultaneously utilized for the proposed control of the MMC.

The fast response and the accurate tracking of the proposed scheme for sub-modules capacitor voltage estimation are revealed from the simulation and the HIL results under different dynamic conditions. Moreover, the proposed MMC controller succeeds to balance the capacitors voltages. The accurate performance of the proposed FDU, for detecting and localizing different faults, results from its hybrid processing where the fault decision is not only based on the reduction on the estimated per-unit voltage but also on the rate of change of the difference between the estimated voltages from the proposed ADALINE and RLS algorithms. Furthermore, the proposed FTCU succeeds to stabilize the operation of the averaging and balancing control loops of the MMC under

204

different types of faults. In addition, the circulating current is tightly tracking its reference signal.

REFERENCES

- [1]
- [2]
- INFERENCES
 S. Debnath, J. Qin, B. Bahrani, M. Saeedifardand and P. Barbosa, "Operation, Control, and Applications of the Modular Multilevel Converter: A Review", IEEE transactions on power electronics, vol. 30, no. 1, January 2015.
 W. Li, L. Gregoire and J. Bélanger, "Control and Performance of a Modular Multilevel Converter System", CIGRÉ Canada Conference on Power Systems, September 2011.
 S. Xu and A. Huang, "Circulating current control of double-star Annual Conference on IEEE Industrial Electronics Society IECON 2012, October 2012.
 M. Saeedifard and R. Iravani, "Dynamic performance of a modular [3]
- [4]
- 2012, October 2012.
 M. Saeedifard and R. Iravani, "Dynamic performance of a modular multilevel back-to-back HVDC system", IEEE Transactions on Power Delivery, vol. 25, pp. 2903-2912, October 2010.
 R. Darus, J. Pou, G. Konstantinou, Salvador Ceballos and Vassilios G. Agelidis, "A Modified Voltage Balancing Sorting Algorithm for the Modular Multilevel Converter: Evaluation for Staircase and Phase-Disposition PWM", 29th Conference on Applied Power Electronics APEC 2014, March 2014.
 A. Aptronomicals, L. Anordit, H.D. Marchand, and Market Applied Power Statement and Phase-Disposition PWM", 29th Conference on Applied Power Electronics APEC 2014, March 2014. [5]
- A Antonopoulos, L. Angquist, H-P. Nee, "On dynamics and voltage control of the Modular Multilevel Converter", 13th European Conference on Power Electronics and Applications EPE'09, September 2009
- 2009. M. Hagiwara and H. Akagi, "Control and Experiment of Pulse width-Modulated Modular Multilevel Converters", IEEE Transactions On Power Electronics, Vol. 24, No. 7, July 2009. L. Hui, L. Chiang, F. Blaabierg, "Review of fault diagnosis and fault-tolerant control for modular multilevel converter of HVDC", 39th Annual Conference of the IEEE Industrial Electronics Society, IECON, 2013. Wenping, X. Dehong, N.E. Prasad, L. Haijin, T.H. Joshua and S.K. Harish, "Survey on Fault-Tolerant Techniques for Power Electronic Converters" IEEE transactions on power electronics up 29 no. [8]
- [9] Converters", IEEE transactions on power electronics, vol. 29, no. 1,

- Converters", EEE transactions on power electronics, vol. 29, no. 1, December 2014.
 [10] R. Wu, F. Blaabjerg, H. Wang, M. Liserre, & F. Iannuzzo, "Catastrophic failure and fault-tolerant design of IGBT power electronic converters-An overview", 39th Annual Conference of the IEEE Industrial Electronics Society, IECON, 2013.
 [11] H. Wang & F. Blaabjerg, "Reliability of Capacitors for DC-Link Applications in Power Electronic Converters-An Overview". IEEE Transactions on Industry Applications, 2014.
 [12] Z. Wang, X. Shi, L. M. Tolbert, F. Wang & B. J. Blalock, "A di/dt feedback-based active gate driver for smart switching and fast overcurrent protection of IGBT modules". IEEE Transactions on Power Electronics, 2014.
 [13] M. Rodriguez-Blanco, D. Theilliol, G. Vela-Valdes, P. Sibaja-Terán, L. Hernandez-Gonzalez, & J. Aguvo-Alquicira, "A failure-detection"
- M. Roonguez-Blanco, D. Infelinoi, G. Vela Values, P. Stoga Ieran, L. Hernandez-Gonzalez, & J. Aguayo-Alquicira, "A failure-detection strategy for IGBT based on gate-voltage behavior applied to a motor drive system". IEEE Transactions on Industrial Electronics, 2011. S. Sedghi, A. Dasfaha and A. Ahmadyfard, "Fault Detection of a Seven Level Modular Multievel Inverter via Voltage Histogram and Neural Neurosci". Interneticael Conference on State Detection on a Seven
- [14]
- [17] Steagn, T. Bakani metric in Annual Strategy in the Recent of a result Level Modular Multilevel Inverter via Voltage Histogram and Neural Network", International Conference on Power Electronics, pp. 1005-1012, Shilla Jeju, Korea, Asia, May 30-June 3, 2011.
 [15] W. Zhang, D.Xu, P. N. Enjeti, H. Li, J. T. Hawke, & H. S. Krishnamoorthy, "Survey on fault-tolerant techniques for power electronic converters", IEEE Transactions on Power Electronics, 2014.
 [16] G. T. Son, H. J.Lee, T. S.Nam, Y. H.Chung, U. H.Lee, S. T. Baek & J. W. Park, "Design and control of a modular multilevel HVDC converter with redundant power modules for noninterruptible energy transfer", IEEE Transactions on Power Delivery, 27(3), 1611-1619.2012
 [17] Song, W., & Huang, A. Q. (2010). Fault-tolerant design and control strategy for cascaded H-bridge multilevel converter-based STATCOM. Industrial Electronics, IEEE Transactions on, 57(8), 2700-2708.
 [18] S. Yang, A. Bryant, P. Mawby, D. Xiang, L. Ran, P. Tavner, "An industry-based survey of reliability in power electronic converter", "An

- [18] S. Yang, A. Bryant, P. Mawby, D. Xiang, L. Ran, P. Tavner, "An industry-based survey of reliability in power electronic converter", IEEE Transactions on Industry Applications, 2011.
 [19] L.G. Franquelo, J. Rodriguez, JJ. Leon, S. Kouro, R. Portillo, and M.A.M. Prats, "The age of multilevel converters arrives", Industrial Electronics Magazine, IEEE, vol.2, no.2, pp.28,39, June 2008.
 [20] A. Beddard, M. Barnes and R. Prece, "Comparison of detailed modeling techniques for MMC employed on VSC-HVDC schemes." IEEE transactions on power delivery, 2015.
 [21] O. Kaijan, R. Hong, C. Zexinag, G. Haiping, L. Xuehua, G. Lin, M. Trevor, W. Bruce, and C. Yuan, "MMC-HVDC Simulation and Testing

205

Based on Real-Time Digital Simulator and Physical Control System." IEEE Journal Emerging and Selected Topics in Power Electronics, 2014.

- [22]
- [24]
- System, 'IEE Journal Enterging and Sciected Topics in Power Electronics, 2014.
 G. Konstantinou and V. Agelidis, "Performance evaluation of half-bridge cascaded multilevel converters operated with multicarrier sinusoidal PWM techniques", 4th Conference on Industrial Electronics and Applications ICIEA 2009, May 2009.
 A. Hassampoor, S. Norga, H. Nee, and L. Angquist, "Evaluation of different carrier-based PWM methods for modular multilevel converters for HVDC application", 38th Annual Conference on IEEE Industrial Electronics Society IECON 2012, October 2012.
 D-W. Kang, W.-K. Lee, and D.-S. Hyun, "Carrier-Rotation strategy for voltage balancing in flying capacitor multilevel inverter", IET Proceedings on Electric Power Applications, March 2004.
 B.-S. Jin, W. kyo Lee, T.-J. Kim, D.-W. Kang, and D.-S. Hyun, "A study on the multi-carrier PWM methods for voltage balancing of flying capacitor in the flying capacitor multi-level inverter", 31st Annual Conference of IEEE Industrial Electronics Society IECON 2005, November 2005. [25] 2005. November 2005.
- 2005, November 2005. G.S. Konstantinou, and V.G. Agelidis, , "Performance evaluation of haff-bridge cascaded multilevel converters operated with multicarrier sinusoidal PWM techniques", 4th IEEE Conference on Industrial Electronics and Applications ICIEA2009, May 2009. [26]
- M.L. Marci, E. F. El-Saadany, and M. M. A. Salama, "Experimental evaluation of envelope tracking techniques for voltage disturbances," Electric Power System Research (Elsevier), Vol. 80, No. 3, pp. 339 344, March 2010. [27]
- [28] M.I. Marei, N. El-Savad, and A. A. El-Sattar "PV interface system with ULT Rando, Marka and Araba Lobata a Transformer and the system with UKRT capability based on a current controlled HFAC link converter Sustainable Energy Technologies and Assessments, Vol. 9, pp. 55-62, March 2015 B. Widrow and M. A. Lehr, "30 years of adaptive neural networks:
- [29] perceptrons, madaline and backpropagation" IEEE Proc., Vol. 78, pp. 1415-1442, September 1990. H. Stefen, "Robust stability analysis of adaptation algorithms for single perceptron," IEEE Transactions on Neural Networks, vol. 2, pp. 325–
- 328, 1991.
- R. Johansson, System Modeling and Identification, Prentice-Hall. Inc.. [31] NJ. 1993.
- NJ, 1995. H. Akagi, S. Inoue, and T. Yoshii, "Control and performance of a transformerless cascade PWM STATCOM with star configuration", IEEE Transactions on Industrial Applications, vol. 43, no. 4, pp. 1041– 1041–1042
- [33] [34]
- IEEE Transactions on Industrial Applications, vol. 43, no. 4, pp. 1041– 1049, Jul/Aug. 2007.
 Bjorn Jacobson, et al., VSC-HVDC Transmission with Cascaded Two-Level Converters, CIGRE 2010.
 K. Tahata, S. El Oukaili, K. Kamei, D. Yoshida, Y. Kono, R. Yamamoto, and H. Ito, "HVDC circuit breakers for HVDC grid applications", 11th IET International Conference on AC and DC Power Transmission, 2015.

Reliability Enhancement of Modular Multilevel Converter by Applying Fault Tolerant Control

Mahmoud Abdelsalam, Hatem Diab, Sarath Tennakoon and Alison Griffiths School of electrical engineering Staffordshire university Stoke on Trent, United Kingdom mahmoud.abdelsalam@research.staffs.ac.uk

Abstract— Modular Multilevel Converters (MMC) are considered very suitable for the transmission of bulk power. Increasing the reliability of MMCs forms a real challenge since they are built from series connection of sub-modules. Applying the concept of fault tolerance to the MMC control system can significantly increase the reliability of the MMC as it will be available in normal as well as faulty conditions. This paper presents an analysis of the MMC reliability enhancement gained from applying different fault tolerant control techniques. The reliability is investigated through solving differential equations generated from Markov chains.

Keywords-Markov chains; fault tolerant control; MMC.

I. INTRODUCTION

The fault tolerant control (FTC) system is the system which is able to perform its desired control functions at high level of stability and security during the occurrence of a certain fault. FTC systems are gaining the interest of many researchers in different research fields because of their ability to enhance the overall reliability of systems. FTC systems are classified into two main types: passive and active [1-3]. As shown in Fig. 1, active FTC system receives the

As shown in Fig. 1, active FTC system receives the continuous fault status from the FDU. Then at the fault condition, it re-configures the control system to a new system which isolates the faulty part of the system and maintains the stability of the plant. It is important to mention that the redesign process of the system is performed online which normally may allow some changes to desired control functions or degradation in the performance [4-5].

The main disadvantage of active FTC techniques is the slow response of them due to the time consumed on the fault detection and controller redesign. However, the main advantage of these techniques is their ability to detect wide range of faults [5].

In passive FTC, the controller is designed in the offline mode then it operates with fixed behavior during normal and abnormal conditions. Therefore, it should be designed such that it can tolerate the faults [6-7]. The main advantages of passive FTC techniques can be summarized in the following points: There is no need for the fault detection process as the system is working with the same parameters during normal as well as fault conditions.

• The response of them is very fast due to the elimination of FDU and the control redesign processes.

However, it is very difficult for one passive FTC system to detect different types of faults since they are designed in the offline mode [8].



Fig. 1: Structure of active FTC.

According to the literature, there are different types of submodule faults that can strike MMCs and affect their operation [9-10]. Since MMCs are used in important applications and at the same time they form a big investment, it is extremely important to detect these fault very quickly to protect the whole system from sudden failures. For this particular purpose, the sub-module faults are investigated in the following subsections. Moreover, popular fault detection techniques are discussed to find out the advantages and disadvantages of each method and pointing what is needed in the process of fault detection of MMCs.

MMCs should work at high level of reliability as they usually work in critical applications that do not have the luxury of losing the power transfer. Based on this fact, it is highly recommended to provide the control of the MMC with fault tolerance capability to make the MMC able from delivering the desired power during faulty conditions without the need to block the converter [11].

This paper presents an investigation of MMC reliability enhancement when two different fault tolerant control techniques are used. The first technique is based on using redundant sub-modules while the second depends on modifying the control of the MMC during faults. Section II presents the operation concept of the MMC and its mathematical model. The reliability analysis using Markov chains of the MMC is illustrated in section III. Section IV shows the conclusion and major findings of the presented work.

II. MODULAR MULTILEVEL CONVERTERS

A. The structure of MMCs

The MMC consists of sub-modules connected in series forming a leg in each phase. The sub-module can be a halfbridge or a full-bridge and each sub-module has a capacitor that buffers the energy from the dc to the ac side and vice versa; thus the dc-link capacitor is not required. A high-speed bypass switch is added to the output port of the sub-module to isolate the sub-module in case of a fault. As shown in Fig. 1, each phase leg is divided into two arms upper and lower. Each arm has identical numbers of sub-modules to generate balanced voltages in the two arms of each phase. Inductors are installed in arms to smooth and filter the currents [12-13].



Fig. 1. The structure of the MIMC.

B. MMC Operating Modes

- The MMC half bridge sub-module has four operation modes during which energy is transferred:
- Mode 1: When S1 is closed, S2 is opened, and the arm has positive polarity. The current flows into the capacitor, charging it. The sub-module is then inserted. See Fig. 2(a).
- Mode 2: When S1 is opened, S2 is closed, and the arm has positive polarity. The sub-module is bypassed, and the current flows towards the next sub-module, keeping the capacitor charge constant. The submodule is then bypassed. See Fig. 2(b).
- Mode 3: S1 is closed, S2 is opened, and the arm has negative polarity. The capacitor starts to discharge and then the sub-module is inserted. See Fig. 2(c).
- Mode 4: When S1 is opened, S2 is closed, and the arm has negative polarity. The current flows towards the

next module, keeping the capacitor charge constant, after which and the sub-module is bypassed. See Fig. 2(d).



Fig. 2. Different operating modes of MMC sub-modules. (a) Mode 1. (b) Mode 2. (c) Mode 3. (d) Mode 4.

C. Mathematical Model of MMC

For the purpose of designing an inner control system for the MMC, it is necessary to mathematically analyze the converter. The converter arms, shown in Fig. 3, are represented as variable capacitors connected in a series with arm resistance and inductance. The number of sub-modules per arm and the switching frequency is assumed as infinite to simplify the analysis because of the perfect sinusoidal output voltage and total voltage balancing between arms. The arm capacitance insertion is represented with a modulation value m_{armx} , where x may be μ for upper arm or l for lower arm. The m_{armx} is varied from 0 to 1. For example, when $m_{armx} = 0$, this indicates that all sub-modules in the upper arm are bypassed and when $m_{armt} = 1$, this indicates that all sub-modules in the lower arm arm inserted. The value Σv_{Cx} is the sum of arm capacitor voltages. The arm voltage is given by: $v_x(t) = m_{armx}(t)$. Σv_{Cx} .

$$C_{insx} = \frac{C_{SM}}{Nm_{arm\,x}(t)}$$
(2)

where C_{SM} is the capacitance of one sub-module and N is the number of sub-modules per arm. If the arm current is $i_x(t)$, the total capacitor voltage dynamics can be expressed by: $G_{SM} = -i_x(t)$

$$\frac{d2v_{Cx}}{dt} = \frac{t_x(t)}{c}.$$
(3)

 $dt = C_{insx}$ Substituting (2) in (3), the capacitor voltage dynamics can be expressed for the upper and lower arms by: $d\Sigma_{Port} = Nm_{expressed}$

$$\frac{d2V_{Cu}}{dt} = \frac{NMarmulu}{C_{SM}}$$
(4)

$$\frac{d\Sigma v_{Cl}}{d\Sigma v_{cl}} = \frac{N.m_{arm\,l}.i_l}{M_{arm\,l}}$$
(5)

 $dt = C_{SM}$ The current i_{aiff} is the differential current, which circulates between the phase legs. The circulating current of any phase is described as follows:

 $i_{diff} = \frac{i_u + i_l}{2}$. (6) By performing the necessary circuit analysis, the output phase

$$v_{0} = \frac{v_{dc}}{2} - L_{arm \ dt} - m_{arm \ u} \Sigma v_{Cu}$$
(7)

$$v_o = -\frac{v_{da}}{2} + L_{arm} \frac{di_l}{dt} + m_{arm \ l} \Sigma v_{Cl}.$$
 (8)



Fig. 3. The MMC average model

III. MMC RELIABILITY ANALYSIS

A system failure can be represented by its probability to A system hande can be represented by its protonly to occur, this is done considering the process behavior as a random variable which receive its value from finite state space elements. To assess the reliability of the MMC, firstly the failure rate of the IGBT λ_0 should be assumed. The reliability function R(i) is the probability of success for the system which is expressed as:

R(t) = 1 - F(t)(9) Where F(t) is the probability of the system failure which is complementary to the R(t) and is given by:

 $F(t) = \int_0^t f(\tau) d\tau$ (10)Where f(t) is rate of change for the failure probability of a certain components, thus (10) can be rewritten as: $f(t) = \frac{dF(t)}{dt}$ (11)

 $f(t) = \frac{dt}{dt}$ (11) Another important definition is the mean time to failure (MTTF) which gives an indication for the expected life time of a component that cannot be repaired or maintained, the MTTF is expressed as:

 $MTTF = \int_0^\infty R(t)dt$

For a system without a fault tolerant control, the MMC will fail if one sub-module suffers from a fault in either the upper or lower legs. To model the reliability of the converter in this case, a Markov chain is developed as shown in Fig. 4. In this case the failure rate for one sub-module in a certain leg

(12)

should equal:

$$\begin{split} \lambda_1 &= (2N-1)\lambda_0 \eqno(13) \\ \text{Where } \lambda_0 \text{ is the failure rate of each IGBT switch.} \end{split} \tag{13}$$
this gives:

 $\frac{dP_1}{dt} = -2\,\lambda_1\,P_1(t)$ (14)

 $\frac{dt}{dt} = -2 \lambda_1 t_1(t)$ Solving the first order differential equation presented in (10), (15)

Solving the first order all refendual equation presented in the reliability function can be expressed as: $R(t) = P_1(t) = e^{-2(2N-1)\lambda_0 t}$ From (15) substituting in (12), the mean time to failure is: $MTTF_{MMC without FTC} = \frac{1}{2\lambda_0} \left(\frac{1}{(2N-1)}\right)$ (16)



IV. MMC Reliability Enhancment Using FTC

The enhancement of the MMC reliability is calculated while applying two FTC techniques. The first technique achieves the fault tolerance using redundant sub-modules while the second depends on modifying the control of the MMC without using redundant sub-modules.

A. FTC using redundant sub-modules

Regarding tolerating the fault using redundant sub-modules, the idea depends on disconnecting the faulty sub-module and inserting a new redundant sub-module without the disconnection of the MMC as shown in Fig. 4 [13-15].

The process of insertion of the redundant sub-modules is done using a strict control. At each triggering time, the control system identifies the state of each sub-module by the aid of a suitable fault detection algorithm. If any sub-module is subjected to high switching commutations or a fault, the control system bypasses the faulty sub-module and replaces it by a reductor the method [15]. redundant sub-module [15].



Fig. 4: Structure of MMC with redundant sub-modules

In this case, the reliability of the MMC can be figured out from In this case, the remaining of the MAC can be figured out from Fig. 5 which has two failure rates. The first failure rate when Iosing one sub-module, λ_1 . While the second failure rate, λ_2 indicates losing two sub-modules in the same leg. In the sub-module redundant based FTC, each leg is assumed to be

equipped with one redundant sub-module. Thus, the submodule failure rates are given by:

$$\lambda_1 = \left(\frac{2N-1}{N^2} + 1\right)\lambda_0 \tag{17}$$

 $\lambda_2 = (\frac{2N-1}{2})\lambda_0$ (18) According to Markov chain, the failure probability of the first state is:

(19)

 $\frac{dP_1}{dP_1} = -\lambda_1 P_1(t)$

dtSolving the differential equation given in (19) gives:

 $P_1(t) = e^{-(\frac{2N-1}{2}+1)\lambda_0 t}$

(20) Following the same way, the failure probability of state 2 gives: $\frac{dP_1}{dt} = \lambda_1 P_1(t) - \lambda_2 P_2(t)$ (21)

^{dt} Solving (21) gives:

Solving (21) gives: $P_2(t) = \left(\frac{2N-1}{2}\right)\left(e^{-\left(\frac{2N-1}{2}\right)\lambda_0 t} - e^{-\left(\frac{2N-1}{2}+1\right)\lambda_0 t}\right)$ (22) From (20) and (22), the probability of the fail state is the summation of $P_1(t)$ and $P_2(t)$:

$$R(t) = P_{1}(t) + P_{2}(t) = \left(\left(\frac{1-2N}{2} \right) e^{-\left(\frac{2N-1}{2}+1\right)\lambda_{0} t} \right) + \left(\frac{2N-1}{2} \right) \left(e^{-\left(\frac{2N-1}{2}\right)\lambda_{0} t} \right)$$
(23)

 $(\frac{2N-1}{2})(e^{-(\frac{2N-1}{2})\lambda_0 t})$

From (23) substituting in (12), the MTTF is: $MTTF_{MMG with FTG 1} = \frac{1}{\lambda_0} \left(\frac{1-N}{N+1} \right)$





Fig. 5. Markov chain of the MIMC with the first FTC technique.

B. FTC based on modifing the MMC inner control

Unlike the previous technique, the MMC is not equipped by redundant sub-modules. The MMC inner control modifies the control and isolates the faulty sub-modules. To simplify the calculations and give a sense for the results, the reliability calculations will be performed for the simulation model which incorporates four sub-modules per arm. In this particular case, the Markov reliability chain is developed as shown in Fig. 6. The Markov reliability chain shows that the MMC with the proposed FTCU have four possible finite states before reaching the failure state instead of one as shown in the previous section. The possible states are:

- State 1: The healthy state (no faulty sub-modules).
- . State 2: Losing a sub-module in the upper arm.
- . State 3: Losing a sub-module in the lower arm. State 4: Losing two sub-modules (one in the upper arm
- and one in the lower arm).
- Fail state: Losing two sub-modules in the upper arm or two sub-modules in the lower arm or both.

In this case, two failure rates should be presented λ_1 and λ_2 . λ_1 is the failure rate of one sub-module and will be the same as calculated in (13). λ_2 is the failure rate for losing two submodules and should equal: (25)

 $\lambda_2 = (2N-3)\lambda_0$ Based on the Markov chain, the probability for the first state P1will remain the same as presented in (15). Regarding the probability of the second state P2, it can be expressed as follows:

 $\frac{dP_2}{dt} = -(\lambda_1 + \lambda_2)P_2(t) + \lambda_1 P_1(t)$ (26) From (26) substituting in (11), the solution of the differential equation gives:

$$P_2(t) = \frac{2N-3}{2} \left(e^{-(\lambda_1 + \lambda_2)t} - e^{-2\lambda_1 t} \right)$$
(27)

 $P_3(t)$ has the same probability of $P_2(t)$ since the conditions are the same. Moving to the forth state probability, it is also can be

emphasized from the Markov chain as follows: $\frac{dP_4}{dt} = -(2\lambda_2)P_4(t) + \lambda_1P_2(t) + \lambda_1P_3(t)$ (28)

From (28) and substituting in (11), the solution of this differential equation is expressed as: $P_4(t) = \frac{(2N-3)^2}{4} \left(e^{-2\lambda_1 t} + e^{-2\lambda_2 t} \right) - \frac{(2N-3)^2}{4} e^{-(\lambda_1 + \lambda_2)t}$ (29)

The reliability function R(t) is the summation of the four (24) probabilities as follows $R(t) = P_1(t) + P_2(t) + P_3(t) + P_4(t)$

$$= \left((2N-2) + \frac{(N-3)^{\alpha}}{4} \right) e^{-2(2N-1)\lambda_0 t} + \left((2N-3) - \frac{(2N-2)^2}{4} \right) e^{-2(2N-2)\lambda_0 t} + \left((2N-2) - \frac{(2N-2)^2}{4} \right) e^{-2(2N-2)$$

 $\frac{e^{2}}{2}e^{-2(2N-2)\lambda_{0}t} + \left(\frac{(2N-3)^{2}}{4}\right)e^{-2(2N-3)\lambda_{0}t}$ (30)From (6.18) substituting in (6.8), the mean time to failure is:



Fig. 6. Markov chain of the MIMC with the second FTC technique.

V. RESULTS AND DISCUSSIONS

To compare between the reliability before and after the application of the two fault tolerant control strategies, the reliability density functions presented in (15), (23) and (30) are plotted against time for five level MMC. The IGBT failure rate λ_0 is assumed to be $10^{-7}h^{-1}$ which is almost three years.

As shown in Fig. 7, the reliability density function is considerably enhanced which means that the rate of failure for the MMC has been decreased. This is also clearly appearing in table I where the MTTF has significantly increased when applying the two FTC strategies.



Fig.7: The reliability of the MMC using different FTC techniques

| TABLE I Comparison between MTTF Associated with different FTC techniques | | | |
|--|-----------------|--|--|
| CONDITION | MTTF (IN HOURS) | | |
| Without FTC | 714285 | | |
| With redundancy FTC | 8750000 | | |
| With control based FTC | 163958335 | | |

VI. CONCLUSION

In this paper, the reliability of the MMC has been carefully studied and analized using Markov chains. The effect of using FTC on the MMC is investigated. Results showed that using the FTC concept will significantly increase the MTTF of the MMC which means that the MMC will be very reliable since it is availability has increased to contain normal and faulty operation conditions. E

REFERENCES

- REFERENCES
 [1] M. Blanke, R. Izadi-Zamanabadi, S. A. Bogh, and Z. P. Lunau. Fault tolerant control systems-a holistic view. Control Engineering Practice, 5(5):693– 702, 1997.
 [2] M. Blanke, M. Staroswiecki, and N. E. Wu. Concepts and methods in fault-tolerant control. In American Control Conference, volume 4, pages 2606– 2620, 2001.
 [3] M. Blanke, M. Kinnaert, J. Lunze, and M. Staroswiecki. Diagnosis and Fault-Tolerant Control. Springer-Verlag, 2006.
 [4] K. Tsuda, D. Mignore, G. Ferrari-Trecate, and M. Morari. Reconfiguration strategies for hybrid systems. In American Control Conference, volume 2, pages 867–873, 2001.
 [5] Y. Zhang and J. Jiang. Bibliographical review on reconfigurable fault tolerant control systems. Annual Reviews in Control, 32(2):229–252, 2008.
- [6] G. H. Yang, Zhang S. Y., J. Lam, and J.L. Wang. Reliable control using redundant controllers. IEEE Transactions on Automatic Control, 43(11):1588–1593, Nov 1998.
 [7] R.J. Veillette. Reliable linear-quadratic state-feedback control. Automatica, 21(1):237–244, 1005
- [7] K.S. Verheit, K.Ruber III, and San and San Periodek Control. Protoinated, 31(1):137–144, 1995.
 [9] G.H. Yang, J.L. Wang, Y.C. Soh, and K.Y. Lou. Reliable state feedback control synthesis for uncertain linear systems. Asian Journal of Control, 5(2):301–308, 2003.
- E. Wolfgang, 'Examples for failures in power electronics systems," in presented at ECPE Tutorial Reliability of Power Electronic Systems, Nurenberg, Germany, Apr. 2007.
 K. Fischer, T. Stalin, H. Ramberg, T. Thiringer, J. Wenske, and R. Karlsson, Investigation of converter failure in wind turbines," in Elforsk report, Nov. 2012.

- (1904, 1904, 2012)
 [12] J. Qin, & M. Saeedifard, "Predictive control of a modular multilevel converter for a back-to-back HVDC system", IEEE Transactions on Power Delivery, vol 27 no. 3, 1538-1547, 2012.
 [13] S. Debuath, J. Qin, B. Bahrani, M. Saeedifardand and P. Barbosa, "Operation, Control, and Applications of the Modular Multilevel Converter: A Review", IEEE transactions on power electronics, vol. 30, no. 1, January 2015.
 [14] W. Zhang, D.Xu, P. N. Enjeti, H. Li, J. T. Hawke, & H. S. Krishnamoorthy, "Survey on Bault-tolerant techniques for power electronic converters", IEEE Transactions on Power Electronics, 2014.
 [15] Son, G. T., Lee, H. J., Nam, T. S., Chung, Y. H., Lee, U. H., Baek, S. T., ..., & Park, J. W. (2012). Design and control of a modular multilevel HVDC converter with redundant power modules for noninterruptible energy transfer.Power Delivery, IEEE Transactions on, 27(3), 1611-1619.

Enhanced Approach for Modelling and Simulation of Modular Multilevel Converter Based Multiterminal DC Grids

Mona F. Moussa Energy Engineering Department, College of Engineering and Technology, Arab Academy for Science and Technology and Maritime Transport Cairo, Egypt c-mail: mona.moussa@yahoo.com

Abstract— Multi-terminal direct current (MTDC) grids are gaining the interest of many researchers because of their unique features that improve the transmission network functionality and reliability. Unlike two level and three level voltage source converters (VSC), modular multi-level converters (MMC) are suitable for the application of MTDC grids because of their generous advantages such as the elimination of DC link capacitor, the improved quality of the output power and the lower footprint. However, simulation of MMC based MTDC grids is a real challenge because of the huge computational burden due to the high number of sub-modules. This paper proposed a modeling approach for a four terminal MTDC grid which is based on average MMC models with 401 voltage levels. The control system structure which controls the power through the MTDC grid and the MMCs at each terminal is presented. The whole system is tested and simulated using PSCAD/EMTDC under different operating conditions to verify the validity and evaluate the performance of the proposed modelling approach.

Keywords-MTDC; MMC; Power control; PSCAD

I. INTRODUCTION

The high voltage direct current (IIVDC) transmission systems are currently involved in many projects especially for transmission systems that transmit bulk amounts of power for long distances. This is simply because of the reduced losses of HVDC compared to the high voltage alternating current (HVAC) transmission as shown in figure.1. Also the biplar HVDC circuit is formed from two cables only compared to four cables in HVAC circuits. Moreover, the HVDC systems do not suffer from stability problems due to the absence of reactance effect which makes them have no distance limitations [1].

Instead of having many point to point transmission lines, MTDC grids are attractive for many applications such as: connecting different grids together, integration of off-shore wind energy to AC grids, etc [2]. Figure.2 shows a general diagram for a four terminal DC grid. The conversion from AC to DC and vice versa is performed using VSC which has different topologies. One of the promising topologies in IIVDC transmission is the MMC because of the improved quality of the output power due to the sufficient number of voltage levels which eliminates the need of adding filtering Mahmoud O. Abdelsalam and Hatem Y. Diab School of Electrical Engineering, Faculty of Computing, Engineering and Sciences , Staffordshire University Stoke on Trent, United Kingdom e-mail: Mahmoud.abdelsalam@research.staffs.ac.uk

equipment [3]. The modelling of the MMC based MTDC grids is necessary for studies and analysis purposes to enable the researchers and engineers to have better understanding of the operation and behavior of such networks. However, the modelling of the MMC is not an easy task since the number of sub-modules can be in hundreds which makes it very difficult to develop a detailed simulation model due to the excessive and massive computational burden [4]. To solve this problem, a suitable average model can be used to lower the calculations necded [5]. This paper presents a modelling approach for the simulation of four terminals DC grid using PSCAD/EMTDC simulation software, each terminal includes an average model for a 401 level MMC. Section II explains the MMC structure including the mathematical analysis for the developed average model while section III and IV introduce the control system structure which controls the terminal, whereas, section V indicates an analysis of the results acquired under different operating conditions. Finally, section VI demonstrates the conclusion of the research verifying the main ideas based on the simulation results.



Figure, 1: Comparison between the losses associated with the HVAC and HVDC transmission lines

Figure. 2: A general diagram for a four terminal grid.

II. MODELLING OF THE MMC

A. MMC structure and operation modes

As demonstrated in figure 3, the MMC is formed from three legs, each leg is formed from two arms connected to each arm through arm inductors, these inductors are responsible for smoothing the arm currents. Each arm consists of series connected sub-modules which can be either full bridge or half bridge connected to a capacitor. The sub-module is considering to be the basic unit of the MMC as it exchanges the energy between the AC and DC sides of the converter. There are two main operation modes for the MMC as shown in table 1 [6].



Figure.3. Three-phase MMC topology and internal sub-module.

| TABLE I. | MMC MODES OF OPERATION |
|----------|------------------------|
| | |

| Mode | I | п | |
|-------------|---|---|--|
| \$1 \$2 | Switched on | Switched off Switched on | |
| | Switched off | | |
| Description | Output voltage from the sub- module has a positive polarity. The capacitor is charging/discharging, depending on the current direction. | The sub-module is bypassed and the current flows towards the next sub-module keeping the capacitor charge constant. | |

B. Mathematical representation of the average model

The average model used in the simulation depends on converting the detailed model of the sub-module to its thevenin equivalent. As shown in figure. 4, the voltage across the sub-module is given by identifying its equivalent resistance and voltage. The equivalent resistance can be expressed as shown below [5]:

$$R_{eq} = R_2 \left(1 - \frac{\kappa_2}{R_1 + R_2 + R_c} \right) \tag{1}$$

 R_1 and R_2 are the resistances of the upper and lower IGBTs, when the IGBT is switched on, its resistance will equal the value of the on state resistance which is about 0.001 Ω compared to 100000 $k\Omega$ in the off state. R_c is the equivalent resistance of the sub-module capacitor having a capacitance of C_i this resistance can be calculated using the following expression:

$$R_c = \frac{\Delta T}{2c}$$
(2)

Where ΔT is the time step of the simulation. The equivalent voltage of the sub-module thevenin model is given by: $V = -\frac{\kappa_2}{2} V$ (3)

 $V_{eq} = \frac{R_2}{R_1 + R_2 + R_c} V_{eq}$ (3) V_{ceq} is the equivalent voltage across the capacitor which is given by:



Figure. 3: Thevenin equivalent of the sub-module.

III. CONTROL OF ACTIVE AND REACTIE POWER THROUGH THE MTDC GRID

Controlling the flow of active and reactive powers in the grid is done by controlling both the DC and AC voltage at each terminal. The control system at each terminal is formed from five units as described in the following sub-sections.

A. Phase locked loop

The phase locked loop technique is responsible for synchronizing the AC voltage of the MMC with the grid voltage. This is done using a feed forward PI control loop which receives the quadrature axis voltage of the MMC AC terminals V_{tq} . The output of the PI controller is then added to the initial angular frequency ω_0 . The output of the addition is passed through integrator resulting into the grid phase θ .

B. AC and DC voltage control loops

The active and reactive power control loops are very simple as they utilize two PI controllers to decide the values of the

MMC direct and quadrature components of the desired MMC output current i_d^* and i_q^* .

C. Inner current loop

The inner current control loop is used to regulate the MMC output current based on the received references from the DC and AC control loops, the loop is designed based on the MMC dynamics of the AC side: (5)

$$L_{grid} \frac{di_d}{dt} = i_q \omega L_{grid} - i_d R_{grid} + V_{td} - V_{sd}$$

 $\begin{array}{l} L_{grid} \begin{array}{c} d_{t} & q & q, \\ L_{grid} \begin{array}{c} d_{t} \\ d_{t} \end{array} = -i_{d} \omega L_{grid} - i_{q} R_{grid} + V_{t q} - V_{s q} \end{array} \tag{6}$ Where L_{grid} and R_{grid} are the grid inductance and resistance. V_{t} is the MMC terminal voltage while V_{s} is the

grid voltage. According to (5) and (6), two PI controllers are added to

regulate the i_d and i_q . The output of the two controllers is then added to the measured components of the grid voltage V_{sd} and V_{sq} forming two feed forward loops. The decoupled components of i_d and i_q are then subtracted from the two loops resulting in the reference terminal voltages V_{td}^* and V_{ta}^* . Figure 4 shows the block diagram of the active and reactive power control [7].



Figure. 4: Active and reactive power control.

IV. THE INNER CONTROL OF THE MMC

The inner control system of the MMC is responsible for achieving three important tasks: the first is balancing the voltage between the capacitors while the second is suppressing the differential current. The third task is the switching of the MMC.

A. Capacitor voltage balancing using the energy control

In this method, the capacitor voltages are balanced by controlling the total energy stored in the leg and the differential energy between the upper and lower arms [8]. For better understanding of this method, the energies of the upper and lower arms should be mathematically expressed as follows: (7)

$$W_{Cx}^{\Sigma} = \Sigma C_{effx} (V_{Cx}^{\Sigma})$$
Where user has the upper arm of *l* for lawy

can be u for the upper arm or l for lower arm. Differentiating equation (3.1), this gives:

$$\frac{dW_{Cu}^{2}}{dt_{-}} = (\frac{i_{\nu}}{2} + i_{diff})(\frac{v_{dc}}{2} - e_{\nu} - v_{diff})$$

$$\frac{dW_{Cl}^{\sigma}}{dt} = \left(\frac{i_v}{2} + i_{diff}\right)\left(\frac{v_{dc}}{2} + e_v - v_{diff}\right)$$

From (8) and (9), the following equations express the total energy stored in the leg W_C^Σ and the energy difference between the arms W_{c}^{Δ}

$$V_C^{\Sigma} = W_{Cu}^{\Sigma} + W_{Cl}^{\Sigma}$$
(10)
$$V_C^{\Delta} = W_{Cu}^{\Sigma} - W_{Cl}^{\Sigma}$$
(11)

 $W_C^{\Delta} = W_{Cu}^{\Sigma} \cdot W_{Cl}^{\Sigma}$ Differentiating (10) and (11) and equalizing them with (8) and (9) gives:

$$\frac{dW_c^2}{dt} = (V_{dc} - 2v_{diff})i_{diff} - e_v i_v \tag{12}$$

$$\frac{dW_c^2}{dW_c^2} = (\frac{v_{dc}}{v_{diff}})i_v - 2e_v i_{diff} \tag{12}$$

$$\frac{du_{i}}{dt} = \left(\frac{du}{2} - v_{diff}\right)i_{v} - 2e_{v}i_{diff} \tag{1}$$

Where $v_{diff} = Ri_{diff} + L \frac{a t_{diff}}{dt}$. Equation (11) and (12) shows that the differential current i_{diff} plays a major role in balancing the capacitor voltages of the MMC sub-modules. Based on this, the modulation signals can be expressed as: V_{de}

$$m_u = \frac{\frac{1}{2} - e_{vref} - v_{diff} ref}{v_{Cu}^2}$$
(13)
$$m_l = \frac{\frac{v_{dc} + e_{vref} - v_{diff} ref}{v_{v\Sigma}}$$
(14)

Vδ Although the equation (13) and (14) look very simple, the direct application of them makes the converter unstable [9]. To maintain the stability of the converter, two control loops are needed as shown in figure 5. The first loop is a PI control loop which is responsible for enabling the total stored energy from following its reference, while the second loop is responsible for balancing the energy between the two arms, in other words for canceling W_C^{Δ} .



Figure. 5: Block diagram of the capacitor voltage balancing based on energy control.

B. Circulating current supressing control

The circulating current is a virtual current which circulates between the two arms; it can be expressed by:

$$i_{circ} = \frac{i_u + i_l}{2} \tag{15}$$

Where i_u and i_l are the arm currents. Having a high level of the circulating currents means that the MMC will be subjected to negative consequences such as the increased heat effect and electrical losses. Thus, it is a must to suppress the circulating current to a very small amount. This can be achieved by eliminating the double line frequency component

(8)

(9)

of it. Firstly, the three phase circulating current is transformed into the dq reference frame. Then the dq currents are subtracted from the reference value which is zero. The error signals are passed through two PI control loops and then the output signals are transformed to the three phase reference signals of the circulating voltage. Figure 6 shows the block diagram of the circulating current suppressing control.

at the same time did not contribute to the faulty terminal.



Figure. 6: Block diagram of the capacitor voltage balancing based on energy control.

SIMULATION RESULTS AND DISCUSSIONS V.

A four terminal DC grid is simulated using PSCAD/EMTDC, each terminal contains an average model for 401 level MMC. The simulation parameters are found in table 2.

| GRID | | |
|-----------------------|-----------|--|
| Parameter | Value | |
| Rated AC voltage | 380 kV | |
| Rated DC voltage | 400 kV | |
| Rated power | 10000 MW | |
| MMC voltage levels | 401 level | |
| Arm inductance | 15 mH | |

TABLE II. PARAMETERS OF THE SIMULATED FOUR TERMINALS DC

The simulated case study is demonstrated to show the behavior of the MTDC and the reaction of different control systems under three conditions: 1. The start-up and system energization.

220 µF

The normal operation mode. The faulty condition. 2

Sub-module

capacitance

Firstly the AC grids of the rectifier terminals start to inject active power towards the MMCs to build up the voltage of the DC link. Using the MTDC grid power control loops, the DC voltage reaches its rated level which is 400 kV and settles at Voltage reaches its rated level which is 400 kV and settles at this value after 0.6 sec. A three phase fault is applied to one terminal at the instant of 0.8 sec, then cleared at 1 sec. Figure 7 shows the three phase voltages, currents and the RMS voltage of the MMC where the three phase fault has been applied. It is obvious that the MTDC grid control succeeded applied. It is obvious that the MTDC grid control succeeded to protect the converter since it was blocked in order to isolate the fault. After clearing the fault, the P-Q control loops started quickly to build the DC voltage again. Regarding other terminals, they regulated the DC voltage during the fault and



voltage of a healthy terminal. It is clear that, during the total simulation time, the MTDC control system was able to control

the DC voltage during normal and faulty conditions as shown in figure 9. $\,$

Figure. 7: The MMC performance of the faulty terminal: a) Three phase voltages, b) three phase currents and c) The RMS voltage

Moreover, it has succeeded to control the flow of the power smoothly as demonstrated in figure 10. This also can be proven by figure 11 which shows how the direct and quadrature currents track their reference during the whole period. Regarding the MMC inner control, the capacitor voltages of all MMCs have been successfully balanced during the whole simulation period as presented in figure 12. In addition, the circulating currents have been suppressed to its minimum value to limit the conversion losses.





Figure. 9: The DC voltages of the four terminals.



Figure. 8: The MMC performance of a healthy terminal: a) Three phase voltages, b) three phase currents and c) The RMS voltage

Figure. 11: The DQ currents versus their reference



Figure. 12: The sub-module capacitor voltages of one MMC.

VI. CONCLUSION

This paper presents an enhanced modelling technique for a MMC based four terminal DC grid; where, each terminal includes an average model for a 401 level MMC. An average model for the MMC has been built to lower the computational burden. In addition, the power control including the inner control system of the MMC has been developed according to different industrial standards and physical projects in order to control the flow of active and reactive powers in the grid by controlling both the DC and AC voltage at each terminal. The control system at each terminal was formed from five units.

The simulation model was completed using PSCAD/EMTDC software package to evaluate the performance of the modeled DC network under various operating conditions. **Theoretical** results indicate an efficient behavior of the MTDC control loops during normal and faulty conditions providing a full safe to fail operation of the MMCs and guaranteeing a smooth power flow between the terminals. It can be concluded that the proposed model is suitable as a test platform for real MTDC projects and research purposes.

References

- Meah, K. and Ula, S., "Comparative Evaluation of HVDC and HVAC Transmission Systems," in Power Engineering Society General Meeting, 2007. IEEE, pp. 1–5, 2007.
- [2] Jiebé Zhu; Booth, C. "Puture multi-terminal HVDC transmission systems using Voltage source converters," Universities Power Engineering Conference (UPEC), 2010 45th International , vol., no., pp.1,6, Aug. 31 2010-Sept. 3 2010.
- [3] W. Li, L. Gregoire and J. Bélanger, "Control and Performance of a Modular Multilevel Converter System", CIGRÉ Canada Conference on Power Systems, September 2011.
- [4] S. P. Teeuwsen, "Similified dynamic model of a voltage-sourced converter with modular multilevel converter design," presented at the IEEE/Power Eng. Soc. -Power Systems Conf. Expo., Seattle, WA, Mar. 2009.
- [5] U. N. Gnanarathna, A. M. Gole, and R. P.Jayasinghe, "Efficient Modeling of Modular Multilevel HVDC Converters (MMC) on Electromagnetic Transient Simulation Programs," IEEE Transactions on Power Delivery, vol.26, no.1, pp.316-324, Jan. 2011.
- [6] S. Xu and A. Huang, "Circulating current control of double-star chopper cell modular multilevel converter for HVDC system", 38th Annual Conference on IEEE Industrial Electronics Society IECON 2012, October 2012.
- [7] Chandra Bajracharya, Control of VSC-HVDC for wind power, MSc thesis, Norwegian University of Science and Technology, Department of Electrical Power Engineering, Norway 2008.
- [8] A. Antonopoulos, L. Angquist, H-P. Nee, "On dynamics and voltage control of the Modular Multilevel Converter", 13th European Conference on Power Electronics and Applications EPE'09, September 2009.
- [9] M. Hagiwara, H. Akagi, "Control and Experiment of Pulse width-Modulated Modular Multilevel Converters", IEEE Transactions On Power Electronics, Vol. 24, No. 7, July 2009.